Intrathecal baciofen treatment in children with heurological bacioners

Laura A. Bonouvrié

Intrathecal baclofen treatment in children with neurological disorders

Laura A. Bonouvrié

The work presented in this thesis was performed at the department of Rehabilitation Medicine, VU University Medical Center, Amsterdam, the Netherlands and the department of Child Neurology, Maastricht University Medical Center, Maastricht, the Netherlands.

The trials in this thesis were funded by the Phelps Stichting voor Spastici (projectnumber 2011037 and projectnumber 99.047), HandicapNL (previously Revalidatiefonds) (projectnumber R2011032/), Kinderrevalidatie Fonds Adriaanstichting (projectnumber 11.02.17-2011/0035) and the Johanna KinderFonds (projectnumber 2011/0035-357).

Printing of this thesis was supported by non commercial grants from Phelps Stichting voor Spastici, Dystonie vereniging, OIM Orthopedie (Assen, the Netherlands), Kamer Orthopedie, and Ipsen Farmaceuticals.



ISBN: 978-94-0281448-4

Design cover and chapter pages by Rik Vrijman, Reklamebureau.it Layout and design by Eduard Boxem, persoonlijkproefschrift.nl. Printed by Ipskamp Printing, proefschriften.net

© L.A. Bonouvrié, Amsterdam, the Netherlands

All rights reserved. No part of this publication may be reproduced, stored or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording or otherwise without the prior written permission of the author.

Intrathecal baclofen treatment in children with neurological disorders

ACADEMISCH PROEFSCHRIFT

ter verkrijging van de graad Doctor aan de Vrije Universiteit Amsterdam, op gezag van de rector magnificus prof.dr. V. Subramaniam, in het openbaar te verdedigen ten overstaan van de promotiecommissie van de Faculteit der Geneeskunde op woensdag 30 oktober 2019 om 13.45 uur in de aula van de universiteit, De Boelelaan 1105

door

Laura Alexandra Bonouvrié

geboren te Utrecht

promotoren: prof.dr. J.G. Becher

prof.dr. R.J. Vermeulen

copromotor: dr. A.I. Buizer

I love it when a plan comes together

-Col. Hannibal Smith-

TABLE OF CONTENTS

Chapter 1	General introduction	9
Chapter 2	Clinical presentation and management of dyskinetic cerebral palsy	35
Chapter 3	Effects of intrathecal baclofen on daily care in children with secondary generalized dystonia: a pilot study	61
Chapter 4	Intrathecal baclofen treatment in dystonic cerebral palsy: a randomized clinical trial: the IDYS trial	75
Chapter 5	The effect of intrathecal baclofen in dyskinetic cerebral palsy: the IDYS trial	93
Chapter 6	Intrathecal baclofen for progressive neurological disease in childhood: a systematic review of literature	119
Chapter 7	Satisfaction with intrathecal baclofen treatment in paediatric patients with progressive neurological disease	131
Chapter 8	The effect of intrathecal baclofen treatment on activities of daily life in children and young adults with cerebral palsy and progressive neurological disorders	137
Chapter 9	General discussion	165
Chapter 10	Summary Samenvatting	179
Chapter 11	Appendix I. Dyskinesia measurement scales II. IDYS study group	191
Chapter 12	List of abbreviations	197
Chapter 13	Dankwoord / Word of thanks	201
Chapter 14	About the author	209



INTRODUCTION

Movement disorders in childhood are mostly caused by dysfunction of the developing brain due to brain lesions or brain abnormalities. These so called cerebral movement disorders can lead to positive signs (signs of abnormal muscle activity) such as spasticity or dyskinesia, together with negative signs such as muscle weakness, decreased selective motor control and impaired coordination.¹⁻³ Secondary musculoskeletal problems can arise during growth including muscle contractures, bony deformities, scoliosis and hip displacement.³ Consequently, the impairments on the level of body functions and structures can lead to problems in activities and participation such as (but not limited to) mobility, self-care, communication and learning. Activities and participation are furthermore influenced by environmental factors such as the availability of assistance for personal care and aids for mobility. To comprehend the extent of problems due cerebral movement disorders, the International Classification of Functioning, Disability and Health for Children and Youth (ICF-CY) can be used as a framework (Figure 1 and 2).⁴ The spectrum of the severity of cerebral movement disorders is broad. This thesis will focus on severely affected children, which are children with cerebral palsy (CP) classified with the Gross Motor Functioning Classification System (GMFCS) in levels IV and V or children with progressive neurological disorders (PND) who are equally affected (Figure 3).



Figure 1. The International Classification of Functioning, Disability and Health for Children and Youth model (ICF-CY).⁴

Cerebral movement disorder						
Body function	Activities and					
Positive signs Spasticity and or dyskinesia Abnormal reflexes Co-activation Secondar Contra Scolio Hip dia Bony dia 	 Negative signs Muscle weakness Impaired selective motor control Poor motor coordination Impaired sensory functions (vision , hearing) and pain y problems actures sis splacement deformities	Participation Impairments in: • Mobility • Self-care • Learning and applying knowledge • Performing general tasks and demands • Interpersonal interaction and relationships • Communication				
Environmental factors						
	Products and technologySupport and relationships					

Figure 2. The International Classification of Functioning, Disability and Health for Children and Youth model for cerebral movement disorders

	GMFCS Level I
A A	Children walk at home, school, outdoors and in the community. They can climb stairs without the use of a railing. Children perform gross motor skills such as running and jumping, but speed, balance and coordination are limited.
	GMFCS Level II
	Children walk in most settings and climb stairs holding onto a railing. They may experience difficulty walking long distances and balancing on uneven terrain, inclines, in crowded areas or confined spaces. Children may walk with physical assistance, a hand- held mobility device or used wheeled mobility over long distances. Children have only minimal ability to perform gross motor skills such as running and jumping
	GMFCS Level III
	Children walk using a hand-held mobility device in most indoor settings. They may climb stairs holding onto a railing with supervision or assistance. Children use wheeled mobility when traveling long distances and may self-propel for shorter distances.
~	GMFCS Level IV
	Children use methods of mobility that require physical assistance or powered mobility in most settings. They may walk for short distances at home with physical assistance or use powered mobility or a body support walker when positioned. At school, outdoors and in the community children are transported in a manual wheelchair or use powered mobility.
-	GMFCS Level V
	Children are transported in a manual wheelchair in all settings. Children are limited in their ability to maintain antigravity head and trunk postures and control leg and arm movements.

Figure 3. Gross Motor Functioning Classification System (GMFCS)⁵. Image on the courtesy of K. Graham, The Royal Children's Hospital, Melbourne, Victoria, Australia.

12

CEREBRAL PALSY

The most common cause of cerebral movement disorders and physical disability in childhood is cerebral palsy (CP). The prevalence of CP in Western countries is 2.11 per 1000 livebirths and has been fairly stable for many years.⁶ A European cohort study showed a prevalence of 1.96 per 1000 livebirths in 2010.⁷ CP comprises a group of developmental disorders of movement and posture, attributed to non-progressive disturbances that occurred in the developing fetal or infant brain.⁸ The motor impairments are often accompanied by disturbances of sensation, perception, cognition, communication and behaviour, by epilepsy and by secondary musculoskeletal problems.⁸⁻¹⁰

Neuroimaging in CP reveals several types of underlying brain lesions (Figure 4). The pattern of lesions is strongly related to the stages of brain development in which they occurred: prenatal, perinatal or neonatal.¹¹ Furthermore, the pattern of lesions corresponds with clinical findings.¹²

Brain maldevelopment origins from the first and second trimester.¹³ Disorders of proliferation, migration or organization can occur at this stage. Underlying causes can be genetic, vascular and infectious. Depending on the localisation and extent of the maldevelopment clinical symptoms range from easily controllable epilepsy without further abnormalities to uncontrollable epilepsy accompanied by spasticity and, severely impaired motor and cognitive development.¹⁴

In the early to mid-third trimester periventricular white matter lesions are the dominant finding.¹³ This type of damage occurs approximately before 34 weeks of gestation and is the main pattern in preterm born children.^{12,14,15} However, 25% of children with these lesions are born at term.^{12,13} In this last group of children the origin of the lesions is presumably prenatal due to chronic placental insufficiency with prolonged periods of moderate intensity insufficient oxygenation.^{11,14} Ischemia and inflammation, which can be due to perfusion failure, infection or both can cause periventricular leukomalacia (PVL).¹⁵ A less common cause of white matter lesions is periventricular haemorrhagic infarction (PHI). PHI is a complication of intraventricular haemorrhage which occurs in the first days after birth, especially in very preterm children due to susceptibility for haemorrhage of the germinal layer.¹⁶ White matter lesions result mostly in spasticity.^{12,17} The more extensive the brain lesions, the more severe the clinical findings will be, not only for the movement disorder but also for additional disturbances such as cognitive impairment and cerebral visual impairment.^{12,14,16}

Events in the late third trimester mostly cause cortical/subcortical and basal ganglia gray matter lesions.^{13,14} Lesion of the basal ganglia and in some cases additional damage to the central cortex is a common pattern in asphyxiated infants born at term who present with low Apgar scores after birth.^{11,14,17,18} Possible mechanisms of the selective vulnerability are the high glucose metabolism during the neonatal period and the pattern of myelination starting in the affected central cortical region.¹⁴ Children with basal ganglia lesions present with the dyskinetic form of CP.^{11,12,18} Imaging studies show involvement of the lentiform nuclei (putamen and globus pallidus) in all children with dyskinetic CP. There appear to be two patterns: lesions in the putamen only or in both the putamen and globus pallidus.¹⁹ Cognitive deficits are less profound in children with basal ganglia compared to children with white matter lesions, but when additional (i.e. severe) central, (sub)cortical and/or hippocampal involvement is present, spasticity can be the dominant movement disorder with additionally cognitive impairment.^{18,20} Lesions to the globus pallidus can also occur due to postnatal hyperbilirubinemia-induced kernicterus. In these children hearing impairment is often accompanying the movement disorder.^{21,22} Kernicterus can both occur in preterm and term born infants.^{23,24} The bilirubin threshold for development of basal ganglia lesions is lower in very preterm infants (25 to 29 week of gestation) making them more susceptible to the occurrence of kernicterus.^{22,24} In high income countries this condition has become less common due to preventive strategies.²³



Figure 4. Brain lesions in cerebral palsy. A) Normal brain (1) T2 weighted MRI image); (2) FLAIR MRI image; B) Brain maldevelopment: neuronal migration disorder with complete periventricular band heterotopia (double cortex syndrome) and dilated posterior horn of ventricles (T2 weighted MRI imaged). Clinically bilateral spastic CP, GMFCS II, epilepsy and moderate cognitive impairment; C) Brain maldevelopment: hemimegalencepahly with polymicrogyria of the left hemisphere (FLAIR MRI image). Clinically neonatal status epilepticus; D) Periventricular leukomalacia with loss of white matter and enlargement of ventricles (FLAIR MRI image). Clinically bilateral spastic CP, GMFCS IV. E) Acute perinatal asphyxia with (1) lesions in the ventromedial thalamus and posterior lentiform nucleus and (2) lesions in the motor cortex (rolandic cortex) (both FLAIR MRI image). Clinically bilateral dyskinetic CP, GMFCS V; F) Kernicterus: bilateral globus pallidus lesions (T2 weighted MRI image). Clinically bilateral dyskinetic advinetic cerebral palsy, GMFCS II; G) Metachromatic leukodystrophy: symmetric bilateral white matter abnormalities and corticospinal tract abnormalities (T2 weighted MRI image). Clinically severe bilateral spasticity, legs more than arms, comparable to GMFCS V.

Classification of cerebral palsy

The severity of the impairment of functional mobility in CP is classified using the GMFCS (Figure 3).⁵ Similar 5 point scales are used to describe the level of manual function (Manual Ability Classification System, MACS), and the level of communication function (Communication Function Classification System, CFCS).^{25 26} This thesis will focus on CP patients with GMFCS level IV and V.

CP is furthermore categorised by the dominant movement disorder: spastic, dyskinetic or ataxic.¹⁰ Spasticity is the dominant movement disorder in 72 to 91% of CP patients,^{9,11} and dyskinesia in approximately 15%.¹¹

Spasticity

Spasticity is defined as a velocity-dependent, increased resistance to externally applied passive muscle stretch which is forthcoming during physical examination.^{3,27} Spasticity is the result of an imbalance of the excitatory and inhibitory responses to a sensory input signal. Lesions of the central nervous system result in loss of descending inhibitory commands and abnormal impulses, subsequently resulting in overactive and spastic muscles.² The intensity of spasticity can vary depending on posture, activity, emotional state, pain and other triggers.²⁷

Dyskinesia

Dyskinesia is characterized by involuntary, uncontrolled, recurring, occasionally stereotyped movements with fluctuating muscle tone.²⁸ Dyskinesia can be further separated in dystonia and choreo-athetosis.^{28,29} Dystonia is a movement disorder with involuntary movements, distorted voluntary movements and abnormal postures due to sustained or intermittent muscle contractions such as slow rotation, extension or flexion of body parts.^{28,29} The abnormal postures can give the impression of hypokinesia while they are actually caused by sustained muscle contractions. Tone is fluctuating but easily increased (hypertonia).²⁸ Choreo-athetosis is characterised by (faster) hyperkinetic movements and tone fluctuation (mainly hypotonia).²⁸ Dystonia and choreoathetosis can be simultaneously present with dystonia often being the more dominant feature. They can both manifest during rest and both increase during activities.³⁰ Studies shows that dystonia is, in contrast to choreo-athetosis, significantly related to activities of daily life, participation in society and quality of life.³¹

Clinicians find dyskinesia difficult to recognize and the distinction between dystonia and choreo-athetosis troublesome.³² Measurement scales are therefore not used often.³² Two measurement scales for dystonia are specifically designed for CP, the

Barry-Albright Dystonia Scale (BADS) and the Dyskinesia Impairment Scale (DIS).³³⁻³⁵ The Burke-Fahn-Marsden Dystonia Rating scale was designed to assess primary dystonia but is frequently used in CP as well.³⁵ Scales are presented at the end of this thesis (Appendix I. Dyskinesia Measurement Scales). All three scales assess dyskinesia on the ICF-CY level of body functions and structures. For all scales, studies looking at the validity, reliability, responsiveness and clinical utility in children with CP are limited in number and/or quality. There are no test-retest studies. For clinical utility the BADS seems the easiest and quickest scale to use. The DIS is the most extensive and time consuming scale making it less suitable for clinical practice but it could be of added value in research. The DIS is also the only scale addressing choreo-athetosis and the only scale comparing rest versus activity.³⁵

Dyskinetic CP is the most disabling form of CP with 59-80% of patients being GMFCS level IV or V.^{36,37} This is substantially higher compared to children with bilateral spastic CP in whom GMFCS IV and V account for under 20% of cases.³⁷ Comprehensive studies on dyskinetic CP are rare and outcomes are mostly on the level of body functions and structures (e.g. dystonia). Reported goals for treatment of patients with dystonia are on all domains of the ICF-CY: improve pain and comfort, prevent worsening of deformities (level of body functions and structures), improve positioning and transfers (level of activities and participation), and improve ease of personal care for both patients and caregivers (including dressing and hygienic related care) (level of activities and participation and level of environmental factors).^{38,39}

PROGRESSIVE NEUROLOGICAL DISEASE OF CHILDHOOD

Spasticity and dyskinesia in children can also be caused by progressive neurodegenerative or neurometabolic diseases (PND).⁴⁰⁻⁴² Neurodegenerative diseases are disorders with progressive loss of neurological function due to structural abnormalities of the central nervous system.⁴² Neurometabolic diseases refer to a group of disorders that are characterized by a lack or dysfunction of an enzyme or vitamin necessary for a specific chemical reaction in the body.⁴³ Most of the neurodegenerative and neurometabolic disorders are (very) rare. A brief selection of the (relatively) most common disorders seen in clinical practice and described in literature, causing spasticity or dystonia, is summarized in table 1.⁴⁰⁻⁴³

Progression of PND can be rapid or slow and other symptoms such as epilepsy, cognitive deterioration and muscle weakness can appear during disease progression. Treatment is often symptomatic and aimed at increasing comfort and decreasing pain, especially when the disease is rapidly progressive. For spasticity and dystonia the treatment is

largely similar to that of cerebral palsy. For some diagnoses promising options are available when early treatment is possible such as enzyme replacement, bone marrow transplantation or substrate inhibition. Early recognition of the disease and early start of these specific treatments may benefit outcome.⁴⁴

Diagnosis	Pathogenic features	Spasticity/ Dystonia
Leucodystrophy - Metachromatic leucodystrophy - Adrenoleucodystrophy - Vanishing white matter - Pelizeus-Merzbacher disease	White matter abnormalities due to demyeliniation caused by deficient activity of different lysosomal enzymes	Both
Rett's syndrome	Cerebral atrophy and hyperammonaemia	Both
Neuronal ceroid-lipofuscinosis	Failure of fatty oxidation	Both
Niemann-Pick-type C (dystonic lipidosis)	Defective cholesterol esterification, neuronal storage	Dystonia
Niemann Pick type A	Sphingomyelinase activity deficient, neuronal and visceral storage of sphingomyeline	Spasticity
Mucopolysaccharidosis	Deficient activity of specific lysosomal enzymes causing accumulation of acidic polysaccharides in various organs including the brain	Both
Lesh Nyhan Syndrome	Hypoxanthine-guanine phosphoribosyltransferase (HGPRT) deficiency	Both
Spinocerebaillar degeneration - Friedreich's ataxia - Spinocerebellar ataxia	Degeneration in both central and peripheral nervous system	Both
Hereditairy spastic paresis	Genetic cause, several known and unknown genes	Spasticity
Leigh-syndrome	Subacute necrotizing encephalomyopathy Autosomal recessive	Both

 Table 1. A selection of progressive neurodegenerative and neurometabolic diseases of

 childhood causing spasticity and/or dystonia⁴⁰⁻⁴³

TREATMENT OPTIONS

Since this thesis focusses on severely affected children with generalized spasticity or dyskinesia (GMFCS level IV or V for CP and equally affected for non-CP disorders), treatment options discussed below apply to this group of children and involve pharmacological and advanced (neurosurgical) treatment options. Focal treatment options are therefore not discussed.

Spasticity

Oral pharmacological treatment with baclofen, tizanidine, dantrolene or diazepam has been described in literature. There is insufficient data to make clear recommendations for oral pharmacological treatment since study numbers are low and the described populations small.⁴⁵ Baclofen is the agent most commonly prescribed in clinical practice, however, there is insufficient evidence for the use for either decreasing spasticity or improving motor function in patients with spastic cerebral palsy.^{45,46} The working mechanism of baclofen and intrathecal use are described later in this chapter.

A surgical option for the treatment of severe spasticity of the lower limbs is selective dorsal rhizotomy (SDR). SDR is mostly used in children with GMFCS I and II and aims to preserve or even improve mobility during growth.⁴⁷ In children with GMFCS IV and V daily comfort and care (dressing and washing) can improve after SDR.⁴⁸

Dyskinesia

Since dystonia is the dominant feature in dyskinetic CP, treatment is mostly aimed to decrease dystonia and not choreo-athetosis.³⁰ For generalized dystonia, the use of trihexyphenidyl, baclofen, benzodiazepines and other agents has been described in literature. The level of evidence is low and effects in studies are mostly disappointing and inconsistent.⁴⁹⁻⁵³

Intrathecal baclofen treatment (ITB) is an option in patients with severe dyskinesia in whom oral pharmacological treatment is insufficient or causes side effects such as sedation, prohibiting further increase to a therapeutic dosage. Another surgical treatment option for severe dystonia is deep brain stimulation (DBS). Good results are shown in pediatric patients with primary dystonia but the reported effects are variable in patients with dyskinetic cerebral palsy.^{50,54}

BACLOFEN

Baclofen is a gamma-aminobutyric-acid (GABA) B receptor agonist and binds on receptors on the pre- and post-synaptic neuron.^{55,56} The presynaptic action is likely to be the most dominant in decreasing spasticity.⁵⁶ By binding to the pre-synaptic receptors, baclofen inhibits the release of the excitatory neurotransmitter glutamate via downregulation of the calcium channels (Figure 5). Calcium uptake required for the release of the excitatory neurotransmitters is inhibited. Hereby neuronal signal transmission onto spinal motor neurons is prevented.^{55,56}

The site of action of baclofen in treatment of spasticity is thought to be the superficial layers of the dorsal spinal cord.⁵⁷ This is different for treatment of dystonia where baclofen is thought to act on the intracranial level^{55,58}. Clinical findings support this hypothesis: in contrast to spasticity, a single intrathecal lumbar bolus of baclofen does not decrease dystonia. It actually takes several days after continuous intrathecal infusion to see improvement of dystonia, time needed for the baclofen to spread to the intracranial level.⁵⁹

Dystonia in CP is the result of lesions in the basal ganglia causing disturbance of the indirect basal ganglia pathway. Lesions of the putamen and globus pallidus cause decreased inhibition of the thalamus. Subsequently the thalamus stimulates the supplementary motor and premotor cortex causing excessive movements (e.g. dystonia). Intrathecal baclofen can hypothetically act upon different intracranial levels. One idea is that by administrating baclofen, the external globus pallidus is exogenously suppressed, resulting in normalization of the indirect basal ganglia pathway with inhibition of the thalamus and thereby inhibition of stimulation of the supplementary motor cortex, decreasing dystonia.⁵⁵ Another working mechanism could be that GABA is an inhibitory neurotransmitter in the human cerebral cortex, ^{57,60} and that baclofen directly inhibits post-synaptic signal transmission from the cortex. A third option is that baclofen inhibits the thalamus. An overview of different working mechanisms is presented in figure 5.

Due to its low lipid solubility, orally administered baclofen does not pass the blood brain barrier well.⁵⁷ Plasma concentration levels ranging from 50 to 445 (ng/ml) render cerebrospinal fluid (CSF) concentration levels between <12 and 64 ng/ml. A higher plasma concentration does not correspond to a higher CSF concentration.⁶¹ This means that in many cases, high oral dosages are needed to acquire sufficient CSF concentrations. In severe cases the therapeutic effect remains insufficient while additional adverse effects, such as sedation, occur.^{57,61}

Intrathecal baclofen treatment

The blood brain barrier can be bypassed by delivering baclofen intrathecally using an implanted micro-infusion pump (Figure 6). Higher levels of CSF concentration with simultaneous very low plasma concentrations can be easily achieved without the limitative adverse effects seen with oral administration.^{57,62} Sedation for instance occurs in less than 5% of patients.⁵⁷



Figure 5. Baclofen working mechanism. On the spinal cord level baclofen binds to the presynaptic GABA_B receptors and hereby inhibits the release of the neurotransmitter glutamate. As a consequence, neuronal transmission onto the lower motor neurons is inhibited. It also binds to the postsynaptic GABA_B receptors inhibiting the signal transmission in the lower motor neuron. This is the working mechanism in the treatment of spasticity (lower right image). On the cerebral level baclofen inhibits the excessive stimulation of the cortex by inhibiting the globus pallidus, thalamus or supplementary motor and premotor cortex (upper right image). This is the presumed working mechanism of intrathecal baclofen in the treatment of dystonia.

To provide the baclofen intrathecally, a micro-infusion pump (Medtronic Synchromed II) is implanted subcutaneously, mostly in the left lower abdomen (Figure 6). A catheter connects the pump with the intrathecal space providing the baclofen at the site of action: the spinal cord for spastic patients and at the intracranial level for dyskinetic patients.⁵⁸ Since there is a significant reduction of concentration of drugs in the CSF cranially along the spinal canal from the place of insertion, the tip of the catheter is placed at the low (effects on legs) or mid (effects on arms and legs) thoracic level in spastic patients and at the cervical level for dyskinetic patients.^{58,63-65} Intraventricular baclofen administration has also been described in patients with dyskinetic CP.⁶⁶⁻⁶⁸ Once the pump is implanted, the dosage can be adjusted to the patients' needs using an external pump programmer. The effective therapeutic dosage varies enormously between patients and is not predictable: a low dosage for one patient can cause

symptoms of overdose in another. The pump has to be refilled approximately every 2 to 6 months by puncture through the skin and a small membrane located in the center of the pump. After approximately 7 years, the battery expires and the pump has to be replaced.



Figure 6. The intrathecal baclofen pump

A. The baclofen pump (Medtronic, Synchromed II) and catheter. Source: www.medtronic.com B. X-ray of an implanted pump in the abdomen of a 10 year old boy with cerebral palsy, frontal and lateral view.

The most common complications of ITB are infection (14-28%), CSF leak (33%, with 8% requiring surgery) and catheter related problems (including catheter migration, occlusion and disconnection) (21-23%).^{58,69,70} Signs of underdosing or acute withdrawal can be symptoms of catheter related problems but the cause can also be iatrogenic. Overdose is mostly iatrogenic due to wrong pump settings or inadequate pump filling.^{58,69,70}

In 2010 a consensus was reached on the selection criteria for ITB for patients with spasticity (Table 2).⁷¹ There are no specific selection criteria formulated for the treatment of dystonia and in clinical practice selection criteria analogous to those for spastic CP can be used.

Table 2. Selection criteria for ITB

- Intractable spasticity or dystonia
 - o With preferably known etiology
 - o Not optimally managed by other treatment options
 - o Which impedes activities of daily life or quality of life
- Clear and realistic treatment goals
- Ability and motivation to attend follow-up and monitoring
- Sufficient body size to allow pump implantation and filling

The italic written criteria are criteria not specifically described in the consensus but are criteria that are used additionally in our clinical practice.

The efficacy of intrathecal baclofen treatment

Spastic cerebral palsy

Spasticity is the most common outcome measure on the level of body functions and structures. A few placebo controlled double blind randomized trials looking at the effect of ITB test treatment with multiple single bolus injections, show a significant decrease of spasticity after ITB injections.⁷²⁻⁷⁴ Several other studies with various designs (prospective and retrospective, small cohort or case series studies, some with and some without a control group awaiting ITB treatment) report a significant decrease in spasticity after continuous ITB treatment via an implanted pump (follow up ranging from six months to nine years).^{38,73-80}

Pain and discomfort are problems which are often reported in children with severe spastic CP. The same studies as described above report improvement of pain and comfort during ITB test treatment.^{72,74} Similar results are found with long term continuous ITB treatment.⁷⁸⁻⁸⁰ Quality of life is assessed in several of the described studies. Various questionnaires are used and all studies report improvement of quality of life.^{38,75,79,80}

Outcomes on the level of activities and participation are often not systematically measured. Furthermore, outcome measures are heterogeneous, making comparison between studies difficult. Individually formulated problems of daily life measured on a visual analogue scale (VAS) are described to significantly improve after ITB bolus test treatment compared to placebo, as did VAS scores for ease of care.⁷² Prospective cohort studies show that improvements maintain after 6 months, 12 months, and even after 6 to 9 years of continuous ITB treatment.⁷⁸⁻⁸⁰ In individual cases, improvements were noted on transfers, operating the electric wheelchair and arm function.⁷² In a telephone

survey, parents indicated that seventy-two percent of pre-set goals were completely achieved during ITB treatment.³⁸

Active functioning and social participation are found not to improve in children with GMFCS IV and V during ITB treatment.⁷⁶ On global scales of functioning, such as the Gross Motor Function Measure (GMFM) and the Pediatric Evaluation of Disability Inventory (PEDI), no changes are reported with ITB.^{38,76,78} For severely affected GMFCS IV and V patients, no significant changes on standardized questionnaires are expected since they do not adequately meet the functional level of these children.

Taking care of a child with severe disability puts a burden on the caregivers (level of environmental factors) and a prospective cohort study shows that with long term ITB treatment, caregivers report to experience fewer emotional concerns and fewer limitation in time.⁸⁰ One other study reported no change in the impact of disability for children and parents.⁷⁶ Caregiver satisfaction is high, and most of them would choose for ITB treatment again.^{38,75,80}

Dyskinetic cerebral palsy

For the effect of ITB in children with dyskinetic CP few studies are available. Studies contain only prospective and retrospective case series, resulting in a low level of evidence for the effect of ITB in these patients.^{58,81,82}

On the level of body functions and structures dystonia (BADS) is the most common outcome measure. BADS scores are reported to decrease in all studies.^{58,81,82} A significantly higher decrease in BADS scores was seen with a higher placed catheter tip (T4 and higher) compared to a lower placed tip (T6 and lower).⁵⁸ Other reported outcomes on the level of body functions and structures are reduction in pain, better mood, improved sleep, and a stable range of motion. ^{58,81,82} Quality of life improved in 86% of the questioned patients.^{58,81,82}

On the level of activities and participation several case series report the outcomes of structured interviews or questionnaires.^{58,81,82} Improvement is reported for feeding and swallowing, sitting and posture control, upper limb use, communication/speech. In the majority of patients no change in autonomy of carrying out daily activities was seen.⁸² Furthermore, there was no change in gross motor function.⁸¹

On the level of environmental factors (e.g. caregiving by others), ease of care is reported to improve in most patients.^{58,81,82}

Treatment goals are fully reached in 76% and partly in the remainder of patients.⁸¹ Almost 80% of patients were satisfied with ITB treatment and 74% of patients would choose for ITB treatment again.⁸²

Progressive neurological disease in childhood

ITB has been used in clinical practice for the treatment of spasticity or dystonia in patients with PND. Treatment goals are mostly to improve comfort and ease of care. In literature, reports on the effect of ITB in children with progressive neurologic disease are very rare and the few existing papers describe mostly single cases. Spasticity, spasms and pain seem to decrease and activities of daily life are reported to improve.⁸³⁻⁸⁸ It is however unclear if the effects of ITB maintain during disease progression.

CONCLUSION

Treatment of disabling movement disorders such as spasticity or dyskinesia in children is challenging. Oral pharmacological treatment options are often insufficient and/or cause side effects. Intrathecal baclofen treatment might be an option for these patients. There is some evidence, provided by single bolus randomized trials, for the effectiveness of ITB for treatment of spasticity in children with spastic cerebral palsy on the level of body functions and structures (spasticity). There is a low level of evidence of the effectiveness of continuous ITB, especially on the level of activities/participation. For dystonia in children with dyskinetic CP, there is no substantial evidence for the effect of ITB treatment on any ICF-CY level. For the treatment of spasticity and/or dystonia, not due to CP but due to PND, it is unclear what the effect is and what the evidence for the effect is, especially on the longer term when disease progresses.

AIM OF THE STUDY

- 1. The primary aim of this thesis is to investigate the effect of ITB in the treatment of dyskinetic CP. The focus lies mostly on the level of activities and participation and the level of environmental factors (attainment of individual treatment goals) and secondary on the level of body functions and structures (dystonia, choreoathetosis, spasticity, pain, comfort).
- 2. The secondary aim is to describe the effect and the current level of evidence of ITB treatment in PND of childhood on all ICF-CY levels.

OUTLINE OF THIS THESIS

Chapter 2 provides a narrative review on the presentation and management of dyskinetic CP. Chapter 3 describes the results of a pilot study comparing ITB and placebo bolus dosing in patients with dyskinetic CP. Individual problem of daily life were individually formulated and scored on a VAS. Dystonia was measured using the BADS. Chapter 4 contains the study protocol of the IDYS trial (Intrathecal baclofen treatment in Dystonic cerebral palsy). This study is a multi-center, placebo-controlled, double-blind randomized trial. The primary aim was to determine the effects of ITB in dyskinetic CP mostly on the level of activities and participation, and environmental factors, but also on the level of body functions and structures (goal attainment scaling of individual treatment goals). Secondary outcomes are on the level of body functions and structures (dystonia, spasticity, pain, comfort). Chapter 5 shows the results of the IDYS trial. Chapter 6 provides a systematic review of literature on the reported effects of ITB in PND of childhood. Chapter 7 shows the results of a questionnaire about satisfaction of ITB, conducted in patients with PND and their caregivers. Satisfaction was scored using a visual analogue scale (VAS). In Chapter 8 the similarities and differences in the effect of ITB between spastic CP, dyskinetic CP and PND are described. A questionnaire addressing the level of activities and participation was used. The effects were scored using a 5 point Likert Scale (much better, somewhat better, no change, somewhat worse, much worse). Chapter 9 discusses the overall findings of this thesis. The aims of the thesis as described above are reviewed in this chapter and suggestions for further studies are done.

REFERENCES

- Graham HK, Rosenbaum P, Paneth N, Dan B, Lin JP, Damiano DL, et al. Cerebral palsy. Nat Rev Dis Primers. 2016;2.
- Ivanhoe CB, Reistetter TA. Spasticity: The misunderstood part of the upper motor neuron syndrome. Am J Phys Med Rehabil. 2004;83(suppl):S3-S9.
- Delgado MR, Albright AL. Movement disorders in children: definitions, classifications, and grading systems. J Child Neurol. 2003;18:S1-S8.
- Organization WH. International classification of functioning, disability and health: children & youth version: ICF-CY. 2007.
- Palisano R, Rosenbaum P, Walter S, Russell D, Wood E, Galuppi B. Development and reliability of a system to classify gross motor function in children with cerebral palsy. Dev Med Child Neurol. 1997;39:214-23.
- Oskoui M, Coutinho F, Dykeman J, Jetté N, Pringsheim T. An update on the prevalence of cerebral palsy: a systemetic review and meta-analysis. Dev Med Child Neurol. 2013;55:509-19.
- Himmelmann K, Uvebrant P. The panorama of cerebral palsy in Sweden part XII shows that patterns changed in the birth years 2007-2010. Accta Paediatr. 2018;107(3):462-8.
- Himmelmann K, Beckung E, Hagberg B, Uvebrant P. Gross and fine motor function and accompanying impairments in cerebral palsy. Dev Med Child Neurol. 2006;48(6):417-23.
- Odding E, Roebroeck ME, Stam HJ. The epidemiology of cerebral palsy: indicence, impairments and risk factors. Disability and Rehabilitation. 2006;28(4):183-91.

- Rosenbaum P, Paneth N, Leviton A, Goldstein M, Bax M, Damiano D, et al. A report: the definition and classification of cerebral palsy April 2006. Dev Med Child Neurol. 2007;109 (Suppl):8-14.
- Himmelmann K, Hagberg G, Beckung E, Hagberg B, Uvebrant P. The changing panorama of cerebral palsy in Sweden. IX. Prevalence and origin in the birthyear period 1995-1998. Acta Paediatrica. 2005;94:287-94.
- Bax M, Tydeman C, Flodmark O. Clincal and MRI correlates of cerebral palsy. JAMA. 2006;296(13):1602-8.
- Krägeloh-Mann I, Hagberg G, Meisner c, Haas G, Eeg-Ologsson KE, Selbmann HK, et al. Bilateral spastic cerebral palsy - a collaborative study between southwest Germany and western Sweden. III. Aetiology. Dev Med Child Neurol. 1995;37:191-203.
- Krägeloh-Mann I. Imaging of early brain injury and cortical plasticity. Experimental Neurology. 2004;190:S84-S90.
- Volpe J. Brain injury in premature infants: a complex amalgam of destructive and developmental disturbances. Lancet Neurol. 2009;8(1):110-24.
- de Vries LS. Neurological assessment of the preterm infant. Accta Paediatr. 1996;85:765-71.
- 17. Himmelmann K, Uvebrant P. The panorama of cerbral palsy in Sweden. XI. Changing patterns in the birth-year period 2003-2006. Acta Paediatrica. 2014;103:618-24.
- Krägeloh-Mann I, Helber A, Mader I, Staudt M, Wolff M, Groenendaal F, et al. Bilateral lesions of thalamus and basal ganglia: origin and outcome. Dev Med Child Neurol. 2002;44:277-484.

- Aravamuthan BR, Waugh JL. Localization of basal ganglia and thalamic damage in dyskinetic cerebral palsy. Pediatr Neurol. 2016;54:11-21.
- 20. Geytenbeek JJ, Vermeulen RJ, Becher JG, Oostrom KJ. Comprehension of sponken language in non-speaking children with severe cerebral palsy: an explorative study on associations with motor type and disabilities. Dev Med Child Neurol. 2015;57(3):294-300.
- Choi JY, Choi YS, Rha D, Park ES. The clinical outcomes of deep gray matter injury in children with cerebral palsy in relation with brain magnetic resonance imaging. Res Dev Disabil. 2016;55:218-25.
- 22. Olds C, Oghalai JS. Bilirubin-induced audiologic injury in preterm infants. Clin Perinatol. 2016;43(2):313-23.
- 23. Rose J, Vassar R. Movement disorders due to bilirubin toxicity. Semiin Fetal Neonatal Med. 2015;20(1):20-5.
- 24. Govaert P, Lequin M, Swarte R, Robben S, de Coo R, Weisglas-Kuperus N, et al. Changes in globus pallidus with (pre)erm kernicterus. Pediatrics. 2003;112:1256-63.
- 25. Eliasson AC, Krumlinde-Sundholm L, Rösblad B, Beckung E, Arner M, Ohrvall AM, et al. The Manual Ability Classification System (MACS) for children with cerebral palsy: scale development and evidence of validity and reliability. Dev Med Child Neurol. 2006;48(7):549-54.
- 26. Hidecker MJ, Paneth N, Rosenbaum PL, Kent RD, Lillie J, Eulenberg JB, et al. Developing and validating the Communication Function Classification System for individuals with cerebral palsy. Dev Med Child Neurol. 2011;53(8):704-10.
- Sanger TD, Delgado MR, Gaebler-Spira D, Hallett M, Mink JW. Classification and definition of disorders causing hypertonia in childhood. Pediatrics. 2003;111(1):e89e97.

- Krägeloh-Mann I, Petruch U, Weber PM.
 SCPE Reference and Training Manual (R&TM). Grenoble: Surveillance of Cerebral Palsy in Europe; 2005.
- 29. Sanger TD, Chen D, Fehlings DL, Hallett M, Lang AE, Mink JW, et al. Definition and Classification of Hyperkinetic Movements in Childhood. Mov Disord. 2010;25(11):1538–49.
- Monbaliu E, de Cock P, Ortibus E, Heyrman L, Klingels K, Feys H. Clinical patterns of dystonia and choreoathetosis in participants with dyskinetic cerebral palsy. Dev Med Child Neurol. 2016;58(2):138-44.
- 31. Monbaliu E, De Cock P, Mailleux L, Dan B, Feys H. The relationship of dystonia and choreoathetosis with activity, participation and quality of life in children and youth with dyskinetic cerebral palsy. European Journal of Paediatric Neurology. 2017;21(2):327-35.
- 32. Stewart K, Tavender E, Rice J, Harvey A. Identification, classification and assessment of dyskinesia in children with cerebral palsy: A survey of clinicians. J Paediatr Child Health. 2017.
- Barry MJ, VanSwearingen JM, Albright AL. Reliability and responsiveness of the Barry-Albright Dystonia Scale. Dev Med Child Neurol. 1999;41:404-11.
- 34. Monbaliu E, Ortibus E, de Cat J, Dan B, Heyrman L, Prinzie P, et al. The dyskinesia Impairment Scale: a new instrument to measure dystonia and choreoathetosis in dyskinetic cerebral palsy. Dev Med Child Neurol. 2012;54:278-83.
- 35. Stewart K, Harvey A, Johnston LM. A systematic review of scales to measure dystonia and choreoathetosis in children with dyskinetic cerebral palsy. Dev Med Child Neurol. 2017;59(8):786-95.

- Himmelmann K, McManus V, Hagberg G, Uvebrant P, Krägeloh-Mann I, Cans C, et al. Dyskinetic cerebral palsy in Europe: trends in prevalence and severity. Arch Dis Child. 2009;94:921-6.
- Himmelmann K, Hagberg G, Wiklund LM, Eek MN, Uvebrant P. Dyskinetic cerebral palsy: a population-based study of children born between 1991 and 1998. Dev Med Child Neurol. 2007;49:246-51.
- Campbell WM, McLaughlin JF, Grant GA, Loeser JD, Graubert C, Bjornson K. Longterm safety and efficacy of continous intrathecal baclofen. Dev Med Child Neurol. 2002;44:660-5.
- Liew PY, Stewart K, Khan D, Arnup SJ, Scheinberg A. Intrathecal baclofen therapy in children: an analysis of individualized goals. Dev Med Child Neurol. 2018;60(4):367-73.
- 40. Klein C, Münchau A. Progressive dystonia. Handb Clin Neurol. 2013;113:1889-197.
- 41. Brett EM. Paediatric Neurology. New York: Pearon Professional Limited; 1997.
- 42. Dyken P, Krawiecki N. Neurodegenerative diseases of infancy and childhood. Ann Neurol. 1983;13(4):351-64.
- Karimzadeh P. Approach to neurometabolic diseases from a pediatric neurologiccal point of view. Iran J Child Neurol. 2015;9(1):1-16.
- Müller vom Hagen J, Karle KN, Schüle R, Krägeloh-Man I, Schöls L. Leukodystrophies underlying cryptic spastic paraparesis: frequency and phenotype in 76 patients. European Journal of Neurology. 2014;21:983-8.
- 45. Delgado MR, Hirtz D, Aisen M, SAshwal S, Fehlings DL, McLaughlin J, et al. Practice parameter: pharmacologic treatment of spasticity in children andd adolescents with cerebral palsy (an evidence-based) review. Neurology. 2010;74:336-43.

- Benini R, Shevell MI. Updates in the treatment of spasticity associated with cerebral palsy. Curr Treat Options Neurol. 2012;14(6):650-9.
- 47. Bolster EAM, van Schie PEM, Becher JG, van Ouwerkerk WJR, Strijers RLM, Vermeulen RJ. Long-term effect of selective dorsal rhizotomy on gross motor function in ambulant children with spastic bilateral cerebral palsy, compared with reference centiles. Dev Med Child Neurol. 2013;5:610-6.
- 48. Buizer AI, van Schie PEM, Bolster EAM, van Ouwerkerk WJR, Strijers RL, van de Pol LA, et al. Effect of selective dorsal rhizotomy on daily care and comfort in non-walking children and adolescents with severe spasticity. Eur J Paediatr Neurol. 2017;21:350-7.
- 49. Jankovic J. Medical treatment of dystonia. Movement Disorders. 2013;28(7):1001-12.
- 50. Fehlings D, Brown L, Harvey A, Himmelmann K, Lin JP, Macintosch A, et al. Pharmacological and neurosurgical interventions for managing dystonia in cerebral palsy: a systematic review. Dev Med Child Neurol. 2018;epub before print.
- 51. Masson R, Pagliano E, Baranello G. Efficacy of oral pharmaccological treatments in dyskinetic cerebral palsy: a systematic review. Dev Med Child Neurol. 2017;59:1237-48.
- 52. Sanger TD, Bastian A, Brunstrom J, Damiano D, Delgado M, Dure L, et al. Prospective open-label clinical trial of trihexyphenidyl in children with secondary dystonia due to cerebral palsy. J Child Neurol. 2007;22(5):530-7.
- 53. Vidailhet M. Treatment of movement disorders in dystonia-choreoathtosis cerebral palsy. Handb Clin Neurol. 2013;111:197-202.

- 54. Elia AE, Bagella C, Ferre F, Zorzi G, Calandrella D, Romito LM. Deep brain stimulation for dystonia due to cerebral palsy: a review. Eur J Paediatr Neurol. 2018;22:308-15.
- 55. Albright AL. Intrathecal baclofen in cerebral palsy movement disorders. Journal of Child Neurology. 1996;11(suppl 1):S29-S35.
- 56. Brennan PM, Whittle IR. Intrathecal baclofen therapy for neurological disorders: a sound knowledge base but many challenges remain. Britisch Journal of Neurosurgery. 2008;22(4):508-19.
- Albright AL. Baclofen in the treatment of cerebral palsy. J Child Neurol. 1996;11:77-83.
- Albright AL, Barry MJ, Shagron DH, Ferson SS. Intrathecal baclofen for generalized dystonia. Dev Med Child Neurol. 2001;43:652-7.
- 59. Albright AL, Barry MJ, Fasick P, Barron W, Schulz B. Continous intrathecal baclofen infusion for symptomatic generalized dystonia. Neurosurg. 1996;38(5):934-9.
- McCormick DA. GABA as an inhibitory neurotransmitter in human cerebral cortex. J Neurophysiol. 1989;62(5):1018-27.
- 61. Knutsson E, Lindblom U, Martensson A. Plasma and cerebrospinal fluid levels of baclofen (Lioresal[®]) at optimal therapeutic responses in spastic paresis Journal of the Neurological Sciences. 1974;23:473-84.
- Albright AL, Shultz BL. Plasma baclofen levels in children receiving continuous intrathecal baclofen infusion. Journal of Child Neurology. 1999;14:408-9.
- 63. Kroin JS, Ali A, York MRN, Penn RD. The distribution of medication along the spinal canal after chronic intrathecal admission. Neurosurg. 1993;33(2):226-30.

- 64. Vender JR, Hester S, Waller JL, Rekito A, Lee MR. Identification and management of intrathecal baclofen pump complications: a comparison of pediatric and adult patients. J Neurosurg: Pediatrics. 2006;104:9-15.
- Albright AL, Turner M, Pattisapu JV. Best-practice surgical techniques for intrathecal baclofen therapy. J Neurosurg. 2006;104:233-9.
- Albright AL, Ferson SS. Intraventricular baclofen for dystonia: techniques and outcomes. J Neurosurg Pediatrics. 2009;3(1):11-4.
- 67. Rocque BG, Albright AL. Intraventricular versus intrathecal baclofen for secondary dystonia: a comparison of complications. Neurosurg. 2012;70(2 Suppl Operative):325-6.
- Bollo RJ, Gooch JL, Walker ML. Stereotactic endoscopic placement of third ventricle catheter for long-term infusion of baclofen in patients with secondary generalized dystonia. J Neurosurg Pediatrics. 2012;10(1):30-3.
- 69. Gooch JL, Oberg WA, Grams B, Ward LA, Walker ML. Complications of intrathecal baclofen pumps in children. Pediatr Neurosurg. 2003;39:1-6.
- Ward A, Hayden S, Dexter M, Scheinberg A. Continuous intrathecal baclofen for children with spasticity and/or dystonia: Goal attainment and complications associated with treatment. Journal of Paediatrics and Child Health. 2009;45:720-6.
- Dan B, Motta F, Vles JSH, Vloeberghs M, Becher JG, Euson P, et al. Consensus on the appropriate use of intrathecal baclofen (ITB) therapy in paediatric spasticity. Eur J Paediatr Neurol. 2010;14:19-28.

- 72. Hoving MA, Van Raak EPM, Spincemaille GHJJ, Palmans LJ, Sleypen FAM, Vles JSH. Intrathecal baclofen in children with spastic cerebral palsy: a doubleblind, randomized, placebocontrolled, dosefinding study. Dev Med Child Neurol. 2007;49:654-9.
- Albright AL, Cervi A, Singletary J. Intrathecal baclofen for spasticity in cerebral palsy. JAMA. 1991;265(11):1418-22.
- 74. van Schaeybroeck P, Nuttin B, Lagae L, Schrijvers E, Borghgraef C, Feys P. Intrathecal baclofen for intractable ccerebral spasticity: a prospective placebo-controlled, double-blind study. Neurosurg. 2000;46(3):603-12.
- 75. Kraus T, Gegenleitner K, Svehlik, Novak M, Steinwender G, Singer G. Long-term therapy with intrathecal bacclofen improves quality of life in children with severe spastic cerebral palsy. Eur J Paediatr Neurol. 2017;21:565-9.
- 76. Morton RE, Gray N, Vloeberghs M. Controlled study of the effects of continuous intrathecal baclofen infusion in non-ambulant children with cerebral palsy. Dev Med Child Neurol. 2011;53(8):736-41.
- Murphy NA, Nicole Irwin MC, Hoff C. Intrathecal baclofen therapy in children with cerebral palsy: efficcacy and complications. Arch Phys Med Rehabil. 2002;83:1721-5.
- 78. Hoving MA, Van Raak EPM, Spincemaille GHJJ, Palmans LJ, Becher JG, Vles JSH, et al. Efficacy of intrathecal baclofen therapy in childen with intractable spastic cerebral palsy: a randomised controlled trial. Eur J Paediatr Neurol. 2009;13:240-6.

- 79. Hoving MA, Van Raak EPM, Spincemaille GHJJ, Van Kranen-Mastenbroek VHJM, van Kleef M, Gorter JW, et al. Safety and one-year efficacy of intrathecal baclofen therapy in children with intractable spastic cerebral palsy. Eur J Paediatr Neurol. 2009;13:247-56.
- 80. Vles GF, Soudant D, Hoving MA, Vermeulen RJ, Bonouvrie LA, van Oostenbrugge RJ, et al. Long-term follow-up of continous intrathecal baclofen therapy in nonambulant children with intractable spastic cerebral palsy. Eur J Paediatr Neurol. 2013;17:639-44.
- Eek MN, Olsson K, Lindh K, Askljung B, Pahlman M, Corneliusson O, et al. Intrathecal baclofen in dyskinetic cerebral palsy: effects on function and activity. Dev Med Child Neurol. 2018;60(1):94-9.
- 82. Motta F, Stignani C, Antonello CE. Effects of intrathecal baclofen on dystonia in children with cerebral palsy and the use of functional scales. Journal of Pediatric Orthopaedics. 2008;28:213-7.
- Ben Smail D, Jacq C, Denys P, Bussel B. Intrathecal baclofen in the treatment of painful, disabling spasms in Freidreichs ataxia. Mov Disord. 2005;20(6):758-9.
- Chu MLY, Sala DA, Weiner HL. Intrathecal baclofen in X-linked adrenoleukodystrophy. Pediatr Neurol. 2001;24(2):156-8.
- 85. Dan B, Cheron G. Intrathecal baclofen normalizes motor startegy for squatting in familial spastic paraplegia: a case study. Neurophysiol Clin. 2000;30(1):43-8.
- 86. Hjartarson HT, ehrstedt C, Tedroff K. Intrathecal baclofen treatment an option in X-linked adrenoleukodystrophy. Eur J Paediatr Neurol. 2018;22:178-81.
- Kadyan V, Clairmont AC, George RJ, Johnson EW. Intrathecal baclofen for spasticity management in Rett syndrome. Am J Phys Med Rehabil. 2003;82(7):560-2.

Chapter 1

 Molteni F, Carda S, Cazzaniga M, Magoni L, Rossini M, Caimmi M. Instrumental evaluation of gait modifications before and during intrathecal baclofen therapy: a 2-year follow-up case study. Am J Phys Med Rehabil. 2005;84(4):303-6.
CHAPTER 2

Clinical presentation and management of dyskinetic cerebral palsy

E Monbaliu, K Himmelmann, J-P Lin, E Ortibus, L Bonouvrié, H Feys, RJ Vermeulen, B Dan

Lancet Neurology 2017; 16:741-749



ABSTRACT

Cerebral palsy is the most frequent cause of severe physical disability in childhood. Dyskinetic cerebral palsy (DCP) is the second most common type of cerebral palsy after spastic forms. DCP is typically caused by non-progressive lesions to the basal ganglia or thalamus, or both, and is characterised by abnormal postures or movements associated with impaired tone regulation or movement coordination. In DCP, two major movement disorders, dystonia and choreoathetosis, are present together most of the time. Dystonia is often more pronounced and severe than choreoathetosis, with a major effect on daily activity, quality of life, and societal participation. The pathophysiology of both movement disorders is largely unknown. Some emerging hypotheses are an imbalance between indirect and direct basal ganglia. Rehabilitation strategies are typically multidisciplinary. Use of oral drugs to provide symptomatic relief of the movement disorders is limited by adverse effects and the scarcity of evidence that the drugs are effective. Neuromodulation interventions, such as intrathecal baclofen and deep brain stimulation, are promising options.

INTRODUCTION

Cerebral palsy comprises a group of developmental disorders of movement and posture, attributed to non-progressive disturbances that occurred in the developing fetal or infant brain. The motor impairments are often accompanied by disturbances of sensation, perception, cognition, communication, and behaviour, by epilepsy, and by secondary musculoskeletal problems.¹

Cerebral palsy is the most common cause of severe physical disability in early childhood, with a prevalence of 1.7–3.1 per 1000 livebirths in high-income countries, and higher prevalence in low-income countries.^{2,3} Cerebral palsy is clinically categorised into spastic, dyskinetic, and ataxic cerebral palsy on the basis of the predominant motor disorder.¹ Dyskinetic cerebral palsy (DCP) is characterised by abnormal postures or movements associated with impaired muscle tone regulation, movement control, and coordination comprising two major movement disorder patterns: dystonia and choreoathetosis.⁴ DCP is sometimes referred to as dystonic, athetoid, extrapyramidal, choreoathetotic, or choreoathetoid cerebral palsy. DCP accounts for up to 15% of cerebral palsy cases, making it the second most common type after spastic cerebral palsy.^{2,5}

DCP can have many causes, including perinatal hypoxia–ischaemia in infants born near to term, neonatal hyperbilirubinaemia, brain maldevelopment, intracranial haemorrhage, stroke, or cerebral infection.⁵ Hyperbilirubinaemia-induced kernicterus (ie, deposition of bilirubin in the basal ganglia) can occur during the preterm and term period. This condition has become less common in high-income countries because of preventive strategies, but is still a notable issue in low-income countries. In addition to primary prevention of hyperbilirubinaemia, postnatal monitoring remains essential, and promising prevention approaches have been developed in resource-constrained settings.⁶ Causes outside the neonatal period, such as cardiorespiratory arrest or near drowning in the first year, are rarer.

We critically reviewed the literature to provide an update on the diagnosis and management of DCP, and identify research priorities. This Review will focus on the clinical presentation, the management, and emerging therapeutic approaches of this motor disorder.

CLINICAL PRESENTATION

Motor impairments are often more severe in people with DCP than in patients with the other types of cerebral palsy.⁷ Non-motor comorbidities are also common; more than half of patients with DCP have a combination of severe intellectual impairment, anarthria, and epilepsy.^{4,5,7} Visual and hearing impairments are also common.⁵ Despite normal birthweight in most cases, more than half of patients became underweight at follow-up in a population-based study from western Sweden, presumably as a result of high energy expenditure due to involuntary movements combined with poor feeding, dysphagia, gastro-oesophageal reflux, and suboptimal nutritional intake.^{7,8} Sleep and respiratory function are often disturbed. Drooling, dental problems, constipation, faecal and urinary incontinence, pain, and musculoskeletal deformities⁹ (including scoliosis) are common, and degenerative cervical changes causing myelopathy and motor deterioration have been described in adults.¹⁰ Mental health problems, including depression, are increasingly recognised. The risk for early death in these patients is higher than in other types of cerebral palsy,¹¹ and in the general population, most commonly through respiratory failure due to aspiration and pneumonia.

Dystonia and choreoathetosis

Dystonia and choreoathetosis are present in most DCP cases (figure 1; videos 1–4),^{4,12} but they can be identified as distinct movement disorders.^{1,13,14} Both movement disorders occur independently, but dystonia predominates in most patients with DCP.⁴ In childhood, dystonia refers to abnormal postures, involuntary twisting, and repetitive movements due to sustained or intermittent muscle contractions.^{11–13} Choreoathetosis in cerebral palsy is characterised by hyperkinesia and muscle tone fluctuation. Although choreoathetosis can be separated into chorea (ie, rapid, involuntary, jerky, and often fragmented movements) and athetosis (ie, slower, constantly changing, writhing, or contorting movements), this distinction does not appear to be clinically useful in people with DCP.^{13,14}

Dystonia and choreoathetosis are complex and difficult to measure. However, reliable measurement is crucial for the characterisation of clinical patterns and evaluation of the effects of targeted management. Clinical measures can be used for discriminative, predictive, or evaluative purposes. The Hypertonia Assessment Tool^{15,16} distinguishes dystonia from other hypertonic movement disorders, such as spasticity and rigidity. However, the current evidence base is insufficient to precisely differentiate between different types of cerebral palsy or between motor disorders, such as dystonia and choreoathetosis, in so-called pure DCP and spasticity. Dystonia can be exacerbated by non-specific stimuli, including emotion, cognitive tasks, stress, pain, and the

intention to move, and is always relieved by sleep; by contrast, spasticity is a velocitydependent increase in muscle tone.¹⁷ To date, no established clinical measure can discriminate between hyperkinetic movement disorders. Therefore, good familiarity with the operational consensus definitions is crucial for the reliable recognition of choreoathetosis and other hyperkinetic movement disorders. Several video-based measurement scales can be used to evaluate the severity of dystonia in DCP, such as the Burke-Fahn-Marsden Dystonia Rating Scale (BFMS),¹⁸ the Barry-Albright Dystonia Rating Scale,¹⁹ and the Dyskinesia Impairment Scale.²⁰ The BFMS and Barry- Albright Dystonia Rating Scale show good reliability in DCP but have limited sensitivity.²¹ These limitations and the absence of choreoathetosis measurements are resolved in the Dyskinesia Impairment Scale (DIS), which was specifically developed for the assessment of people with DCP.^{20,22} The DIS consists of dystonia (DIS-D) and choreoathetosis (DIS-CA) subscales that determine the presence of and rate the severity (amplitude and duration) of either movement disorder in various regions of the body during activity and rest. The BFMS and DIS are used as outcome measures in intervention studies, such as those investigating deep brain stimulation (DBS)²³ or intrathecal baclofen.²⁴

Dystonia is often more noticeable and severe than choreoathetosis.⁴ Both are generalised over all body regions (arms, legs, trunk, neck, mouth, and eyes), with higher severity in the upper limbs than in the lower limbs, and both substantially increase with activity. Dystonia has a major effect on the daily activity, quality of life, and societal participation of individuals with DCP.^{8,25} Dystonia particularly affects posture, mobility, hand and oral-motor function, and—to a lesser extent—non-verbal communication.⁴ By contrast, no functional associations with choreoathetosis have been found, which suggests that the sustained muscle contractions typical of dystonia restrict functional abilities to a much greater extent than the hyperkinetic hallmarks of choreoathetosis.⁴ Occasionally, status dystonicus can occur, which is characterised by prolonged or increasingly frequent generalised dystonia and requires early detection and urgent management.²⁶

Around 70% of patients with DCP have lesions in the basal ganglia or thalamus, or both on MRI (figures 2, 3); other brain lesions can also occur and few patients have apparently normal scans.^{4,7,28–30} The basal ganglia and thalamic lesions can be associated with the higher vulnerability of these regions because of high metabolic demand during the late third trimester of pregnancy²⁷ or the perinatal period. In neonates with kernicterus, often associated with high-frequency deafness, the globus pallidus is usually involved,³¹ but brain lesions are not always seen in the years following development of kernicterus. More severe presentations in DCP include cortical-subcortical involvement with increased dystonia and spasticity.^{27,32} A pattern of increasing clinical severity has been described based on an MRI classification system.³³ MRI does not necessarily detect the full extent of the affected brain region, therefore research with more advanced imaging techniques is recommended; however, imaging is difficult to perform in these patients because of the involuntary movements and postures.

At present, it is unclear how brain abnormalities produce dystonia and choreoathetosis. Emerging hypotheses based on inherited or primary (mainly focal) dystonia studies highlight three pathophysiological aspects, namely loss of inhibition, sensory dysfunction, and impaired plasticity in basal ganglia circuits.³⁴ Briefly, two basal ganglia pathways exist: a direct, excitatory pathway (from the striatum, through the globu-s pallidus internus, and to the thalamus) and an indirect, inhibitory pathway (from the striatum, through the globus pallidus externus, the subthalamic nucleus, and the globus pallidus internus to the thalamus). The direct pathway controls voluntary movements, whereas the indirect pathway inhibits unwanted movements. Imbalance between the direct and indirect pathways, for example because of overactivity in the direct pathway or underactivity in the indirect pathway, causes excessive movement and a loss of inhibition.³⁵

Abnormal sensorimotor integration is a key feature in the pathogenesis of many movement disorders,³⁶ and the basal ganglia might function as a gate for sensory input. Abnormal plasticity during motor learning might lead to abnormal sensorimotor integration, resulting in consolidation of abnormal motor engrams.³⁴ According to this hypothesis, therapeutic management might have little immediate effect on dystonia because so-called bad motor memories are difficult to erase.34 This hypothesis suggests that living longer with dystonia makes clinical management more difficult.37 However, based on the different underlying causes of the condition, whether this hypothesis also applies to dystonia in DCP is questionable.

A study in patients with DCP found no significant association between the severity of dystonia and brain lesions in the basal ganglia and thalamus.⁴ Some findings suggest that different brain regions that are involved in motor control can cause or contribute to dystonia, including the cerebellum, brainstem, cerebral cortex, and other motor and sensory thalamo-cortical pathways.^{38,39} Therefore, dystonia might be most accurately described as a neuronal network disorder, with different phenotypes possibly reflecting different pathophysiological states triggered by various insults or abnormalities.⁴⁰ This network model provides testable hypotheses that are directly relevant to new treatment strategies that extend beyond the basal ganglia.⁴¹ From this perspective,

we hypothesise that the pathophysiology of dystonia caused by an inherited disorder (ie, primary dystonia) differs from that of secondary dystonia in DCP, as supported by different responses to treatment and by differences in motor cortex plasticity and cerebellar function in the modulation of dystonic features.^{40,41} Therefore, further research is needed to unravel the pathophysiology of dystonia, specifically in DCP, because the focus of pathophysiological studies over the past two decades has been almost exclusively placed on inherited (primary) dystonia.

Choreoathetosis has been underexplored in clinical studies, except in selected conditions (ie, not DCP). Recent findings have shown a stronger association of selective thalamus and basal ganglia lesions for choreoathetosis than for dystonia,⁴ suggesting that further research on choreoathetosis in DCP is warranted.

Combined approaches can further advance understanding of dystonia and choreoathetosis in people with DCP. High-resolution MRI and voxel-based morphometry might improve understanding of structural injury, and clarify the implication of selected deep grey matter nuclei and other brain lesions, including the motor cortex and the cerebellum, and their correlation with appropriately described motor phenotypes. The current description of DCP might not be relevant to a pathophysiological explanation of the motor manifestations; therefore, it may be necessary to define several motor phenotypes. Diffusion tensor imaging provides information about connectivity and might help to predict the likelihood of responsiveness to DBS.⁴² Electrophysiological assessment, including transcranial magnetic stimulation,⁴³ central motor conduction time⁴⁴ and somatosensory evoked potentials,^{45,46} might also contribute towards diagnosis and management of DCP.



Figure 1. Abnormal posture and movement in dyskinetic cerebral palsy. (A) Primitive asymmetric tonic neck reflex typically present in patients with dyskinetic cerebral palsy. (B) Lack of coordination when in an upright position and grasping objects. (C) Large, uncontrolled involuntary choreo-athetosis movements during activities. (D) Dystonic postures and movement due to sustained or intermittent muscle contractures.



Figure 2. Basal ganglia and thalamic lesions after hypoxic-ischaemic encephalopathy

T2-weighted MRI images of the transverse plane (A) and coronal plane (B) show bilateral focal hyperintensity in posterior putamen (blue arrow), mediolateral thalamus (red arrow), and central region. Images are courtesy of Kate Himmelmann (Department of Pediatrics, Sahlgrenska Academy, University of Gothenburg, Gothenburg, Sweden).



Figure 3. Mild, moderate, and severe MRI patterns of basal ganglia and thalamus lesions

(A–C) Mild pattern of basal ganglia and thalamus lesions (T2-weighted, red and green arrows; T1-weighted, blue arrow); central region not involved (T2-weighted, white arrow). Clinical outcome: dyskinetic cerebral palsy. (D–F) Moderate pattern: Basal ganglia and thalamus lesions (T2-weighted, red and green arrows); central cortical and subcortical lesions (FLAIR, white arrow); and hippocampus not involved (T2-weighted, blue arrow). Clinical outcome: dyskinetic crebral palsy with spastic traits, or spastic cerebral palsy with dyskinetic trait, depending on the involvement of the motor cortex. (G–I) Severe pattern of basal ganglia and thalamus lesions (T2-weighted, red arrow); central cortical and subcortical lesion (T2-weighted, white arrow); and hippocampus lesions (T2-weighted, blue arrow). Clinical outcome: severe bilateral spastic cerebral palsy. Images are courtesy of Krägeloh-Mann and colleagues.²⁷

MANAGEMENT

Management of dystonia and choreoathetosis should be timed and targeted to maximise the potential for cerebral plasticity in people with DCP.⁴⁷ As suggested by the WHO International Classification of Functioning, Disability and Health, management should aim to improve daily activity and quality of life by improving movement and posture and relieving any associated disability, pain, and discomfort.²⁵

Non-motor comorbidities, such as epilepsy or depression, require specific treatment. To ensure optimal nutrition, gastrostomy tube feeding can be considered on the basis of a comprehensive assessment by a multidisciplinary team.^{48,49} Management should also focus on prevention of contractures and orthopaedic complications.⁴⁹

Little evidence supports the use of pharmacological treatment of dystonia and choreoathetosis in people with DCP.⁵⁰ Many patients with DCP take multiple oral drugs with low efficacy and unwanted side-effects.⁵¹ Nevertheless, in some patients, oral drugs can be useful. Before considering medical treatment of dystonia, it is important to determine whether the patient also shows spasticity, as this requires a specific approach. Particularly in patients without MRI abnormalities, the possibility of dopamine-responsive dystonia as a mimic DCP should be considered. However, levodopa shows limited efficacy in patients with DCP.⁵² Less research has been done concerning treatment for choreoathetosis. Dopamine-depleting agents can influence the hyperkinetic component of dyskinesia.⁵¹ Other studies have reported a positive effect of levetiracetam in children with choreoathetosis.^{53,54}

The table provides an overview of medications used to manage dystonia and choreoathetosis in people with DCP. Oral baclofen, a GABA-B agonist used for spasticity management, is the most commonly used oral drug for DCP despite its low efficacy.⁵¹ The efficacy of trihexyphenidyl, an anticholinergic drug, on dystonia is also low^{61,62} and its use in people with DCP is often restricted by the risk of adverse effects. Adverse effects can include worsening of choreoathetosis, possibly owing to so-called unmasking when severe dystonia is reduced (ie, dystonia prevented the full expression of choreoathetosis).⁴

Intrathecal baclofen treatment was introduced as an alternative to oral administration with fewer side-effects, and much lower dosage. Many studies of intrathecal baclofen treatment have reported significant decreases in dystonia^{63–65} and subjective improvements in mobility,^{66,67} speech and communication, swallowing,⁵⁹ head control,⁶⁸ sleep,⁶⁶ pain and mood,64 comfort,^{66,68–71} and ease of care.⁶⁶ Patient satisfaction and

attained goals are generally high.⁷¹ Although intrathecal baclofen decreases dystonia in most patients, the treatment usually leads to no changes in functional independence for daily activities⁶⁴ Therefore, treatment goals mainly focus on improving pain and preventing deformities.^{64,69,71,72} Such therapy could be particularly appropriate in patients with both dystonia and spasticity.⁷² Whereas the benefits to improve dystonia are well known, the effect of intrathecal baclofen on choreoathetosis has not been investigated.

In people with DCP, the intrathecal baclofen catheter tip is often placed high at the cervical spinal level, because major sites of action of baclofen are likely to be intracranial.⁶⁵ The response cannot be fully predicted and the dosage must be individualised.⁷³ If the effect of a continuous dose is not sufficient or has a gradually reduced effect, intermittent bolus dosing can be used.⁷³ Complications after intrathecal baclofen implantation are common. Infection risk is high (33–44%),^{71,74} particularly in the first month after surgery,^{73,75} and infection treatment always requires pump removal. CSF leaks occur in 0–21% of cases,^{64,71} with an increased risk during patient transfers and mobilisation. Extended bed rest usually resolves the leak and accompanying light-headedness, but surgery is sometimes needed.⁶⁵ Catheter-related problems, such as migration, disconnection, and blockage, often manifest with signs of baclofen withdrawal, for example an increase in spasticity or dystonia, or both, generalised itching and agitation, and hypertension.⁷³ These issues occur in 5–33% of patients with catheter-related problems.^{63,64,71} When patients are awaiting surgery, oral baclofen can be given to decrease the severity of withdrawal symptoms.

On the basis of dramatic positive outcomes in patients with inherited (primary) dystonia,^{76,77} DBS has been increasingly used in people with DCP over the past decade.^{23,37,78–81} Studies of DBS in people with DCP often describe the reduction of dystonia rather than its effects on functionality, quality of life, or choreoathetosis.^{25,82–84} The responsiveness of dystonia to DBS treatment is variable⁸⁵ and usually less beneficial than in patients with inherited (primary) dystonia.⁸⁶ Long-term data are rare, but a case series follow-up study in Italy that included 15 patients reported that improvements in BFMS values were sustained for up to 2 years after implant.⁷⁹

The relatively modest and often delayed effects of DBS in people with DCP might be due to abnormal plasticity and an altered neuronal cortex-basal ganglia network.^{34,87} Whereas in inherited (primary) movement disorders, dystonia becomes symptomatic after a period of normal development, in people with DCP, dystonia presents in early life

before functional motor patterns have developed and, when they do, those movements and postures are often abnormal.^{37,88}

Currently, the globus pallidus internus is the target of choice for electrode implantation in people with DCP.56 DBS programming is personalised based on subjective evaluations of clinical benefit; an objective approach to parameter selection would require better understanding of the underlying pathophysiology. DBS is generally safe, and a reversible method of treatment. Complications such as intracerebral haemorrhage or ischaemic infarction are rare, but infection and hardware problems of the implanted pulse generator can occur.⁸⁹ More severe adverse events of DBS have been reported in children than in adults;²³ however, large cohort studies have reported low infection rates of less than 5% in children younger than 7 years.⁹⁰

Responsiveness of choreoathetosis to DBS has not been documented, and little is known about the effect of DBS on daily activity, societal participation, or quality of life. Various collaborative efforts are underway to systematically collect data on individuals with DBS, including age of implantation, history of eventual musculoskeletal changes, activity levels, and inclusion in an international registry.^{23,91}

Botulinum toxin is often used to decrease dystonia in people with DCP.⁹² Botulinum toxin causes selective (focal) muscular denervation that is temporary and reversible. Its therapeutic value has been much more extensively assessed in inherited (primary) dystonia than in people with DCP. Botulinum toxin is also used in individuals with hip dislocation or disabling limb dystonia. A few small studies have indicated that botulinum toxin has an effect on dystonia and pain in people with DCP.^{93,94} Many issues remain unresolved because of variation in injection methods and a shortage of knowledge of the determinants of local spread and distant effects.⁵⁵ To date, no specific indication is available for choreoathetosis.

As musculoskeletal deformities seen in patients with DCP and those with other childhood dystonias⁹ are often associated with pain, patients might undergo orthopaedic surgery.⁴⁹ However, before orthopaedic surgery is done, it is recommended that attempts are made to control dystonia and choreoathetosis with oral drugs, intrathecal baclofen, or DBS. Furthermore, based on our clinical experience, the results of soft tissue orthopaedic surgery in patients with DCP are not as predictable as in patients with spastic cerebral palsy, and adverse outcomes can easily occur. By contrast, orthopaedic procedures can be carried out more safely in patients with DCP than in those with spastic CP.

The medical management options discussed are not appropriate as standalone treatment. Combinations with rehabilitation approaches carried out by physiotherapists, occupational therapists, and speech and language therapists are key components of the management of dystonia and choreoathetosis. Evidence for rehabilitative strategies is mostly based on studies in spastic cerebral palsy and remains scarce in DCP.^{95,96} Current practice is therefore mainly based on clinical expertise. This expertise is usually offered by dedicated multidisciplinary rehabilitation teams designing individualised management programmes that begin as early as possible in the life of the patient. Use of a family-centred approach and the concepts developed in WHO's International Classification of Functioning, Disability, and Health can ensure the development of skills and good quality of life.^{97–99}

Most patients have substantial motor disability, therefore optimal seating and positioning should be ensured to increase stability and the coordination of movements for mobility (including electronic wheelchair use) and hand function. Combined with medical management, rehabilitative research should focus on goal-directed strategies that consider general guidelines to handle postural asymmetry and individual needs.⁴⁹ Verbal communication is almost always affected in people with DCP,^{100,101} therefore augmentative and alternative communication intervention is often needed. Without appropriate communication intervention, the intellectual level of the individual with DCP could be underestimated.⁷ In addition, given the potential for brain plasticity, cognitive training and motor learning research is warranted in DCP management.

Table Medications used to manage dystonia and choreo-athetosis in patients with dysk	inetic
cerebral palsy	

	Mechanism of action	Dystonia	Chorea	Side effects
Dopamine agonists (levodopa)55	Enhances the activity of dopamine	Used		Nausea, orthostatic hypotension, and constipation
Anticholinergic (trihexiphenidyl, benzatropine) 51,55	Blocks acetylcholine muscarinic receptor	Used		Drowsiness, confusion, memory difficulty, blurred vision, hallucinations, urinary retention, and worsening chorea
Benzodiazepine receptor agonists (diazepam, clonazepam)55,56	Enhances GABA-A inhibition	Used	Used	Sedation, confusion, depression, ataxia, and dependence
GABA-B receptor agonist (baclofen)51,55–57	Enhances the activity of GABA-B receptor	Used		Worsening chorea, incontinence, sedation, dizziness, dry mouth, and increased blood glucose
Dopamine and serotonin antagonist (clozapine)56	Binds to serotonin and dopamine receptors and prevents release	Used		Decreased white blood cell count, sedation, hypotension, myocarditis, cardiomyopathy, drooling, arrhythmia, seizures, and diabetes mellitus
Pre-synaptic α2 receptor agonist (clonidine)58	Enhances the activity of pre-synaptic α2 receptor	Used		Orthostatic hypotension, bradycardia, sedation, fatigue, and headache
Dopamine antagonists (pimozide, haloperidol)55	Antagonist of the D2, D3, and D4 dopamine receptors, and the 5-HT ₇ receptor		Used	Hypotension, sedation, QT interval prolongation, and ventricular arrhythmias (including torsades de pointes); overdose causes severe extrapyramidal symptoms
Monoamine blockers (tetrabenazine)51,55,56	Inhibits vesicular monoamine transporter 2, resulting in decreased uptake of monoamines into synaptic vesicles and depletion of monoamine storage	Used	Used	Drowsiness, parkinsonism, depression, insomnia, anxiety, and akathisia

	Mechanism of action	Dystonia	Chorea	Side effects
Monoamine depleters (reserpine)51	Blocks the vesicular monoamine transporter		Used	Nasal congestion, nausea, vomiting, weight gain, gastric intolerance, gastric ulceration, stomach cramps and diarrhoea, hypotension, bradycardia, and worsening of asthma
Voltage-gated sodium and calcium channel blocker (carbamazepine)55	Blocks voltage- sensitive sodium channels	Used		Decreased white blood cell count and platelets; increased risk of suicide
Calcium channel blocker (levetiracetam)53,54	Binds to a synaptic vesicle glycoprotein and inhibits presynaptic calcium channels, reducing neurotransmitter release and acting as a neuromodulator		Used	Somnolence, decreased energy, headache, dizziness, and (mild) ataxia
Muscle tone reducer (dantrolene)59	Reduces skeletal muscle tone at the muscle fibre level	Used		Speech and visual disturbances; depression and confusion; hallucinations; headache; insomnia and exacerbation or precipitation of seizures, and increased nervousness
Voltage-gated calcium channel blocker (gabapentin)60	Antagonises binding of thrombospondin to voltage-gated calcium channel a2d-1 receptors and inhibits synthesis of glutaminergic excitatory synapses	Used		Dizziness, drowsiness, sedation, fever, fatigue, viral infection, ataxia

Table 1. Continued

CONCLUSIONS AND FUTURE DIRECTIONS

Many epidemiological studies^{2,102} are currently underway; these studies are mostly population-based or registry-based and focus on the identification of risk factors that will eventually lead to the implementation of preventive strategies. Hypothermia for term and near-term neonates with hypoxia–ischaemia,¹⁰³ and magnesium sulphate and other agents as neuroprotective approaches for fetuses are promising approaches under investigation.¹⁰⁴ Population-based follow-up programmes and studies will add to the evidence base regarding long-term secondary complications and results of management.

Current management options are mainly based on clinical presentation. To evaluate the efficacy of therapy and to adjust and delineate management, well designed studies and systematic use of meaningful outcome measures are needed. Future research should include additional objective and quantitative tools to assess dystonia and choreoathetosis. Moreover, owing to the relatively small population of patients with DCP, multicentre studies are crucial for the development of evidence-based guidelines for the management of dystonia and choreoathetosis. Specific evaluation of rehabilitation strategies, including physical, occupational, and speech and language therapy, is urgently required.¹⁰¹ Robotics and other emerging technologies have the potential to increase independent mobility. Better understanding of motor learning processes is needed to implement these new technologies.

To date, medical interventions in DCP (eg, intrathecal baclofen, DBS, and botulinum toxin) have mainly targeted dystonia but not choreoathetosis, which requires more research attention. Further research into the effect of interventions on daily activities, societal participation, quality of life, pain, nutrition, and sleep is also required. The clinical interrelationship between dystonia and choreoathetosis after different interventions is also worth further research.

The neurophysiological basis of dystonia and choreoathetosis should be explored. To date, there are no animal models for DCP.¹⁰⁵ Increased insight into the neurophysiological circuits in patients with DCP can be obtained during intraoperative pallidal recordings of local field potentials and microelectrode recordings. Neuronal firing frequencies have been shown to correlate with dystonia phenotype and the timing of the insult, which could be partially predictive of DBS outcome.⁴³ Other non-invasive neurophysiological models in children and adults with DCP and their neuroimaging findings are needed to explore cerebral plasticity.^{106,107}

In conclusion, although DCP is the second most common type of cerebral palsy, the condition is under-reported⁷ and management remains empirical and largely unsatisfactory. Therefore, a better scientific foundation to therapeutic approaches is urgently required. Early detection, improved neuroimaging, and improved neurophysiological characterisation of dystonia and choreoathetosis in DCP are needed. Increased understanding of the affected brain structures could contribute to better targeted therapeutic management.

CONTRIBUTORS

All authors contributed significantly to the work, have read the manuscript, attest to the validity of the data and its interpretation, and agree to its submission. EM provided the videos and figure 1 images and KH provided the images in figures 2 and 3.

DECLARATION OF INTERESTS

J-PL has received a Guy's and St Thomas Charity New Services and Innovation Grant (G060708), grants from The Dystonia Society UK (01/2011, 07/2013) and Action Medical Research (GN2097), is a consultant for Medtronic Ltd, and has received unrestricted educational grants from Medtronic Ltd for delivering courses and attending conferences. RJV has received unrestricted educational grants from International GmbH for his advisory board role. All other authors declare no competing interests.

ACKNOWLEDGEMENTS

We would like to thank Marc Vermandere (Dominiek Savio Instituut, Gits, Belgium) for his support in compiling the videos.

Search strategy and selection criteria

We identified references for this Review by searching PubMed for reports published between Jan 1, 2006 and March 31, 2017, and references from relevant articles. Search terms used were "dyskinetic cerebral palsy" and "dystonic cerebral palsy", "hypoxic-ischemic encephalopathy" and "dystonia", and "choreoathetosis" and "pathophysiology" in combination with "management" or "therapy", or "treatment". We only considered peer-reviewed reports in English

REFERENCES

- Rosenbaum P, Paneth N, Leviton A, et al. A report: the definition and classification of cerebral palsy April 2006. *Dev Med Child Neurol* 2007; 49: 8–14.
- Sellier E, Platt MJ, Andersen GL, Krägeloh-Mann I, De La Cruz J, Cans C. Decreasing prevalence in cerebral palsy: a multi-site European population-based study, 1980 to 2003. *Dev Med Child Neurol* 2016; 58: 85–92.
- Yeargin-Allsopp M, Braun KVN, Doernberg NS, Benedict RE, Kirby RS, Durkin MS. Prevalence of cerebral palsy in 8-yearold children in three areas of the United States in 2002: a multisite collaboration. *Pediatrics* 2008; **121:** 547–54.
- Monbaliu E, De Cock P, Ortibus E, Heyrman L, Klingels K, Feys H. Clinical patterns of dystonia and choreoathetosis in participants with dyskinetic cerebral palsy. *Dev Med Child Neurol* 2016; 58: 138–44.
- Himmelmann K, McManus V, Hagberg G, Uvebrant P, Krägeloh-Mann I, Cans C. Dyskinetic cerebral palsy in Europe: trends in prevalence and severity. *Arch Dis Child* 2009; **94:** 921–26.
- Olusanya B, Osibanjo F, Mabogunje C, Slusher T, Olowe S. The burden and management of neonatal jaundice in Nigeria: a scoping review of the literature. *Niger J Clin Pract* 2016; **19**: 1–17.
- Himmelmann K, Hagberg G, Wiklund L, Eek MN, Uvebrant P. Dyskinetic cerebral palsy: a population-based study of children born between 1991 and 1998. Dev Med Child Neurol 2007; 49: 246–51.
- Monbaliu E, De Cock P, Mailleux L, Dan B, Feys H. The relationship of dystonia and choreoathetosis with activity, participation and quality of life in children and youth with dyskinetic cerebral palsy. *Eur J Paediatr Neurol* 2017; 21: 327–35.

- Lumsden DE, Gimeno H, Elze M, Tustin K, Kaminska M, Lin JP. Progression to musculoskeletal deformity in childhood dystonia. *Eur J Paediatr Neurol* 2016; 20: 339–45.
- Kim KN, Ahn PG, Ryu MJ, Shin DA, Yi S, Ha Y. Long-term surgical outcomes of cervical myelopathy with athetoid cerebral palsy. *Eur Spine J* 2014; 23: 1464–71.
- Himmelmann K, Sundh V. Survival with cerebral palsy over five decades in western Sweden. *Dev Med Child Neurol* 2015; 57: 762–67.
- Sanger TD, Chen D, Fehlings DL, et al. Definition and classification of hyperkinetic movements in childhood. *Mov Disord* 2010; 25: 1538–49.
- Krägeloh-Mann I, Petruch U, Weber P. SCPE reference and training manual (R&TM). Grenoble: Surveillance of Cerebral Palsy in Europe, 2005.
- Cans C, Dolk H, Platt M, Colver A, Prasauskiene A, Krägeloh-Mann. Recommendations from the SCPE collaborative group for defining and classifying cerebral palsy. *Dev Med Child Neurol* 2007; 49: 35–38.
- Jethwa A, Mink J, Macarthur C, Knights S, Fehlings T, Fehlings D. Development of the Hypertonia Assessment Tool (HAT): a discriminative tool for hypertonia in children. *Dev Med Child Neurol* 2010; **52**: e83–87.
- Knights S, Datoo N, Kawamura A, et al. Further evaluation of the scoring, reliability, and validity of the Hypertonia Assessment Tool (HAT). J Child Neurol 2014; 29: 500–04.
- Lin JP, Nardocci N. Recognizing the common origins of dystonia and the development of human movement: a manifesto of unmet needs in isolated childhood dystonias. *Front Neurol* 2016; 7: 226.

- Burke RE, Fahn S, Marsden CD, Bressman SB, Moskowitz C, Friedman J. Validity and reliability of a rating scale for the primary torsion dystonias. *Neurology* 1985; 35: 73–76.
- Barry MJ, VanSwearingen JM, Albright AL. Reliability and responsiveness of the Barry-Albright dystonia scale. *Dev Med Child Neurol* 1999; **41:** 404–11.
- 20. Monbaliu E, Ortibus E, De Cat J, et al. The Dyskinesia Impairment Scale: a new instrument to measure dystonia and choreoathetosis in dyskinetic cerebral palsy. *Dev Med Child Neurol* 2012; **54**: 278–83.
- 21. Monbaliu E, Ortibus E, Roelens F, et al. Rating scales for dystonia in cerebral palsy: reliability and validity. *Dev Med Child Neurol* 2010; **52:** 570–75.
- Monbaliu E, Ortibus E, Prinzie P, et al. Can the Dyskinesia Impairment Scale be used by inexperienced raters? A reliability study. *Eur J Paediatr Neurol* 2013; 17: 238–47.
- Koy A, Timmermann L. Deep brain stimulation in cerebral palsy: Challenges and opportunities. *Eur J Paediatr Neurol* 2017; **21:** 118–21.
- 24. Bonouvrie LA, Becher JG, Vles JS, et al. Intrathecal baclofen treatment in dystonic cerebral palsy: a randomized clinical trial: the IDYS trial. *BMC Pediatr* 2013; **13**: 175.
- 25. Gimeno H, Lin JP. The International Classification of Functioning (ICF) to evaluate deep brain stimulation neuromodulation in childhood dystoniahyperkinesia informs future clinical & research priorities in a multidisciplinary model of care. *Eur J Paediatr Neurol* 2017; 21: 147–67.
- 26. Allen NM, Lin JP, Lynch T, King MD. Status dystonicus: a practice guide. *Dev Med Child Neurol* 2014; **56:** 105–12.

- Krägeloh-Mann I, Helber A, Mader I, et al. Bilateral lesions of thalamus and basal ganglia: origin and outcome. *Dev Med Child Neurol* 2002; 44: 477–84.
- Aravamuthan BR, Waugh JL. Localization of basal ganglia and thalamic damage in dyskinetic cerebral palsy. *Pediatr Neurol* 2016; 54: 11–21.
- 29. Himmelmann K. Dyskinetic cerebral palsy: challenges and new approaches. *Dev Med Child Neurol* 2011; **53:** 10–11
- Benini R, Dagenais L, Shevell MI, for the Consortium RdIPCaQ. Normal imaging in patients with cerebral palsy: what does it tell us? *J Pediatr* 2013; 162: 369–74.
- Yilmaz Y, Alper G, Kiliçoglu G, Çelik L, Karadeniz L, Yilmaz-Degirmenci S. Magnetic resonance imaging findings in patients with severe neonatal indirect hyperbilirubinemia. J Child Neurol 2001; 16: 452–55.
- Platt MJ, Krageloh-Mann I, Cans C. Surveillance of cerebral palsy in Europe: reference and training manual. *Med Educ* 2009; 43: 495–96.
- Himmelmann K, Horber V, De La Cruz J, et al. MRI classification system (MRICS) for children with cerebral palsy: development, reliability, and recommendations. *Dev Med Child Neurol* 2017; 59: 57–64
- Quartarone A, Hallett M. Emerging concepts in the physiological basis of dystonia. *Mov Disord* 2013; 28: 958–67.
- Hallett M. Neurophysiology of dystonia: the role of inhibition. *Neurobiol Dis* 2011; 42: 177–84.
- Patel N, Jankovic J, Hallett M. Sensory aspects of movement disorders. *Lancet Neurol* 2014; 13: 100–12.

- Lumsden D, Kaminska M, Gimeno H, et al. Proportion of life lived with dystonia inversely correlates with response to pallidal deep brain stimulation in both primary and secondary childhood dystonia. *Dev Med Child Neurol* 2013; 55: 567–74.
- Filip P, Lungu OV, Bareš M. Dystonia and the cerebellum: a new field of interest in movement disorders? *Clin Neurophysiol* 2013; **124:** 1269–76.
- Stoessl AJ, Lehericy S, Strafella AP. Imaging insights into basal ganglia function, Parkinson's disease, and dystonia. *Lancet* 2014; 384: 532–44.
- Kojovic M, Pareés I, Kassavetis P, et al. Secondary and primary dystonia: pathophysiological differences. *Brain* 2013; **136**: 2038–49.
- 41. Neychev VK, Gross RE, Lehéricy S, Hess EJ, Jinnah H. The functional neuroanatomy of dystonia. *Neurobiol Dis* 2011; **42:** 185– 201.
- 42. Lumsden DE, McClelland V, Ashmore J, Charles-Edwards G, Mills K, Lin JP. Central Motor Conduction Time and Diffusion Tensor Imaging metrics in children with complex motor disorders. *Clin Neurophysiol* 2015; **126:** 140–46.
- Dachy B, Dan B. Electrophysiological assessment of the effect of intrathecal baclofen in dystonic children. *Clin Neurophysiol* 2004; **115**: 774–78.
- 44. McClelland V, Mills K, Siddiqui A, Selway R, Lin JP. Central motor conduction studies and diagnostic magnetic resonance imaging in children with severe primary and secondary dystonia. *Dev Med Child Neurol* 2011; **53**: 757–63.
- McClelland VM, Cvetkovic Z, Mills KR. Modulation of corticomuscular coherence by peripheral stimuli. *Exp Brain Res* 2012; 219: 275–92.

- McClelland V, Valentin A, Rey HG, et al. Differences in globus pallidus neuronal firing rates and patterns relate to different disease biology in children with dystonia. J Neurol Neurosurg Psychiatry 2016; 87: 958–67.
- Ismail FY, Fatemi SA, Johnston MV. Cerebral plasticity: windows of opportunity in the developing brain. *Eur J Paediatr Neurol* 2017; 21: 23–48.
- 48. Dahlseng MO, Andersen GL, Da Graca Andrada M, et al. Gastrostomy tube feeding of children with cerebral palsy: variation across six European countries. *Dev Med Child Neurol* 2012; **54:** 938–44.
- Dan B, Mayston M, Paneth N, Rosenbloom
 Cerebral palsy: science and clinical practice. London: Wiley; 2015.
- Koy A, Lin J-P, Sanger TD, Marks WA, Mink JW, Timmermann L. Advances in management of movement disorders in children. *Lancet Neurol* 2016; 15: 719–35.
- 51. Lumsden D, Kaminska M, Tomlin S, Lin JP. Medication use in childhood dystonia. *Eur J Paediatr Neurol* 2016; **20:** 625–29.
- Maas RP, Wassenberg T, Lin JP, van de Warrenburg BP, Willemsen MA. L-Dopa in dystonia: a modern perspective. *Neurology* 2017 88: 1865–71.
- Vles GF, Hendriksen JG, Visschers A, Speth L, Nicolai J, Vles JS. Levetiracetam therapy for treatment of choreoathetosis in dyskinetic cerebral palsy. *Dev Med Child Neurol* 2009; **51:** 487–90.
- 54. Recio MV, Hauser RA, Louis ED, Radhashakar H, Sullivan KL, Zesiewicz TA. Chorea in a patient with cerebral palsy: treatment with levetiracetam. *Mov Disord* 2005; **20:** 762–64.
- 55. Roubertie A, Mariani LL, Fernandez-Alvarez E, Doummar D, Roze E. Treatment for dystonia in childhood. *Eur J Neurol* 2012; **19:** 1292–99.
- 56. Jankovic J. Treatment of dystonia. *Lancet Neurol* 2006; **5:** 864–72.

- Kolker S, Christensen E, Leonard JV, et al. Diagnosis and management of glutaric aciduria type I—revised recommendations. J Inherit Metab Dis 2011; 34: 677–94.
- Sayer C, Lumsden D, Kaminska M, Lin JP. Clonidine use in the outpatient management of severe secondary dystonia. *Eur J Paediatr Neurol* 2017; 21: 621–26.
- van Karnebeek C, Horvath G, Murphy T, et al. Deep brain stimulation and dantrolene for secondary dystonia in x-linked adrenoleukodystrophy. *JIMD Rep* 2015; 15: 113–16.
- Liow NY, Gimeno H, Lumsden D, et al. Gabapentin can significantly improve dystonia severity and quality of life in children. *Eur J Paediatr Neurol* 2016; 20: 100–07.
- Rice J, Waugh Mary-Clare MC. Pilot study on trihexyphenidyl in the treatment of dystonia in children with cerebral palsy. *J Child Neurol* 2009; **24:** 176–82.
- Sanger TD, Bastian A, Brunstrom J, et al. Prospective open-label clinical trial of trihexyphenidyl in children with secondary dystonia due to cerebral palsy. J Child Neurol 2007; 22: 530–37.
- Motta F, Antonello CE, Stignani C. Upper limbs function after intrathecal baclofen therapy in children with secondary dystonia. J Pediatr Orthop 2009; 29: 817–21.
- Motta F, Stignani C, Antonello CE. Effect of intrathecal baclofen on dystonia in children with cerebral palsy and the use of functional scales. *J Pediatr Orthop* 2008; 28: 213–17.
- Albright AL, Barry MJ, Shafton DH, Ferson SS. Intrathecal baclofen for generalized dystonia. *Dev Med Child Neurol* 2001; 43: 652–57.

- 66. Bonouvrie L, Becher J, Soudant D, et al. The effect of intrathecal baclofen treatment on activities of daily life in children and young adults with cerebral palsy and progressive neurological disorders. Eur J Paediatr Neurol 2016; 20: 538–44
- Motta F, Antonello CE, Stignani C. Intrathecal baclofen and motor function in cerebral palsy. *Dev Med Child Neurol* 2011; 53: 443–48.
- Woon K, Tsegaye M, Vloeberghs M. The role of intrathecal baclofen in the management of primary and secondary dystonia in children. *Br J Neurosurg* 2007; 21: 355–58.
- Bonouvrie LA, van Schie PE, Becher JG, van Ouwerkerk WJ, Reeuwijk A, Vermeulen JR. Effects of intrathecal baclofen on daily care in children with secondary generalized dystonia: a pilot study. *Eur J Paediatr Neurol* 2011; **15**: 539–43.
- Motta F, Buonaguro V, Stignani C. The use of intrathecal baclofen pump implants in children and adolescents: safety and complications in 200 consecutive cases. *J Neurosurg* 2007; **107:** 32–35.
- Ward A, Hayden S, Dexter M, Scheinberg A. Continuous intrathecal baclofen for children with spasticity and/or dystonia: goal attainment and complications associated with treatment. J Paediatr Child H 2009; 45: 720–26.
- Dan B, Motta F, Vles JS, et al. Consensus on the appropriate use of intrathecal baclofen (ITB) therapy in paediatric spasticity. *Eur J Paediatr Neurol* 2010; 14: 19–28.
- Albright AL. Intrathecal baclofen for childhood hypertonia. *Child Nerv Syst* 2007; 23: 971–79.
- Albright AL, Barry MJ, Fasick P, Barron W, Shultz B. Continuous intrathecal baclofen infusion for symptomatic generalized dystonia. *Neurosurgery* 1996; 38: 934–39.

- Ghosh D, Mainali G, Khera J, Luciano M. Complications of intrathecal baclofen pumps in children: experience from a tertiary care center. *Pediatr Neurosurg* 2014; 49: 138–44.
- Coubes P, Roubertie A, Vayssiere N, Hemm S, Echenne B. Treatment of DYT1generalised dystonia by stimulation of the internal globus pallidus. *Lancet* 2000; 355: 2220–21.
- Krauss JK, Pohle T, Weber S, Ozdoba C, Burgunder J-M. Bilateral stimulation of globus pallidus internus for treatment of cervical dystonia. *Lancet* 1999; **354**: 837–38.
- Vidailhet M, Yelnik J, Lagrange C, et al. Bilateral pallidal deep brain stimulation for the treatment of patients with dystonia-choreoathetosis cerebral palsy: a prospective pilot study. *Lancet Neurol* 2009; 8: 709–17.
- Romito L, Zorzi G, Marras C, Franzini A, Nardocci N, Albanese A. Pallidal stimulation for acquired dystonia due to cerebral palsy: beyond 5 years. *Eur J Neurol* 2015; 22: 426–e32.
- Marks WA, Honeycutt J, Acosta F, et al. Dystonia due to cerebral palsy responds to deep brain stimulation of the globus pallidus internus. *Mov Disord* 2011; 26: 1748–51.
- Koy A, Hellmich M, Pauls KA, et al. Effects of deep brain stimulation in dyskinetic cerebral palsy: a meta-analysis. *Mov Disord* 2013; 28: 647–54.
- Gimeno H, Tustin K, Selway R, Lin J-P. Beyond the Burke–Fahn–Marsden Dystonia Rating Scale: deep brain stimulation in childhood secondary dystonia. *Eur J Paediatr Neurol* 2012; 16: 501–08.

- Bimeno H, Gordon A, Tustin K, Lin JP. Functional priorities in daily life for children and young people with dystonic movement disorders and their families. *Eur J Paediatr Neurol* 2013; 17: 161–68.
- Lumsden DE, Gimeno H, Tustin K, Kaminska M, Lin JP. Interventional studies in childhood dystonia do not address the concerns of children and their carers. *Eur J Paediatr Neurol* 2015; 19: 327–36.
- Lin J. Deep brain stimulation in children: clinical considerations. In: Kirton A, Gilbert D eds. Pediatric brain stimulation. Mapping and modulating the developing brain, 1st edn. London: Elsevier, 2016: 401–19.
- Marks W, Bailey L, Reed M, et al. Pallidal stimulation in children: Comparison between cerebral palsy and DYT1 dystonia. J Child Neurol 2013; 28: 840–48.
- Barow E, Neumann W-J, Brücke C, et al. Deep brain stimulation suppresses pallidal low frequency activity in patients with phasic dystonic movements. *Brain* 2014; 137: 3012–24.
- Lin J-P, Lumsden DE, Gimeno H, Kaminska M. The impact and prognosis for dystonia in childhood including dystonic cerebral palsy: a clinical and demographic tertiary cohort study. *J Neurol Neurosurg Ps* 2014; 85: 1239–44.
- Fenoy AJ, Simpson RK Jr. Risks of common complications in deep brain stimulation surgery: management and avoidance: clinical article. *J Neurosurg* 2014; **120**: 132–39.
- Kaminska M, Perides S, Lumsden D, et al. Complications of Deep Brain Stimulation (DBS) for dystonia in children—the challenges and 10 year experience in a large paediatric cohort. Eur J Paediatr Neurol 2017; 21: 168–75.

- Marks W, Bailey L, Sanger TD. PEDiDBS: the pediatric international deep brain stimulation registry project. *Eur J Paediatr Neurol* 2017; 21: 218–22.
- 92. Elkamil AI, Andersen GL, Skranes J, Lamvik T, Vik T. Botulinum neurotoxin treatment in children with cerebral palsy: a population-based study in Norway. *Eur J Paediatr Neurol* 2012; **16**: 522–27.
- Jankovic J. Medical treatment of dystonia. Mov Disord 2013; 28: 1001–12.
- 94. Lundy CT, Doherty GM, Fairhurst CB. Botulinum toxin type A injections can be an effective treatment for pain in children with hip spasms and cerebral palsy. *Dev Med Child Neurol* 2009; **51:** 705–10.
- 95. Hägglund GA, Alriksson-Schmidt A, Lauge-Pedersen H, Rodby-Bousquet E, Wagner PL, Westbom L. Prevention of dislocation of the hip in children with cerebral palsy: 20-year results of a population-based prevention programme. *Bone Joint J* 2014;
 96: 1546–52.
- Novak I, Mcintyre S, Morgan C, et al. A systematic review of interventions for children with cerebral palsy: state of the evidence. *Dev Med Child Neurol* 2013; 55: 885–910.
- Capelovitch S. The Bobath concept did globalization reduce it to a Chinese whisper? *Dev Med Child Neurol* 2017; 59: 557.
- WHO. International Classification of Functioning, Disability and Health. WHO, Geneva, Switzerland, 2001. Jan 27, 2017. http://www.who.int/classifications/icf/ en/ (accessed May 24, 2017).
- 99. National Institute for Health and Excellence. Cerebral palsy in under 25s: assessment and management. 2017. https://www.nice. org.uk/guidance/ng62 (accessed May 24, 2017).

- 100. Nordberg A, Miniscalco C, Lohmander A, Himmelmann K. Speech problems affect more than one in two children with cerebral palsy: Swedish population-based study. *Acta Paediatr* 2013; **102:** 161–66.
- 101. Monbaliu E, De La Peña MG, Ortibus E, Molenaers G, Feys H. Functional outcomes in children and young people with dyskinetic cerebral palsy. *Dev Med Child Neurol* 2017; **59:** 634–40.
- 102. Reid SM, Meehan E, McIntyre S, Goldsmith S, Badawi N, Reddihough DS. Temporal trends in cerebral palsy by impairment severity and birth gestation. *Dev Med Child Neurol* 2016; **58**: 25–35.
- 103. Jacobs S, Berg M, Hunt R, Tarnow-Mordi W, Inder T, Davis P. Cooling for newborns with hypoxic ischaemic encephalopathy. *Cochrane Database Syst Rev* 2013; CD003311.
- 104. Costantine MM, Weiner SJ. Effects of antenatal exposure to magnesium sulfate on neuroprotection and mortality in preterm infants: a meta-analysis. *Obstet Gynecol* 2009; **114:** 354.
- 105. Graham HK, Rosenbaum P, Paneth N, et al. Cerebral palsy. *Nature Rev Dis Primers* 2016; **2:** 16005.
- 106. Damji O, Keess J, Kirton A. Evaluating developmental motor plasticity with paired afferent stimulation. *Dev Med Child Neurol* 2015; **57:** 548–55.
- 107. Young SJ, Sanger TD. Cathodal transcranial direct current stimulation in children with dystonia: a sham-controlled study. J Child Neurol 2014; 29: 232–9.

CHAPTER 3

Effects of intrathecal baclofen on daily care in children with secondary generalized dystonia: a pilot study

LA Bonouvrié, PE van Schie, JG Becher, WJR van Ouwerkerk, A Reeuwijk, RJ Vermeulen

Eur J Paediatr Neurol 2011; 15(6): 539-543

ABSTRACT

Aim Treatment options for dystonic cerebral palsy (CP) are limited. Our aims were to determine whether intrathecal baclofen (ITB) improves daily care, decreases dystonia and decreases pain in patients with dystonic CP.

Methods Patients received randomized blinded treatment with ITB or placebo. Scores on problems of daily care were recorded and dystonia, pain and comfort were assessed.

Results Four patients (three males, average age 12 years 6 months) were included (all Gross Motor Function Classification System level V). During the trial period problem scores and dystonia scores decreased in all four patients.

Conclusion In this pilot study we report positive functional effects of ITB trial treatment in four patients with dystonic CP. A randomized trial with a larger cohort is needed to verify these results.

Keywords Secondary dystonia, Dyskinesia, Cerebral palsy, Intrathecal baclofen, Care giving

INTRODUCTION

Six to fifteen percent of patients with cerebral palsy (CP) have dyskinetic motor disorders^{1,2} of which 70% have dystonic features.³

Dystonia is dominated by abnormal postures and easily elicitable fluctuating hypertonia. Characteristics are involuntary movements, distorted voluntary movements and abnormal postures due to sustained muscle contractions.^{4,5} Dystonia in CP is referred to as secondary dystonia and is often associated with spasticity. It causes progressive disability, discomfort and deformity,⁶ which amongst others leads to impeding of daily care.

Oral medication (e.g. baclofen or trihexyphenidyl) shows limited effect.⁷⁻⁹ Intrathecal baclofen (ITB) treatment was first used for treatment of dystonia in 1991.¹⁰ The working mechanism of ITB in dystonia remains unclear but it is thought that ITB inhibits the excitation of the pre-motor and supplementary motor cortex.^{6,11-14}

Our aims were to perform a pilot study to determine if intrathecal baclofen (ITB) improves daily care and decreases dystonia and pain in patients with dystonic CP.

METHODS AND MATERIALS

The design of the study was a single center double blind randomized case controlled trial. All patients started with a trial period, in which they received ITB treatment. Hereafter they were randomized in two groups, an intervention and a control group.

Participants

Between 2001 and 2005 patients were recruited from the outpatient clinics of the departments of child neurology and pediatric rehabilitation medicine of the VU University Medical Center in Amsterdam, the Netherlands. Patients selected for ITB trial treatment, aged between 4 and 18 years old, with severe generalized dystonia, Gross Motor Function Classification System (GMFCS) level V, resulting in problems concerning daily care, were included if they and their parents were motivated and able to complete the whole study protocol.

Informed consent was signed by all parents. This study was approved by the Medical Ethical Committee of the VU University Medical Center in Amsterdam, The Netherlands.

Intervention

The trial treatment period consisted of a baseline observational period and a dose finding period. After the baseline period, an intrathecal catheter was placed under general anesthesia with the catheter tip at individually specified heights (Table 1). The intrathecal catheter was connected to either an external catheter or an external microinfusion pump as decided by the treating physician.

Administration of ITB was either continuously via the external micro-infusion pump or in a daily bolus dosage in case of the external catheter. The dosage was increased with approximately 25 mg per day until an optimal dosage was found. The maximal dosage was 200 mg per day. If dosage related side-effects occurred, dosage was decreased until side effects disappeared. If unacceptable complications occurred treatment was aborted.

The optimal dosage was maintained for at least three days. After this period, the patients were randomized in two groups for blinded treatment and received either ITB or intrathecal placebo treatment for four days. Hereafter, the intrathecal catheter was removed. After the trial treatment, the patients, caregivers and doctors decided, for or against implantation of a definite programmable pump (Medtronic[®]) for continuous ITB treatment.

Outcome measures

Three main problems in daily care were identified by caregivers in a preparatory interview. The perceived severity of these problems was scored on a Visual Analogue Scale (VAS), with "0" representing "no problems" and "10" representing "impossible to do". Parents filled out these VAS scores daily during the hospital stay. A positive effect was defined as 'a decrease of two or more points in at least one of three problems'. 'An increase of more than two points' was defined to be a negative effect.

For the clinical assessment of generalized dystonia the Barry-Albright Dystonia (BAD) scale, which grades dystonia on a 0 to 4 scale in eight body regions (eyes, face, neck, trunk, upper- and lower limbs), was used.^{11,15} The score ranges between 0 and 32 with a score of 0 representing no dystonia and a score of 32 representing maximal dystonic activity.

Assessments of dystonia were carried out daily at the same time of the day by one of two physicians (RJV, JGB). A reduction of 25% or more, in comparison with the baseline score, is considered clinically significant.^{11,15}

Pain was assessed with a VAS, with "0" representing "no pain" and "10" representing "very severe pain". Parents completed the VAS for pain daily, at the same time of the day, without knowledge of the baclofen dose.

Comfort was assessed using a VAS (score 0 to 10), with a higher score representing more comfort. Parents completed this score daily. Furthermore, the patients were asked to point out how they felt during the day (happiness score). They could choose from pictures, ranging from a very sad looking face (score 1) to very happy looking face (score 5).

A positive effect on pain and comfort was defined as an improvement of at least two points for pain and comfort and at least one point for child's happiness scores.

GMFCS and Pediatric Evaluation of Disability Inventory (PEDI) were only tested prior to starting the screening treatment.

Statistical analysis

Due to the explorative setup of this study and the small number of patients we limited ourselves to descriptive results and did not perform further statistical analysis.

RESULTS

Four patients (three male, one female) with an average age of 12 years 6 months (range 8 years 9 months to 17 years 10 months) were included in this pilot study. An overview of their characteristics can be found in Table 1. All patients were wheelchair bound and fully dependent in daily life for self-care and mobility (GMFCS level V and PEDI score for assistance in self-care and mobility 0). Table 1 shows their characteristics and individual effects of ITB treatment.

Table 1	. Patien	ts chara	icteristics ar	nd effects of ITB tre	eatment						
Case	Sex	Age (y)	Catheter ^a	Complications	Interventions	Reported problems	Effect	BAD	Pain	Comfort	Happiness
-	ш	14	Th 8	Nausea	Dosage decrease	 Wheelchair transfer Sitting in wheelchair Communication^b 	‡ ‡ ‡	+	+	‡	+
7	Σ	17	Th7	Nausea, CSF leakage	Dosage decrease, bed rest	 Dressing Hyperextension in lift Dressing^c 	ррр + + +	‡	Ш	‡	+++
m	Σ	σ	Th10	Catheter block, Meningitis	Catheter removal, antibiotics	1. Sleeping 2. Dressing 3. Speaking	‡	‡	‡	11	+++
4	Σ	ø	CG	CSF leakage	Bed rest, Catheter removal	 Hygienic care Sitting in wheelchair Speaking 	+ + p + + +	р + +		T	
Abbrev on X-ra improv	iations: (y; ^b with ement, =	CSF, cer wizard; = no cha	ebrospinal fl ° Socks and nge; -, 1 poi	uid; C, cervical; ITB, shoes; ^d No baselin nt worsening;, ≥2	intrathecal bac e values, chang 2 points worsen	lofen; F, female; M, male; mo, e compared to blinded place ing.	, monts; T bo treatm	h, thora 1ent; +,	cic; yr, y 1 point i	ears. ª Cath improveme	eter tip position nt, ++ ≥2 points

Chapter 3

Effect on daily care

Problems of daily care reported were problems with: Dressing (n=2), speaking (n=2), hyperextension of head and trunk when sitting (n=2), hyperextension of head and trunk in lift (n=1), communication problems with speech computer due to decreased head control (n=1), transfer from bed to chair (n=1), putting socks and shoes on (n=1), sleeping (n=1) and washing (n=1). During ITB trial treatment, all problem scores decreased (average of 4.7 points). During blinded treatment, an increase in problem scores was seen during both the blinded placebo and ITB treatment (respectively 1.4 and 0.5 points) but scores remained well below baseline scores (respectively with 3.3 and 4.2 points).

Effect on dystonia

During ITB trial treatment mean BAD scores decreased by 72% (19.6-5.4). During blinded treatment, an increase in BAD scores was found during both the blinded placebo and ITB treatment (respectively to 63% and 54% of baseline scores).

Effect on pain and comfort

Pain scores decreased significantly in two children, stayed low in one child and increased in another child. The last child experienced complications, explaining the increase in pain scores. Average pain scores decreased during ITB treatment with 1.1 points, but with 3.0 points if excluding the patient with complication induced pain. During blinded placebo treatment pain scores increased with 0.5 points. During blinded ITB treatment, pain scores were 2.6 points lower as at baseline.

Comfort was assessed in two manners. Parents indicated comfort to increase during ITB treatment by an average of 1.1 points. Comfort decreased in the child who also experienced an increase in pain. During blinded placebo and ITB treatment, average comfort increased by 0.5 points further. Child happiness scores increased by 2.3 points during ITB treatment.

One child, for whom parents indicated no change in comfort, indicated a large positive change (4 points). One other child, who was in pain, indicated a decrease in happiness scores. During blinded placebo treatment happiness scores decreased slightly but were still above baseline. During blinded ITB treatment scores maintained similar as during non-blinded treatment.

Complications

Several complications occurred during trial treatment: Cerebrospinal fluid (CSF) leakage occurred in two patients, which resulted in headache, nausea and premature abortion

of the blinded trial phase in one patient and in an extended hospital admission in the other patient. One child received ITB via an external catheter with bolus injections, the other via a continuous micro-infusion pump. Furthermore, in one patient two previous trial treatments using an external catheter were complicated with catheter blockage and meningitis.

DISCUSSION

In this pilot study we found clear effects of ITB treatment in patients with dystonic CP. Previously, improvement in ease of positioning in wheelchairs^{12,14,16} and ease of care including hygiene, dressing and feeding have been reported with ITB treatment in patients with dystonic CP.^{12,16} Furthermore speech has been reported to improve⁶.

Few authors looked at the attainment of individualized goals^{17,18}. After six months of ITB treatment, goals were attained in 70%, measured by Goal Attainment Scaling. Also scores for satisfaction and performance domains increased, as measured with the Canadian Occupational Performance Measure.¹⁸ Similar to Hoving et al., we used a VAS to measure the effect on individualized problems of daily care during the trial treatment phase.¹⁸ Problems with sitting, dressing, transfer, speech, head control and hygienic care were reported by caregivers. During ITB trial treatment problem scores decreased. Hoving et al. found the same effect in seventeen children with CP of which five had the spastic-dyskinetic type of CP.¹⁷

We expected an increase in problem scores during placebo treatment but remarkably the scores did not change significantly after three days of placebo treatment. They were comparable to scores during blinded ITB treatment. Furthermore, even though dystonia (BAD) scores increased during placebo treatment, they stayed below baseline values after three days of blinded placebo treatment in one patient.

One case report supports our findings.¹⁹ In this case report, a 29 year old man with a history of traumatic brain injury ten years before, received a single ITB bolus. The authors found a prolonged anti-spastic effect measured by surface electromyography. A neural upregulation mechanism is suggested as the reason for this prolonged effect.¹⁹ We agree with the authors that the prolonged effect of ITB cannot be caused by remaining baclofen in the cerebrospinal fluid (CSF) since baclofen in the CSF has a half-life of 1 -5 h¹⁹ and three days is well past the expected washout period. The idea of upregulation of GABA surface receptors due to long term underexposure of spinal cord GABA-receptors causing an increased sensitivity to a bolus ITB is therefore plausible but has never been proven. Recently Lee et al²⁰ used the radiotracer ¹⁸F-fluoroflumazenil, a

central benzodiazepine receptor antagonist to calculated the GABA_A receptor binding potential. They found an increased receptor binding potential in patients with bilateral spastic cerebral palsy in the motor cortex area representing the lower extremities. They hypothesized that upregulation of GABA_A receptors might be caused by a neuronal plasticity process compensating for reduced GABAergic input due to white and/or gray matter loss.

In this study GMFCS and PEDI were only tested prior to starting the screening treatment. We did not expect improvement in dependency on others for personal care and mobilization. This assumption is based on the results of previous studies, which did not show change in the degree of autonomy in carrying out everyday activities during ITB treatment.^{6,14,16,18}

Consistent with other studies we found significant decrease of dystonia. Even though some patients experienced serious illness due to complications of trial treatment, we are confident we observed an anti-dystonic effect in these patients. This is confirmed by the long term effect of ITB in these patients. With continuous ITB treatment the effect on dystonia may remain to be present for many months to years of follow-up.^{6,11,12,14,16,18, 21} Previous studies reported improvement in comfort and mood, during continuous ITB treatment,^{12,14,16} which resulted in an improved quality of life in around 85% of patients.^{6,13} Besides comfort rated by caregivers, we feel it is important to furthermore include the child's happiness scores as a measure of comfort since the patient and caregivers can disagree. This phenomenon occurred in one of our patients.

As can be expected, changes in pain and comfort scores are mostly related. Happiness scores and pain improved during ITB treatment in two patients but in one other patient, the side effect of CSF leakage, causing headache and nausea, caused an increase of pain and a decrease in comfort.

A main concern of ITB treatment are the complications that may occur during ITB treatment. Infections, CSF leaks and catheter problems are the most common complications during trial treatment and continuous treatment.^{13,22,23} In our study, treatment had to be stopped prior to ending the study in three events in two patients due to complications of CSF leakage, catheter dysfunction or meningitis. One other patient had an extended admission to the pediatric ward due to CSF leakage. Most cases occurred during treatment via an external catheter. Nausea occurred in two patients as a side effect of ITB and decreased when lowering the dosage. Despite side-effects and complications during this study, a programmable pump was implanted in all patients.



This study has several limitations. Most important, the number of patients in this pilot study is low and the collected data are not complete, mostly because two patients ended the study prematurely. However, we think that the results (including complications) can be very useful to design a larger prospective study.

CONCLUSION

In this pilot study we showed that intrathecal baclofen treatment decreases dystonia and alleviates daily care in patients with dystonic CP. However, the trial period with an external spinal catheter is hampered by serious complications, and in our opinion, the use of external spinal catheters should be avoided if possible. Due to the small number of patients and incomplete data, further research focused on the problems in daily care with a larger study population and prolonged ITB treatment is needed to further specify the effect of ITB in this population.

ACKNOWLEDGEMENTS

We would like to thank the Phelps stichting voor spastici (Project number: 99.047) for the financial support.
REFERENCES

- Himmelman K, Hagberg G, Wiklund LM, Eek MN, Uvebrant P. Dyskinetic cerebral palsy: a population-based study of children born between 1991 and 1998. Dev Med Child Neurol 2007;49:246-51.
- SCPE. Prevalence and characteristics of children with cerebral palsy in Europe. Dev Med Child Neurol 2002;44:633-40.
- Kyllerman M, Bager B, Bensch J, Bille B, Olow I, Voss H. Dyskinetic cerebral palsy. I. Clinical categories, associated neurological abnormalities and incidences. Acta Paediatr Scand 1982;71:543-50.
- Cans C, Dolk H, Platt MJ, Colver A, Prasaukiene A, Krageloh-Mann I. Recommendations from the SCPE collaborative group for defining and classifying cerebral palsy. Dev Med Child Neurol 2007;49:35-8.
- SCPE. Surveillance of cerebral palsy in Europe: a collaboration of cerebral palsy surveys and registers. Dev Med Child Neurol 2000;42:816-24.
- Albright AL, Barry MJ, Shafron DH, Ferson SS. Intrathecal baclofen for generalized dystonia. Dev Med Child Neurol 2001;43:652-7.
- Jankovic, J. Treatment of hyperkinetic movement disorders. Lancet Neurol 2009;8:844-56.
- Saint Hilaire MH, Burke RE, Bressman SB, Brin MF, Fahn S. Delayed-onset dystonia due to perinatal or early childhood asphyxia. Neurology 1991;41:216-22.
- Sanger TD, Bastian A, Brunstrom J, Damiano D, Delgado M, Dure L, et al. Child Motor Study Group. Prospective openlabel clinical trial of trihexyphnidyl in children with secondary dystonia due to cerebral palsy. J Child Neurol 2007;22:530-7.

- Narayan RK, Loubser PG, Jankovic J, Donovan WH, Bontke CF. Intrathecal baclofen for intractable axial dystonia. Neurology 1991;41:1141-2.
- Albright AL. Intrathecal baclofen in cerebral palsy movement disorders. J Child Neurol 1996;11:S29-35
- Albright AL, Barry MJ, Painter MJ, Shultz B. Infusion of intrathecal baclofen for generalized dystonia in cerebral palsy. J Neurosurg 1998;88:73-6.
- 13. Albright AL. Intrathecal baclofen for childhood hypertonia. Childs Nerv Syst 2007;23:971-9.
- Butler C, Butler Campbell S. Evidence of the effects of intrathecal baclofen for spastic and dystonic cerebral palsy. Dev Med Child Neurol 2000;42:634-45.
- Barry MJ, VanSwearingen JM, Albright AL. Reliability and responsiveness of the Barry-Albright dystonia scale. Dev Med Child Neurol 1999;41:404-11.
- Motta F, Stignani C, Antonello CE. Effect of intrathecal baclofen on dystonia in children with cerebral palsy and the use of functional scales. J Pediatr Orthopedics 2008;28:213-7.
- Hoving MA, van Raak EPM, Palmans LJ, Sleypen FAM, Vles JSH. Intrathecal baclofen in children with spastic cerebral palsy; a double-blind, randomized, placebo-controlled, dose finding study. Dev Med Child Neurol 2007;49:654-9.
- Ward A, Hayden S, Dexter M, Scheinberg A. Continuous intrathecal baclofen for children with spasticity and/or dystonia: goal attainment and complications associated with treatment. J Paediatr Child Health 2009;45:720-6.
- Baguley IJ,BaileyKM, Slewa-YounanS. Prolongedanti-spasticity effects of balus intrathecal baclofen. Brain Inj 2005;19:545-8.

- 20. Lee JD, Park H-J, Park ES, Oh M-K, Park B, Rha D-W, et al. Motor pathway injury in patients with periventricular leucomalacie and spastic diplegia. Brain 2011;134:1199-210.
- 21. Albright AL, Barry MJ, Fasick P, Barron W, Shultz B. Continuous intrathecal baclofen infusion for symptomatic generalized dystonia. Neurosurgery 1996;38:934-9.
- 22. Albright AL, Ferson SS. Intrathecal baclofen therapy in children. Neurosurg Focus 2006;21:e3.
- 23. Guillaume D, Van Havenbergh A, Vloeberghs M, Vidal J, Roeste G. A clinical study of intrathecal baclofen using a programmable pump for intractable spasticity. Arch Phys Med Rehabil 2005;86:2165-71.

ITB in secondary dystonia: a pilot study

CHAPTER 4

Intrathecal baclofen treatment in dystonic cerebral palsy: a randomized clinical trial: the IDYS trial

LA Bonouvrié, JG Becher, JSH Vles, K Boeschoten, D Soudant, V de Groot, WJR van Ouwerkerk, RLM Strijers, E Foncke, JJ Geytenbeek, PM van de Ven, O Teernstra, RJ Vermeulen

BMC Pediatrics 2013; 13: 175-182

ABSTRACT

Background Dystonic cerebral palsy is primarily caused by damage to the basal ganglia and central cortex. The daily care of these patients can be difficult due to dystonic movements. Intrathecal baclofen treatment is a potential treatment option for dystonia and has become common practice. Despite this widespread adoption, high quality evidence on the effects of intrathecal baclofen treatment on daily activities is lacking and prospective data are needed to judge the usefulness and indications for dystonic cerebral palsy. The primary aim of this study is to provide level one clinical evidence for the effects of intrathecal baclofen treatment on the level of activities and participation in dystonic cerebral palsy patients. Furthermore, we hope to identify clinical characteristics that will predict a beneficial effect of intrathecal baclofen in an individual patient.

Methods/design A double blind placebo-controlled multi-center randomized clinical trial will be performed in 30 children with dystonic cerebral palsy. Patients aged between 4 and 25 years old with a confirmed diagnosis of dystonic cerebral palsy, Gross Motor Functioning Classification System level IV or V, with lesions in the cerebral white matter, basal ganglia or central cortex and who are eligible for intrathecal baclofen treatment will be included. Group A will receive three months of continuous intrathecal baclofen treatment and group B will receive three months of placebo treatment, both via an implanted pump. After this three month period, all patients will receive intrathecal baclofen treatment will be the effect on activities of and participation in daily life measured by Goal Attainment Scaling. Secondary outcome measurements on the level of body functions include dystonia, spasticity, pain, comfort and sleep-related breathing disorders. Side effects will be monitored and we will study whether patient characteristics influence outcome.

Discussion The results of this study will provide data for evidence-based use of intrathecal baclofen in dystonic cerebral palsy.

Trial registration Nederlands Trial Register, NTR3642

Keywords Cerebral palsy, Dystonia, Dyskinesia, Goal attainment scaling, Intrathecal baclofen, Randomized controlled trial

BACKGROUND

Dystonic cerebral palsy

Cerebral palsy (CP) is a group of disorders caused by non-progressive disturbances that occurred in the developing fetal or early infant brain. The classification of cerebral palsy includes the classical neurological terms for central motor disorders: spasticity, dyskinesia and ataxia.^{1,2} Dyskinesia can be further differentiated into dystonia and choreoathetosis.² Dystonia is described as an abnormal pattern of posture and/or movement that is involuntary, uncontrolled, recurring and occasionally stereotyped. These movements can interfere with daily care and may be painful and uncomfortable. The dyskinetic form of cerebral palsy, including the dystonic form, is in most patients caused by lesions in the basal ganglia. Additional lesions of the central cortex are found in some cases. This type of brain damage is a common pattern in asphyxiated infants born at term.³

Treatment

The results of pharmacological treatment of severe dystonic CP have been rather disappointing. Positive effects on dystonia with levodopa, anticholinergic drugs or muscle relaxants including benzodiazepines and baclofen, have been reported by some authors.^{4,5} In addition, Albright and co-workers described an antidystonic effect of intrathecal baclofen (ITB) treatment. Despite the fact that studies on the effects of ITB on dystonic cerebral palsy are limited in number and the level of evidence is low, ITB is now common practice in the treatment of severe dystonic CP.

Not all patients are eligible for ITB treatment. The following criteria apply: 1. The etiology is preferably known; 2. Management of aggravating factors, such as pain and discomfort, should be optimal; 3. Other treatment options should have been explored. Oral pharmacological treatment with levodopa, anticholinergic drugs or muscle relaxants including benzodiazepines and baclofen must have been attempted and must have resulted in high oral dosages with either unacceptable side effects or insufficient efficacy; 4. The movement disorder should be so severe that it interferes with activities of daily life or quality of life; 5. Treatment goals should be clear and applicable and, to avoid disappointment, it is important that patients and parents understand these goals; 6. Patients and parents should be motivated and able to adhere to the requirements of treatment, such as the frequent pump fillings and checkups in the outpatient clinic; 7. Patients should have sufficient body size to allow pump implantation.⁶

Objectives

The primary objective of the present study is to show whether ITB treatment improves activities of and participation in daily life (for example dressing, transfer, sitting in a wheelchair, hygienic care, speech) in dystonic CP patients. Certain patient characteristics, such as the location and severity of MRI lesions and Gross Motor Functioning Classification Score (GMFCS) level, might influence the effects of ITB treatment in patients with dystonic CP. Therefore, we will study whether individual patient characteristics (GMFCS, gender, age, MRI findings and co-medication) influence outcome and could be used in determining the indication for ITB in future patients. A secondary objective is to provide evidence for the effect of ITB on the level of body functions. The relevant clinical questions to be addressed are: 1. Does ITB decrease dystonia? 2. Does ITB decrease spasticity in dystonic patients? 3. Does ITB decrease pain? 4. Does ITB increase comfort? 5. Does ITB influence screening results of sleep-related breathing disorders? 6. What are the side effects of ITB?

METHODS/DESIGN

Study design

The design of the study is a double blind placebo controlled multi-center randomized clinical trial. It will be conducted in the VU University Medical Center (VUMC) in Amsterdam and the Maastricht University Medical Center (MUMC) in Maastricht (both in the Netherlands). The Medical Ethical Committee of the VU University Medical Center approved the study. In both centers local practicability was granted subsequently. Subjects will be included over a period of two and a half years and they will participate in the study for one year. Figure 1 shows the flow scheme for subjects and timing of measurements throughout the study.

Participants

Thirty subjects will be recruited from the outpatient clinics of the pediatric neurology and pediatric rehabilitation departments of the VUMC and the MUMC. Table 1 shows the inclusion and exclusion criteria. We selected ages between 4 and 25 years old because older patients often show secondary complications, such as contractures, that could introduce greater variation into both the effects of treatment and treatment goals. Furthermore, to achieve a homogeneous patient group, only GMFCS IV and V (non-walkers) will be included. All patients and/or their caregivers will sign an informed consent form before participating in the study.



Figure 1. Flow chart subjects

Table 1. Summary of inclusion and exclusion crit	eria
--	------

Inclusion criteria	Exclusion criteria
 Dystonic cerebral palsy GMFCS IV or V Eligible for ITB treatment using criteria for common practice Lesions on MRI (cerebral white matter, basal ganglia, central cortex) Aged 4 to 25 years old Able and willing to complete study protocol Consensus about inclusion 	 Contra-indications for general anesthesia Contra-indications for baclofen Oral pharmacological treatment is sufficient Inadequate knowledge of Dutch language Ventriculoperitoneal drain Other disorders interfering with treatment

Sample size calculation

We will use goal attainment scaling (GAS) as our primary outcome measure. The within group change for the placebo group is anticipated to be 0, with a standard deviation (SD) of 8.5. For the intrathecal baclofen group, the within group change is anticipated to be 12.5, with a SD of 10. To achieve a power of 90% when testing for a difference in means using a two-tailed independent samples t-test and a significance level of 5%, a total of 13 subjects per group will be needed.

In a previous study in the VU University Medical Center, 4 infections occurred in 34 pump implantations⁷. Ward and coworkers reported pump removal in 5.9% of the cases with infections following implantation.⁸ Extrapolating these numbers, inclusion of 30 patients is expected to lead to 3.5 infections and 1.2 pump removals. To prevent the study from becoming underpowered, due to unexpected protocol violations or subject dropout due to complications other than those related to intrathecal baclofen treatment, we will include 15 subjects per group.

Intervention

Included subjects will be randomized in two groups. Group 1 will receive placebo treatment via an implanted micro-infusion pump for three months. Group 2 will receive ITB treatment via an implanted micro-infusion pump for three months. In our opinion, it is unethical to continue placebo treatment for more than three months. If interim analysis shows that subjects in one of the groups have significant disadvantages compared to the other group, the study will be prematurely terminated. On the other hand, we know from experience that dosages for some patients are still being modified three months after starting treatment. As a consequence, the effect of treatment in these patients may not yet be optimal, which may then result in no detectable effect of ITB treatment at that point. We will try to reduce this possible effect through frequent dose modification. Patients will be seen once or twice weekly to increase their dosage, until either a stable dosage or a maximal dosage of 800 μ g/day is achieved.

Subjects will be assessed at three months and the effect of treatment in the two groups compared. Since determining the most beneficial pump setting can take more than three months and because the initial effect may recede in some cases, all subjects will receive subsequent ITB treatment for nine months. Following this 9 month period, clinical evaluation will be performed in order to determine long-term effects (see Figure 1. Flow Chart of Subjects).

Dosage

Patients with dystonia seem to require larger ITB doses than the doses used to treat spasticity.⁹ Albright and colleagues noticed a change in dystonia only after 4 days of continuous infusion.¹⁰ The effective dose of ITB was in the range of 350 to 750 μ g/ day.⁹⁻¹¹ Since a screening period will not be included, we do not know how subjects will respond and which dose will be sufficiently effective. Therefore, the starting dosage will be 50 μ g/24 h.

Dosage can be increased by 10-20% daily. When subjects are discharged from the hospital following pump implantation, they will be seen once a week or once every two weeks for dosage modification. After three weeks of increasing dosage, with disappearance of the effect of the increase, daily boluses will be administered to a maximum of four times a day, with a minimal interval of four hours. For the blinded part of the study, a maximum dosage of 800 μ g/day will be administered. This is a reasonable estimation of the maximal dosage after three months. Since treatment during the study is blinded, the physician in charge of regular follow-up after pump implantation, including pump filling and dosage adjustments, will be unaware of the subject's allocated group. To provide optimal treatment, pump settings will be individually altered to the extent that the physician would consider necessary, based on the findings of physical examination and parental and patient interview, as were the patient not participating in the study. As a consequence, the pump settings of subjects in the placebo group will be altered as if the patient was receiving ITB treatment.

Lack of response to ITB can be caused by pump or catheter dysfunction. Placebo treatment may also cause lack of effect and for this reason no action will be taken during the first three months of the study in cases where pump or catheter dysfunction would normally be suspected. An exception will be made when the condition of the patient requires otherwise, such as in the case of signs of withdrawal. One week after pump implantation, all patients will undergo X-ray of the spine to determine the position of the catheter tip. The catheter should be placed approximately at the fourth cervical level, and at least above the level of Th1.

Outcome measures

Outcome measurements are defined on the levels of the International Classification of Functioning, Disability and Health (ICF) model of the World Health Organization (WHO). We distinguish the level of activities and participation and the level of body functions. Body functions are the physiological and psychological functions of body systems. Activities are defined as the execution of a task or action by an individual. Participation is involvement in a life situation.¹²

Primary outcome measure

The primary outcome measurement will be on the level of activities and participation. Goal attainment scaling (GAS) will be used to measure the effect of ITB treatment. Using GAS, achievement of individual set goals can be quantified. This method was introduced for assessing outcomes in mental health settings and has been used in many other areas.^{13,14} Each subject has their own outcome measure, but statistical analysis is possible because they are scored in a standardized way. This procedure is time-consuming and requires about 45 minutes per child.^{15,16} However, after the scale is constructed, it should be possible to score a patient within 10 minutes.¹⁷

The procedure includes the following aspects:

1) Identification of goals

Three main problems of daily care and function will be determined by caregivers. Goals will be set for these problems with the help of the team to ensure that goals are achievable. Goals should be specific, measurable, attainable, realistic and timely (SMART).^{13,14} The target activity, specific support and time period should be specified and performance should be quantified using distance, frequency or time taken to accomplish a task.¹⁴

2) Weighing of goals:

Some goals will be more important for subjects than others.¹³ Goals can be assigned a weighing score by caregivers and a difficulty score by the team.¹⁴ We chose not to assign weighing scores since the weighted and unweighted scores are closely correlated.¹⁸ A value of 1 is applied to 'weight' in the formula described below.

3) Definition of expected outcomes:

Several approaches are described in literature, varying from 5 to 7 point scales.^{14,15} We will use a 6 point scale ranging from -3 to +2, since we wish to include both the possibility of partially achieving the set goal and to avoid a bottoming effect. Baseline scores will be allocated as -2. If the subject achieves the expected level, this is scored as 0. If a subject does not achieve the expected level but shows improvement, this is scored as -1. If they achieve more than the expected outcome, this is scored as +1 for somewhat more and +2 for much more. We chose to

add a score of -3 in case of deterioration. Each goal level will be defined by the investigator so as to be as objective and observable as possible.¹³

4) Scoring goal attainment:

For each assessment, one assessor will make a standardized video recording during trials for each of the three functional ability goals. The recording procedure will be identical for all measurements. Another assessor, blinded for group allocation, will rate the subject's performance from the video recordings. Although the GAS only simulates the subject's own functional setting, parents were convinced that the outcome of scaling was representative of their own setting.¹⁹

Goal attainment scores will be recorded at baseline, after three months of blinded treatment and after nine months of ITB treatment. Goal attainment change scores will be determined by subtracting the baseline score from the outcome score. Subjects who achieve a GAS T-score >50 achieved their goals.⁸ Clinical relevance will be defined as an improvement of at least two points in at least one of the three goals.¹⁹

A single aggregated score (T-score) can be produced by a standardized mathematical formula: Overall GAS = $50+10\Sigma(w_ix_i) / V((1-\rho) \Sigma w_i^2 + \rho(\Sigma w_i)^2)$; W_i is the weight assigned to the goal, x_i is the numerical value achieved (between -3 and +2), ρ (rho) is the expected correlation of the goal scales, which is normally 0.3.^{13,14}

Secondary outcome measures

Dystonia

Videos of all patients will be made using the video protocol described by Monbaliu et al.²⁰ Blinded therapists or physicians will assess all videotapes and rate dystonia using the Barry-Albright dystonia (BAD) scale and the Dyskinesia Impairment Scale (DIS).

The BAD scale is a five point ordinal severity scale to assess secondary dystonia in eight body regions (eyes, mouth, neck, trunk, each upper and lower extremity).²¹ Raters score dystonia as none, slight, mild, moderate or severe. A reduction of 25% or more, in comparison with the baseline score, is considered clinically significant. In our experience, interrater variability is high, but BAD scores are generally accepted as a measurement of dystonia and are widely used to assess dystonia.

Recently, a new instrument to measure dystonia in dyskinetic CP became available, the DIS.²⁰ It consists of two subscales: dystonia and choreoathetosis. Scoring is carried out

in 12 body regions all in two conditions (rest and activity). Both duration and amplitude are evaluated. Since this is a new instrument, we will use the DIS in addition to the BAD.

Electromyography

The DIS has no external validation. We decided that Surface Electromyography (EMG) might provide an impression of muscle activity level underlying dystonia. Therefore, surface EMG measurements will be carried out to determine mean EMG activity in individually determined muscle groups and in multiple conditions such as rest and during activities.

Spasticity

The soleus Hoffmann-reflex (H-reflex) represents excitability of the neural components of the stretch reflex arc.²² The H/M-ratio of the H-reflex represents an increased excitability of soleus motor neurons. The H/M-ratio is increased in subjects with spasticity due to various origins,²³ and it has been shown that the H_{max} decreases significantly after ITB administration in children with spastic cerebral palsy when compared with baseline measurements. This represents a decrease in motor neuron excitability. Furthermore, the response appears to be dose-dependent²². Although not all children tolerate the measurements, feasibility of the H-reflex was 93% in a study by Hoving and co-workers.²²

For clinical assessment of spasticity in children with central motor disorders, the spasticity test (SPAT) is used during standard physical examination. When using the SPAT, we will follow a standardized protocol as described in the guideline for standard physical examination of children with central motor disorders.^{24,25} The difference between the range of motion and the angle of catch will be used as the outcome measure for spasticity.²⁶ The test takes approximately 5 to 8 minutes per limb to perform.²⁵ This test might be difficult to perform in children and adolescents with severe dystonia and its usefulness will become evident during the study.

Pain and comfort

Parents will score pain and comfort on a visual analogue scale (VAS). The VAS is a straight 10 cm long horizontal block consisting of 10 smaller blocks with anchor points at 0 and 10. For pain, a score of 0 represents 'no pain' and a score of 10 represents 'the worst possible pain'. For comfort, a score of 0 will represent being 'very uncomfortable' and a score of 10 is having 'no problems at all'. If patients are able to indicate their mood, it will be scored by pointing out the applicable happy face, with choices of six faces ranging from very happy to very sad.

Sleep-related breathing disorders

Children with CP have a higher risk of sleep-related disorders than typically developing children.²⁷ Bensmail et al. showed that patients with severe spasticity due to spinal cord injury and multiple sclerosis, treated with ITB by bolus administration, showed an increased respiratory disturbance index (the number of apneas/hypopneas per hour of sleep).^{28,29} Polysomnography is the gold standard when measuring sleep-related breathing disorders (SRBD). However, this time-consuming and burdensome diagnostic test is not practical for children participating in research.³⁰ A subscale of the Pediatric Sleep Questionnaire was developed to measure SRBD. This scale consists of 22 items and can be completed in five minutes. Sensitivity and specificity are high (81% and 87%). The scale is positive for a high risk of SRBD when there are 8 or more positive answers to the 22 question items (\geq 33%).³⁰ As the burden of a questionnaire is low for the patients and their families, as compared to polysomnography, we will use this questionnaire to determine if ITB changes the risk of SRBD.

Classification

For classification of the severity of motor abilities we will use the GMFCS [31,32] and the Manual Ability Classification System (MACS).^{33,34} GMFCS and MACS classification will be scored at baseline, 3 months and at twelve months. In a study by Voorman and co-workers, 74% of children with CP had restrictions in communication [35]. Since many children with severe CP (GMFCS IV and V) cannot speak, assessment of language abilities cannot be based on speech production. In addition, language comprehension skills are difficult to assess in children with severe CP. Therefore, we will use the Computer-Based Instrument for Low motor Language Testing (C-BiLLT) to measure comprehension of spoken language at baseline. The validity of this instrument has been tested.³⁶ We do not expect changes in outcomes on the C-BiLLT with ITB treatment since comprehension of spoken language is highly correlated with cognition, and cognition is not effected by ITB treatment. We will use the outcome of the C-BiLLT as a patient characteristic.

Magnetic resonance imaging (MRI) will be used to classify the severity of damage to the gray matter structures (cortex and basal ganglia) and white matter (loss of white matter and gliosis). The severity of brain damage on MRI can be classified in three groups; mild, moderate and severe. The mild pattern includes involvement of nucleus lentiformis and ventro-lateral thalamus, the moderate pattern includes additional involvement of the peri-central region and the severe pattern includes additional involvement of the entire thalamus and hippocampus. In the mild and moderate damage groups, infants suffer from the dyskinetic form of cerebral palsy, whereas the severe type of damage frequently produces purely spastic paresis.³ MRI studies of the brain are necessary to

confirm the diagnosis of cerebral palsy. If MRI studies have been conducted previously, these studies will be critically assessed. If the quality is good, the MRI will be accepted and no further imaging is needed. If the quality of the MRI images is poor, the patient will undergo a new MRI including diffusion tensor imaging (DTI). If a patient had not yet undergone a MRI, one will be made including DTI. Adding DTI images to a regular MRI scan will require an extra scanning time of approximately four minutes. In this patient category MRI imaging has to be done under general anesthesia, since dystonic movements interfere with MRI quality.

Other study parameters

To assess the safety of ITB treatment, side effects and complications will be closely monitored. The complication rate will be calculated by dividing the number of complications by the duration in years of pump implantation.

Functional abilities at baseline will be assessed by the Paediatric Evaluation of Disability Inventory (PEDI) and will be repeated after three and twelve months. The PEDI was developed for children from six months to seven and a half years of age and assesses skills in mobility, self-care and social function. It can also be used for older children if their functional abilities are expected to be below the level of a child of seven and a half years old. The functional scales indicate if children with disability are able or unable to perform certain activities. Separate measures assess the degree of caregiving assistance and equipment modification that is needed to accomplish complex functional skills. Scores on the caregiver assistance scale are noted on a range from independent to maximal assistance and modification scores are scored as none, child-oriented, rehabilitation-oriented or extensive.^{37,38} The PEDI will be administered through direct assessment by a therapist.

Randomization, blinding and treatment allocation

Subjects will be randomized in two groups by block randomization. Randomization will be done by the pharmacies of the VUMC and the MUMC. The pharmacist will be the only holder of the code for randomization. In case of emergency, the code will be accessible 24 hours per day and 7 days per week via either the pharmacist of the VUMC or MUMC or by opening a sealed envelope containing the subject's group, available at the departments concerned. Allocation will be concealed, and the researchers, assessors and the physician responsible for pump filling will be blinded.

Premature termination

Withdrawal of individual subjects

Subjects can end participation in the study at any time without providing a reason and without consequences for their future treatment in the clinic. The investigator can decide to withdraw a subject from the study for urgent medical reasons. Subjects withdrawn from the study will continue their regular follow-up outside of the study protocol. Subjects will be replaced if they withdraw before pump implantation has taken place. Subjects who withdraw after pump implantation has taken place will not be replaced. With subject agreement, a final assessment will take place before definite ending of participation. These subjects will not be included in the analysis but we will present a fact sheet including their information.

Data safety monitoring board

A data safety monitoring board will be formed and will meet periodically to review aggregate and individual subject data related to safety, data integrity and overall conduct of the trial. They will assess the risk/benefit balance, including a statistical analysis if necessary. Depending upon this assessment, the board will provide recommendations to continue, adapt or terminate the trial.

Statistics

Descriptive statistics

Patient characteristics will be described. Gender distribution will be compared between the ITB and placebo group using a Chi-square test. Mean age between the ITB and placebo group will be compared using an independent samples t-test. Means and standard deviation (SD) of the PEDI scores, C-BiLLT scores, GAS t-score, BAD score, SPAT score and mean EMG activity at baseline will be tabulated. Means and SD of the GAS t-score, BAD score, DIS score, SPAT score and mean EMG activity at follow up will be separately tabulated per group.

Univariate analysis

The GAS t-scores, BAD score, SPAT score and EMG activity in the placebo and ITB group will be compared using an independent samples t-test. If the assumption of normality appears not to be valid, the non-parametric Mann–Whitney test will instead be used. A p-value of 0.05 is considered statistically significant for this primary analysis. To assess within-group changes in means between follow-up times and baseline GAS t-scores, BAD score, DIS score, SPAT score and EMG activity, a paired t-test or non-parametric Wilcoxon test will be used (depending on whether the normality assumption is valid). In these analyses, a p-value of 0.05 is considered to be statistically significant.



Multivariate analysis

A multivariate analysis will be used to determine the effect of ITB treatment (primary: functional outcome; secondary: dystonia and the interaction with GMFCS, MACS, MRI classification and use of co-medication).

DISCUSSION

We anticipate that the results of this study will allow evidence-based use of intrathecal baclofen in dystonic cerebral palsy.

ABBREVIATIONS

CP: Cerebral palsy; GMFCS: Gross motor functioning classification system; VUMC: VU University medical center; MACS: Manual ability classification System; EMG: Electromyography; MUMC: Maastricht University medical center; ITB: Intrathecal baclofen; MRI: Magnetic resonance imaging; SD: Standard deviation; PEDI: Pediatric evaluation of disability inventory; RM: Repetitive movements; GAS: Goal attainment scaling; BAD: Barry Albright dystonia scale; DIS: Dyskinesia impairment scale; C-BiLLT: Computer based instrument for low motor language testing; dti: Diffusion tensor imaging; VAS: Visual analogue scale; ICF: International classification of functioning, disability and health.

ACKNOWLEDGEMENTS

We gratefully acknowledge the Phelps Stichting voor Spastici (nr 2011037), the Johanna KinderFonds and the Kinderrevalidatie Fonds de Adriaan Stichting (both nr 2011/0035) and the RevalidatieFonds (nr R2011032) for funding of this project.

REFERENCES

- Hagberg B, Hagberg G, Olow I, von Wendt L: The changing panorama of cerebral palsy in Sweden. V. The birth year period 1979–82. Acta Paediatr Scand 1989, 78:283–290.
- Bax M, Goldstein M, Rosenbaum PL, Levinton A, Paneth N, Dan B, Jacobsson B, Damiano D: Proposed definition and classification of cerebral palsy, April 2005. Dev Med Child Neurol 2005, 47:571–576.
- Krägeloh-Mann I, Helber A, Mader I, Staudt M, Wolff M, Groenendaal F, DeVries L: Bilateral lesions of thalamus and basal ganglia: origin and outcome. Dev Med Child Neurol 2002, 44:477–484.
- 4. Jankovic J: Treatment of hyperkinetic movement disorders. The Lancet Neurology 2009, 8:844–856.
- Sanger TD, Bastian A, Brunstrom J, Damiano D, Delgado M, Dure L, Gaebler-Spira D, Hoon A, Mink JW, Sherman-Levine S, Wlty LJ: Child Motor Study Group. Prospective open-label clinical trial of trihexyphnidyl in children with secondary dystonia due to cerebral palsy. J Child Neurol 2007, 22:530–537.
- Dan B, Motta F, Vles JS, Vloeberghs M, Becher JG, Eunson P, Gautheron V, Lutjen S, Mall V, Pascual-Pascual SI, Pauwels P, Roste GK: Consensus on the appropriate use of intrathecal baclofen (ITB) therapy in paediatric spasticity. Eur J Paediatr Neurol 2010, 14:19–28.
- van Hulst BM, Tel PA, de Groot V, van Ouwerkerk WJ, Vermeulen RJ, Becher JG, Peerdeman SM: Complicaties bij kinderen met intrathecale baclofentherapie en (gerelateerde) verzorgertevredenheid. Tijdschrift voor kindergeneeskunde 2009, 77:191–197.

- Ward A, Hayden S, Dexter M, Scheinberg A: Continuous intrathecal baclofen for children with spasticity and/or dystonia: Goal attainment and complications associated with treatment. J Paediatr Child Health 2009, 45:720–726.
- Albright AL, Barry MJ, Fasick P, Barron W, Shultz B: Continuous intrathecal baclofen infusion for symptomatic generalized dystonia. Neurosurgery 1996, 38:934– 939.
- Albright AL, Barry MJ, Painter MJ, Shultz B: Infusion of intrathecal baclofen for generalized dystonia in cerebral palsy. J Neurosurg 1998, 88:73–76.
- Albright AL, Barry MJ, Shafron DH, Ferson SS: Intrathecal baclofen for generalized dystonia. Dev Med Child Neurol 2001, 43:652–657.
- World Health Organisation: International Classification of Functioning. Geneva: Disability and Health; 2001. Ref Type: Catalog.
- Turner-Stokes L: Goal attainment scaling (GAS) in rehabilitation: a practical guide. Clin Rehabil 2009, 23:362–370.
- Bovend'Eerdt TJH, Botell RE, Wade DT: Writing SMART rehabilitation goals and achieving goal attainment scaling: a practical guide. Clin Rehabil 2009, 23:352– 361.
- Steenbeek D, Ketelaar M, Galama K, Gorter JW: Goal attainment scaling in paediatric rehabilitation: a critical review of the literature. Dev Med Child Neurol 2007, 49:550–556.
- Steenbeek D, Ketelaar M, Galama K, Gorter JW: Goal attainment scaling in paediatric rehabilitation: a report on the clinical training of an interdisciplinary team. Child: care, health and development 2008, 34:521–529.

- Steenbeek D, Ketelaar M, Lindeman E, Galama K, Gorter JW: Interrater reliability of goal attainment scaling in rehabilitation of children with cerebral palsy. Arch Phys Med Rehabil 2010, 91:429–435.
- Choate R, Smith A, Cardillo JE, Thompson L: Training in the use of Goal Attainment Scaling. Community Ment Health J 1981, 17:171–181.
- 19. Steenbeek D, Meester-Delver A, Becher JG, Lankhorst GJ: The effect of botulinum toxin type A treatment of the lower extremity on the level of functional abilities in children with cerebral palsy: evaluation with goal attainment scaling. Clin Rehabil 2005, 19:274–282.
- 20. Monbaliu E, Ortibus E, De Cat J, Dan B, Heyrman L, Prinzie P, De Cock P, Feys H: The Dyskinesia Impairment Scale: a new instrument to measure dystonia and choreoathetosis in dyskinetic cerebral palsy. Dev Med Child Neurol 2012, 54:278–283.
- 21. Barry MJ, VanSwearingen JM, Albright AL: Reliability and responsiveness of the Barry-Albright Dystonia scale. Dev Med Child Neurol 1999, 41:404–411.
- 22. Hoving MA, van Kranen-Mastenbroek VHJM, van Raak EPM, Spincemaille GHJJ, Hardy ELM, Vles JSH: On behalf of the Dutch Study Group on Child Spasticity. Placebo controlled utility and feasibility study of the H-reflex and flexor reflex in spastic children treated with intrathecal baclofen. Clin Neurophysiol 2006, 117:1508–1517.
- Bour LJ, Ongerboer de Visser BW, Koelman JHTM, van Bruggen GJ, Speelman JD: Soleus H-reflex tests in spasticity and dystonia: A computerized analysis. J Electromyogr Kinesiol 1991, 1:9–19.

- Becher JG, Doorenbosch C, Folmer K, Scholtes VAB, Voorman JM, Wolterbeek N: Handleiding standaard lichamelijk onderzoek bij kinderen met een centraal motorische parese. Amsterdam: Reed business; 2011.
- 25. Scholtes VAB, Dallmeijer AJ, Becher JG: The spasticity test: a clinical instrument to measure spasticity in children with cerebral palsy. The effectiveness of multilevel botulinum toxin type A and comprehensive rehabilitation in children with cerebral palsy [dissertation]. Amsterdam: VU University Medical Center; 2007.
- Scholtes VAB, Becher JG, Beelen A, Lankhorst GJ: Clinical assessment of spasticity in children with cerebral palsy: a critical review of available instruments. Dev Med Child Neurol 2006, 48:64–73.
- Sandella DE, O'Brien LM, Shank LK, Warschausky SA: Sleep and quality of life in children with cerebral palsy. Sleep Med 2011, 12:252–256.
- Bensmail D, Quera Salva MA, Roche N, Benyahia S, Bohic M, Denys P, Bussel B, Lofaso F: Effect of intrathecal baclofen on sleep and respiratory function in patients with spasticity. Neurology 2006, 67:1432– 1436.
- 29. Bensmail D, Marquer A, Roche N, Godard A, Lofaso F, Quera Salva MA: Pilot study assessing the impact of intrathecal baclofen administration mode on sleeprelated respiratory parameters. Arch Phys Med Rehabil 2012, 93:96–99.
- Chervin RD, Hedger K, Dillon JI, Pituch KJ: Pediatric Sleep Questionnaire (PSQ): validity and reliability of scales for sleepdisordered breathing, snoring, sleepiness and behavioral problems. Sleep Med 2000, 1:21–32.

- Palisano RJ, Hanna SE, Rosenbaum PL, Russel DJ, Walter EP, Raina PS, Galuppi BE: Validation of a model of gross motor function for children with cerebral palsy. Phys Ther 2000, 80:974–985.
- Wood E, Rosenbaum PL: The gross motor function classification system for cerebral palsy: a study of reliability and stability over time. Dev Med Child Neurol 2000, 42:292–296.
- 33. Eliasson AC, Krumlinde-Sundholm L, Rösblad B, Beckung E, Arner M, Ohrvall AM, Rosenbaum PL: The Manual Ability Classification System (MACS) for children with cerebral palsy: scale development and evidence of validity and reliability. Dev Med Child Neurol 2006, 48:549–554.
- Morris C, Kurinczuk JJ, Fitzpatrick R, Rosenbaum PL: Reliability of the manual ability classification system for children with cerebral palsy. Dev Med Child Neurol 2006, 48:950–953.
- 35. Voorman JM, Dallmeijer AJ, van Eck M, Schuengel C, Becher JG: Social functioning and communication in children with cerebral palsy: association with disease characteristics and personal and environmental factors. Dev Med Child Neurol 2009, 52:441–447.
- 36. Geytenbeek JJM, Heim MMJ, Vermeulen RJ, Oostrom KJ: Assessing comprehension of spoken language in nonspeaking children with cerebral palsy: application of a newly developed computer based instrument. Augment Altern Commun 2010, 26:97–107.
- Feldman AB, Haley SM, Coryell J: Concurrent and construct validity of the Pediatric Evaluation of Disability Inventory. Phys Ther 1990, 70:602–610.

38. Haley SM, Coster WJ, Ludlow LH, Haltiwanger JT, Andrellos PJ: Pediatric Evaluation of Disability Inventory (PEDI): development, standardization and administration manual. Boston, MA: PEDI Research Group, Department of Rehabilitation Medicine; 1992.

CHAPTER 5

The effect of intrathecal baclofen in dyskinetic cerebral palsy: the IDYS trial

LA Bonouvrié, JG Becher, JSH Vles, RJ Vermeulen, Al Buizer, on behalf of the IDYS study group

Ann Neurol 2019; epub ahead of print

ABSTRACT

Objective Intrathecal baclofen treatment is used for the treatment of dystonia in patients with severe dyskinetic cerebral palsy, however, the current level of evidence for the effect is low. The primary aim of this study was to provide evidence for the effect of intrathecal baclofen treatment on individual goals in patients with severe dyskinetic cerebral palsy.

Methods This multi-centre, randomized, double-blind, placebo-controlled trial was performed at two University Medical Centres in the Netherlands. Patients with severe dyskinetic cerebral palsy (Gross Motor Functioning Classification System level IV-V), aged 4 to 24, eligible for intrathecal baclofen, were included. Patients were assigned by block randomization (2:2) for treatment with intrathecal baclofen or placebo during three months via an implanted micro-infusion pump. The primary outcome was Goal Attainment Scaling of individual treatment goals (GAS T-score). A linear regression model was used for statistical analysis with study site as a covariate. Safety analyses were done for number and type of (serious) adverse events.

Results Thirty-six patients were recruited from January 1st 2013 to March 31st 2018. Data for final analysis were available for 17 in the intrathecal baclofen group and 16 patients in the placebo group. Mean (SD) GAS T-score at three months was 38.9 (13.2) for intrathecal baclofen and 21.0 (4.6) for placebo (regression coefficient 17.8, 95% CI 10.4 to 25.0, p<0.001). Number and types of (serious) adverse events were similar between groups.

Interpretation Intrathecal baclofen treatment is superior to placebo in achieving treatment goals in patients with severe dyskinetic cerebral palsy.

INTRODUCTION

Cerebral palsy (CP) is defined as a group of developmental disorders of movement and posture, attributed to non-progressive disturbances that have occurred in the developing fetal or infant brain. Motor impairments are often accompanied by nonmotor symptoms such epilepsy, secondary musculoskeletal problems, and disturbances of sensation, perception, cognition, communication, and/or behaviour.¹ The reported prevalence is 2.11/1000 live births (95% confidence interval (CI) 1.98 to 2.25) in high income countries, and 2.9 to 3.7 /1000 live births in low income countries.²⁻⁴ Based on the predominant motor disorder, CP is classified in three types: spastic, dyskinetic, and ataxic. After spastic CP (80%), dyskinetic CP is the second most common type of CP.¹

Dyskinesia is the predominant motor disorder in 9-15% of CP patients.^{2,4} It is characterised by involuntary, uncontrolled, recurring, occasionally stereotyped movements with fluctuating muscle tone.⁵ Dyskinesia is subdivided into dystonia and choreo-athetosis. Dystonia is characterised by slow involuntary movements, distorted voluntary movements and abnormal postures due to sustained or intermittent muscle contractions. Tone is fluctuating but easily increased (hypertonia).⁵ Choreo-athetosis is featured by fast hyperkinetic movements and tone fluctuation (mainly hypotonia).⁵

The majority of dyskinetic CP patients are severely affected and classified in Gross Motor Functioning Classification System (GMFCS) levels IV and V. These GMFCS levels correspond with having no walking ability.⁶ Dystonia and choreo-athetosis are frequently present simultaneously in dyskinetic CP but dystonia is usually more prominent.⁶ Most treatment options for dyskinetic CP are aimed at decreasing dystonia.⁶

When efficacy of oral medication is insufficient, options requiring neurosurgical intervention such as intrathecal baclofen (ITB) treatment or deep brain stimulation (DBS) are the next step in treatment of dyskinetic CP.^{6,7} DBS is effective in primary dystonia but results in dyskinetic CP have been conflicting.^{6,7} While there is some evidence for the effectiveness of ITB in reducing spasticity in CP, mainly from short term single bolus trial studies, to date, there are only low quality, non-controlled studies, producing low-level evidence for ITB in the reduction of dystonia.^{7,8} Most dyskinetic CP patients for whom ITB is considered aim to gain improvement of dressing, positioning, transfers, pain, and comfort.⁹ There is currently inadequate evidence for the effect of ITB in dyskinetic CP on achievement of individual goals related to quality of life, activities of daily life and participation.⁷

The primary aim of this study is to provide evidence for the effect of ITB on individual treatment goals in patients with severe dyskinetic CP. Secondary aims are to address the effect on dystonia, spasticity, range of motion, pain, comfort, and treatment related complications.

METHODS

Study design

We conducted a multi-centre, randomized, double-blind, placebo-controlled trial at the Amsterdam UMC, location VU University Medical Center (VUMC), Amsterdam, and the Maastricht University Medical Center (MUMC), Maastricht, the Netherlands. The study was approved by institutional review boards at both sites and by the Medical Ethical Review Committee (MERC) of the VU University. The study protocol was previously published and provides additional information about the methods employed.¹⁰ This trial is registered with the Dutch Trial Register, number NTR3642.

Participants

Participants were recruited from the paediatric rehabilitation and/or child neurology outpatient clinic at both sites. In- and exclusion criteria are summarized in table 1. Patients and/or parents/legal guardians gave written informed consent. Measurements were discontinued at signs of discomfort due to the measurements.

At the start of the study in January 2013, participation in the study was with the result that not all eligible patients were included in the trial, though received ITB treatment. The Data Safety Monitoring Board (DSMB) advised the MERC that the study involved an experimental pharmaceutical therapy, since the effect of ITB in dyskinetic CP patients was not conclusively established. The MERC decided that from December 2014, ITB treatment for dyskinetic CP patients could only be provided in the study context, by which all eligible patients were automatically included in the trail, and ITB was not offered as regular treatment.

Table 1. Inclusion and exclusion criteria

Inc	clusion criteria	Exclusion criteria
Inc • •	 Clusion criteria Dyskinetic cerebral palsy GMFCS IV or V (wheelchair users) Lesions on MRI (cerebral white matter, basa central cortex) Aged 4 to 25 years old Eligible for ITB treatment using criteria of corpractice: O Preferably known etiology O Optimal management of aggravating fact pain) O Insufficient effect of oral treatment or unacceptable side effects O Dyskinesia interferes with activities of d quality of life O Treatment goals are clear and applicable O Patients and parents are sufficiently mo adhere to requirements of treatment O Patients have sufficient body size to allog implantation 	Exclusion criteria • Contra-indications for general anesthesia I ganglia, • Contra-indications for baclofen • Oral pharmacological treatment is sufficient ommon • Inadequate knowledge of Dutch language • Deep brain stimulation • Other disorders interfering with treatment aily life or • tivated to
•	Consensus about inclusion	

GMFCS=gross motor functioning classification system. ITB=intrathecal baclofen treatment. MRI=magnetic resonance imaging.

Randomization and masking

Patients were assigned by block randomization (2:2) to receive either ITB or intrathecal placebo via an implanted micro-infusion pump (Medtronic Synchromed[®] II pump, Minneapolis, MN, USA). Randomization was stratified by study site. The last four patients combined from both sites were randomized in one block to assure even groups. All but the pharmacist preparing the study medication were masked for group allocation. A list of group allocation was stored at the pharmacy. Closed envelopes were provided for the study staff to be used for unmasking in case of emergency and after final measurements. Outcome assessors remained masked until after statistical analysis was completed.

Procedures

The pump was implanted in a subfascial or subcutaneous pocket in the lower abdomen depending on local routines and patient characteristics (mostly nutritional status). Considering previous clinical observations suggesting that the site of action for ITB in dyskinetic CP is intracranial,¹¹ the catheter tip was aimed to be placed at the mid cervical level (C4). Best practice surgical techniques were applied, including techniques to

minimize complications such as cerebrospinal fluid (CSF) leakage (48 hours of horizontal bed rest, pressure bandage post-surgery) and infection (pre-operative washing with anti-bacterial soap (until 2017) and from 2018 impregnation/bathing of pump, catheter and pocket with vancomycin, combined with 24 hours intravenous cephazolin starting 30 minutes before first skin incision).^{12,13}

During three months after pump implantation the placebo group received Sodium Chloride (0.9%), and the ITB group received baclofen in the intrathecal pump system. Patients were instructed to continue taking their regular oral medication influencing muscle tone, such as baclofen or trihexyphenidyl, at the pre-operative dose during the whole study.

The starting dose of intrathecal treatment was $50\mu g/24h$ for all patients. Dose increments were applied by the treating physician, who were not study assessors, guided by a dosing schedule developed for the study and based on clinical experience (table 2, figures 1 and 2). Dose was increased at least ten times, except if the effect was deemed satisfactory earlier and personal goals were achieved at a fewer number of increments.

Measurements were performed prior to pump implantation and three months after. Unmasking was done by the treating physician right after finalization of the three-month measurements.

Phase of dose finding	8 years and older (figure 3)	7 years and younger (figure 4)
1. Start treatment at simple continuous mode at 50μg/day	No effect: raise with 20%Some effect: raise with 10%	No effect: raise with 10%Some effect: raise with 5%
2. Starting to notice effect	 When improvement continuous 2 weeks after a 10% raise in dosage, gradually raise dosage further by 10% steps in simple continuous mode When after a few days the effect disappears, raise dosage, 1-2 weeks after last alteration, with 10% (repeat 3 times when needed) When the effect persistently decreases or disappears within 2 weeks of last alteration: start bolus 1 hour before awakening, 15% of the daily dose 	 When improvement continuous 2 weeks after a 5% raise in dosage, gradually raise dosage further by 5% steps in simple continuous mode When after a few days the effect disappears, raise dosage, 1-2 weeks after last alteration, with 5% (repeat 3 times when needed) When the effect persistently decreases or disappears within 2 weeks of last alteration: start bolus 1 hour before awakening, 10% of the daily dose
3. Bolus added	 Continue to raise bolus with 10% until hypotonia at awakening Ask when the effect starts to decrease during the day Add a 2nd bolus of 15% 1 hour before the effect of the bolus decreases More boluses can be added to a maximum of 4 boluses per day 	 Continue to raise bolus with 10% until hypotonia at awakening Ask when the effect starts to decrease during the day Add a 2nd bolus of 10% 1 hour before the effect of the bolus decreases More boluses can be added to a maximum of 4 boluses per day

Table 2. IDYS trial dosing schedule for ITB in dyskinetic cerebral palsy

Chapter 5



Figure 1. Dosing Schedule for age 8 years and older



Figure 2. Dosing schedule for age 7 years and younger

Outcomes

The following patient characteristics were registered: age, sex, GMFCS level, Manual Ability Classification System (MACS) level, level of comprehension of spoken language (determined with the Computer Based instrument for Low motor Language Testing (C-BiLLT),¹⁴ and brain Magnetic Resonance Imaging (MRI) classification.¹⁰ Treatment characteristics registered were catheter tip position and dosage after three months.

The primary outcome measure, achievement of individual goals, was assessed with Goal Attainment Scaling (GAS).¹⁰ Before pump implantation, individual goals were identified by questioning parents and patients. The current situation and the desired outcome for two to three goals were determined and used to extrapolate all other possible outcomes on the GAS scale. An example of a GAS scale is given in table 3. At follow up after three months, goal attainment was assessed, by the same blinded assessor, questioning parents and patients about the current situation of the previously set goals. The outcome was scored for each separate goal. From these separate scores, a single aggregated T-score was produced using a standardized mathematic formula.¹⁰ Subjects who attain a GAS T-score \geq 50 achieved their goals.¹⁰ Furthermore, the number of achieved and partially achieved goals was assessed. Clinical relevance was defined as achievement of at least one goal.⁴

Treatment goals were classified in the domains defined by the International Classification of Functioning, disability and health - Children and Youth version (ICF-CY), developed by the World Health Organization (WHO): body functions and structures, activities and participation, and environmental factors.¹⁵

Secondary outcome measures were assessed both at baseline and at three months after pump implantation.

Dyskinesia was measured using the Barry-Albright dystonia scale (BADS) and Dyskinesia Impairment Scale (DIS). The BADS measures dystonia in different body regions and provides a total dystonia score (range 0-32). A difference of \geq 25% compared to baseline values, has been described to be clinically significant.¹⁰ The DIS measures both dystonia and choreo-athetosis during rest and activities. A total DIS score and subscores for dystonia and choreo-athetosis (total, rest, action) can be computed (range 0-100%).¹⁶ It is not known what the clinically significant cut-off values for the DIS (sub)scores are. There are no test-retest reliability studies or data on responsiveness for either measure.¹⁷ All of our assessors were trained to distinguish dystonia and choreoathetosis. Studies show moderate interrater reliability.¹⁷ Surface electromyography was initially performed with the aim to quantify dystonia by recording activation of individual muscles,¹⁰ but was discontinued after interim evaluation. Technical issues made reliable assessment difficult, and measurements appeared to be stressful for patients.

Spasticity was assessed clinically and electrophysiologically. The spasticity test (SPAT) was used to assess spasticity clinically in hip adductors, knee flexors and extensors, ankle plantar flexors, elbow extensors and flexors.¹⁰ With the SPAT, spasticity is elicited by a passive stretch of the muscle with fast velocity, and scored present when a catch is felt by the examiner. The H/M ratio of the soleus Hoffmann-reflex (H-reflex) was determined, representing the spinal cord neuronal response to an afferent electric stimulus.^{10,18}

The degrees of passive range of motion (ROM) were measured for hip abduction, knee flexion, popliteal angle, ankle dorsiflexion (with flexed and extended knee), elbow flexion and elbow extension.¹⁰

Pain and comfort were scored by parents, using a visual analogue scale (VAS) with scores ranging from 0 to 10. For pain, a score of 0 corresponded with 'no pain' and 10 with 'unsustainable pain'. For comfort, a score of 0 corresponded with 'very uncomfortable' and 10 with 'no discomfort'. If possible, children scored their level of comfort using a faces pain scale corresponding with scores from 0 to 6.¹⁰

The Pediatric Evaluation of Disability Inventory (PEDI) questionnaire was used to assess skills in mobility, self-care and social function.^{10,19}

Participants were asked about their thoughts about group allocation after the measurements, right before unmasking (ITB/placebo/don't know).

All (Serious) Adverse Events (S)(AE)s were assessed. SAEs were submitted to an online research database (toetsingonline.nl), automatically informing the MERC. We furthermore evaluated the change in symptoms related to sleep related breathing disorders (SRBD) using a specific subscale of the Pediatric Sleep Questionnaire ^{10,20}

Statistical analysis

Sample size was calculated on the basis of the primary outcome measure, the difference in GAS T-score between the placebo and ITB group after three months.¹⁰ It was hypothesized that the ITB group would attain a GAS T-score of 50, whereas the score

for the placebo group would not differ from baseline (GAS T-score 22.7). To guarantee a statistical power of 90% at a significance level of 5%, a total of 13 patients per group was needed. Because several complications occurred during the study, inclusion was expanded to 18 patients per group to prevent the study from becoming underpowered.

Baseline data were described with summary statistics. For normally distributed outcome data, linear regression methods were used to compare groups. Baseline values (only applicable for secondary outcome measures) and study site were included as covariates. Where normal distribution was not found, nonparametric methods (Mann-Whitney U test) were used. Safety analysis for number and type of (S)AEs were performed with Pearson Chi Square Test or Mann-Whitney U test. All statistical analyses were performed before group allocation was revealed. Data were analyzed using IBM© SPSS© Statistics Version 22. GAS T-scores and (S)AEs were regularly monitored during the study by an independent DSMB.

Role of the funding source

The funders had no role in the study design, data collection, data analysis, data interpretation, or writing of the article. The corresponding author had full access to all study data and all main authors (LB, JB, JV, RJV, AB) had final responsibility for the decision to submit for publication.

RESULTS

From January 1st, 2013 to March 31st, 2018, a total of 36 patients (30 in Amsterdam UMC, 6 in MUMC) were randomized over the two groups providing 18 patients per group. The trial profile is presented in figure 3. One patient was unmasked before the end of the study period during presentation at the emergency room because of swelling over the pump, possibly due to liquor leakage, 10 days post implantation. Since knowledge on group allocation could influence outcome, GAS, pain and comfort scores were not rated for this patient and therefore were not included in the final analysis. For this patient, all other outcome measures were assessed by a still masked assessor and used for analysis. Brain MRI was available for all patients at the moment of inclusion but for two patients imaging was not present for scoring due to technical problems. Two patients (one in each group) did not tolerate the H-reflex and this measurement was not repeated at their follow up. The catheter tip was placed at a lower level than the aimed C4 due to technical surgical issues in two patients in the ITB group, and four patients in the placebo group.

Chapter 5



Figure 3. Trial profile

Patient characteristics were similar across groups (table 4). Treatment characteristics were similar across groups, except for a significant difference between groups in usage of bolus dosing (p=0.02). Bolus dosing was more frequently applied in ITB (59%) compared to placebo (6%). Mean dosage after three months was 179 μ g/24h (SD 206, range 80-966 μ g) for ITB compared to 170 μ g/24h (SD 70, range 55 to 300 μ g) for placebo (p =0.24).

		ITB group (n=18)	Placebo group (n=18)
Sex	Male	13 (72%)	12 (67%)
	Female	5 (28%)	6 (33%)
Age (years)		13.8 (4.3)	14·7 (4·3)
GMFCS	IV	8 (44%)	5 (28%)
	V	10 (56%)	13 (72%)
MACS	ш	2 (11%)	1(5%)
	IV	4 (22%)	5 (28%)
	V	12 (67%)	12 (67%)
C-BiLLT ^a		65 (16)°	64 (25) ^d
Catheter tip ^b	Th3 or higher	15 (88%)	13 (72%)
	Th4 or lower	2 (12%)	5 (28%)
MRI pattern ^a	Basal ganglia / thalamic lesions	12 (70%)	10 (59%)
	Kernicterus/ globus pallidus	2 (12%)	2 (12%)
	Multicystic encephalomalacia	0 (0%)	1 (6%)
	Periventricular leukomalacia	3 (18%)	4 (23%)

Table 4.	Baseline	characteristics	of the	intrathecal	bacloten	treatment	and	placebo	group
									0

Data are presented as number (%), or as mean (SD). ITB=intrathecal baclofen. n=number. GMFCS=Gross Motor Functioning Classification System. MACS=Manual Ability Classification System. C-BiLLT=Computer Based instrument for Low Motor Language Testing. Th=thorecal. ^aData not available for all randomized patients. ^bData available for all implanted pumps. ^cAge equivalent of 53 to 55 months. ^dAge equivalent of 50 to 52 months.

AS T-scores at three months were significantly different between groups in favor of ITB (Regression Coefficient (RC) 17.8, 95%CI 10.4 to 25.0, Effect size (Beta) 0.672, p<0.001) (table 5). Significant differences were found between groups for percentages of number of goals achieved (figure 4A, p=0.005) and partially achieved (figure 4B, p<0.001). In the ITB group 59% achieved at least one goal, which was considered to be clinically relevant, compared to 13% in the placebo group. Number needed to treat to achieve at least one goal is 2.2. Age and sex as confounders, did not influence the effect. The type of treatment goals was similar in both groups (table 6).

			ITB gr	0up (n=17)	Placebo	o group (n=16)	RC (95%Cl)	Effect size (Beta)	p value
			Baseline	3 months	Baseline	3 months			
Primary	outcome								
GAS T-sc	ore ^a		n.a.	38-9 (13-2)	n.a.	21.0 (4.6)	17.8 (10·4 to 25·0)	0.672	00·0 _{vv}
Seconda	ry outcomes								
DIS	Total		43% (9%)	40% (10%)	41% (13%)	45% (11%)	-6% (-12% to 0%)	-0.28	^^0.045
	Dystonia	Total	67% (11%)	64% (15%)	66% (18%)	72% (11%)	-8% (-14% to -2%)	-0.29	^^0.017
		Rest	58% (16%)	55% (23%)	59% (22%)	68% (12%)	-12% (-22% to -2%)	-0·32	^^0.013
		Activity	73% (12%)	70% (14%)	72% (15%)	75% (12%)	-5% (-13% to 3%)	-0.20	^^0.19
	Choreo- athetosis		19% (15%)	15% (13%)	15% (17%)	17% (20%)			+0·83
BADS			20.6 (8.9)	19-1 (5-9)	19-9 (7-5)	20-4 (4-4)	-1·50 (-4·9 to 1·9)	-0·146	^^0.38
Data are moment interval.	presented as an item could r GAS=goal atta	mean (SD). N not be scored inment scalii	Vot all patients w l, this item was ex ng. n.a.=not appl	/ere able to perf ‹cluded from ana \licable. DIS= Dys	form all items of Ilysis. ITB=intrath kinesia Impairme	the DIS standardi. ecal baclofen. n=n nt Scale. BADS=B	zed video-protocol. umber. RC=Regressic arry Albright Dyston	When at one n on Coefficient. C iia Scale. ^a Data	neasurement I=confidence not available

for all patients. "Linear regression analysis 'Mann Whitney U test.

4 . . ÷. 4 4 2 4 -4 ł •


Figure 4. A. Differences in percentage of GAS goals achieved; B. Differences in percentage of GAS goals partially achieved for intrathecal baclofen treatment (ITB) and placebo group ITB=intrathecal baclofen. 4A, p=0.005. 4B, p<0.001. The percentage of patients (partially) achieving zero, one, two or three goals is shown for the ITB and placebo group.

Table 5 shows the results for DIS and BADS (sub)scores. At three months, no significant difference between groups was seen for the BADS. The total DIS (RC -6%, ES -0.28, p=0.045), DIS dystonia subscale (RC -8%, ES -0.29, p=0.017) and dystonia during rest subscale (RC -12%, ES -0.32, p=0.013) were significantly more favorable with ITB.

There was no significant correlation between DIS (sub)score(s) and GAS T-scores, and between changes (pre-post) in DIS (sub)score(s) and GAS T-scores.

There were no significant differences between groups for any of the other secondary outcome measures, neither in the domain of body functions and structures (spasticity, pain, comfort) nor in the domain of activities and participation (PEDI). Parents' thoughts on group allocation were correct for most patients in both groups (ITB 76%, placebo 87%).

There were a total of 23 AEs and 6 SAE's in 22 patients. There was no significant difference between groups for total number or type of (S)AEs (table 7). There was also no significant difference between groups for the change in parent reported symptoms related to SRBD.

Table 6. Treatment goals

ICF-CY level Goal		Total ITB group (n=18)			Placebo gro (n=18)	oup	p-value
		Total/ evaluated nr of goals	Total/ evaluated nr of goals	Nr (partially) achieved goals	Total/ evaluated Nr of goals	Nr (partially) achieved goals	
Вос	dy functions and s	tructures					
		16/14	6/6	5 (83%)	10/8	1 (13%)	p=0.01+
	Sleep	5/5	1/1	1 (100%)	4/4	1 (25%)	
	Pain/comfort	7/6	4/4	3 (75%)	3/2	0 (0%)	
	Alertness	2/1	0	n.a.	2/1	0 (0%)	
	Weight	1/1	1/1	1 (100%)	0	n.a.	
	Wearing AFO	1/1	0	n.a.	1/1	0 (0%)	
Act	ivities and partici	pation					
Cor	nmunication	6/6	2/2	0 (0%)	4/4	1 (25%)	n.a.
	Using communication device	6/6	2/2	0 (0%)	4/4	1 (25%)	
Мо	bility	36/32	18/18	12 (67%)	18/14	1 (7%)	p=0.001.
	Changing position	5/5	3/3	2 (67%)	2/2	0 (0%)	
	Sitting	8/5	3/3	3 (100%)	5/2	0 (0%)	
	Moving around	11/11	7/7	5 (71%)	4/4	0 (0%)	
	Hand and arm use	12/11	5/5	2 (40%)	7/6	1 (17%)	
Environmental factor		s					
Car oth	e giving by ers	46/41	26/23	18 (78%)	20/18	0 (0%)	p<0·001⁺
	Dressing	28/25	14/12	9 (75%)	14/13	0 (0%)	
	General care giving	1/0	0	n.a.	1/0	n.a.	
	Washing	4/4	4/4	3 (75%)	0	n.a.	
	Hygienic care	13/12	8/7	6 (86%)	5/5	0 (0%)	
Tot	al	104/93	52/49	35 (71%)	52/44	3 (7%)	

Data are presented as number, or number (%). ICF-CY=International Classification of Functioning, disability and health – Children and Youth version. ITB= intrathecal baclofen. n=number. AFO=ankle foot orthosis. n.a.=not applicable. ⁺comparison of number of partially achieved goals between groups with Mann-Whitney U Test.

			ITB group	Placebo group	P-value
Number of		Total	16	13	*1·00
(S)AE		AE	14	9	⁺ 1·00
		SAE	2	4	⁺ 1·00
Type of AE /	Surgery/pump	Liquor leakage	6/1	5/1	^0·84 /
SAE	implantation	Pump infection	0	0/2	^0·29
	related	Catheter related	0	0/1	
	Possibly adverse drug effects	Nausea or vomiting	2	1	
		Obstipation	0/1	0	
	Other ^a		6	3	

Table 7. Adverse events

Data are presented as number. ITB=intrathecal baclofen. AE=adverse event. SAE=serious adverse event. ⁺Mann-Whitney U test. [^]Pearson Chi Square test. ^aOther complications mainly involved infections (e.g. urinary tract infection, gastrointestinal infection).

DISCUSSION

We report the results of the first multi-centre, randomized, double-blind, placebocontrolled trial, that we are aware of, on continuous ITB treatment in dyskinetic CP. Compared to placebo, ITB shows a superior effect on attainment of individual treatment goals. Furthermore, outcome for dystonia, as measured with the DIS, is slightly more favorable with ITB compared to placebo.

Individual goals are achieved significantly more often with ITB compared to placebo. Our study population involves severely affected dyskinetic CP patients with little to no motor skills (GFMCS IV and V). Treatment goals for these patients are mostly to increase comfort, decrease pain, ease caregiving and facilitate mobility (e.g. transfers and positioning)⁹. Standardized questionnaires on quality of life or activities and participation, that were available at the time of the study, do not adequately capture the individual problems in daily life of these patients, and attainment of individual goals is therefore the most useful and meaningful way to determine treatment effects. When looking at the different categories of set goals, we saw a difference between groups in (partial) achievement of goals on the ICF-CY level of body functions and structures, activities and participation (mobility), and environmental factors (caregiving by others). The findings from our study are in line with previous case control and case series studies and add significantly to the level of evidence for the use of ITB to improve goals in these domains.^{6,7} There were significant differences between ITB and placebo for the DIS total dystonia subscale and the DIS dystonia rest subscale in favor of ITB. However, we were not able to confirm results of previous studies, which found a decrease in BADS scores with ITB.^{12,21,22,24} Previous studies had a lower level of evidence because of the non-randomized non-blinded study designs. We used standardized videos, and assessors were both masked for treatment allocation and timing of measurements (baseline or follow-up). The reason for finding a difference between dystonia measured by DIS and BADS might be that the situation of measurement in the DIS is better defined than the BADS, which hypothetically can lead to better reliability and less variation in scores. However, there are no test-retest reliability studies for either the BADS or the DIS. Considering the fluctuation of dystonia in individual patients, test-retest reliability and sensitivity to change could be problematic for both scales.^{17,21}

A significant difference between groups was found for the dystonia during rest subscale, but not for the dystonia during activity subscale. This may be explained by dystonia being subject to fluctuation and aggravated by non-specific stimuli such as emotion, stress and intentional movement.^{6,22} It is likely that, despite treatment, dystonia is still aggravated during periods of activity, and possibly also stress and emotions. The exact clinical implications of the small differences within and between groups are unclear.

Choreo-athetosis scores were similar between groups. We found mean choreo-athetosis scores to be lower than mean dystonia scores which is accordance with previously published findings.^{22,23}

DIS (sub)score(s) at three months and changes in DIS (sub)score(s) were not correlated with GAS T-scores. A previous study by Monbaliu et al. showed lower functional abilities for children with higher levels of dystonia and in addition poorer scores on participation and quality of life questionnaires.²³ Our hypothesis was that the effect on goal attainment would be caused by a reduction of dystonia, but this was not confirmed in our study. Fluctuation of dystonia and reliability of currently available instruments, as discussed above, might be of influence on this finding.

Both clinical and electrophysiological spasticity measures were not significantly different between groups. In many patients with a predominant dyskinetic movement disorder, spasticity is also present.¹ In patients with spastic CP, the H-reflex is found to be a feasible and objective measure to identify spinal cord neuronal response to ITB.¹⁸ Considering the findings of previous studies, spasticity was expected to decrease with ITB.¹⁸ In our patients however, spasticity was often difficult to assess reliably since both

measures used, are subjective to the degree of relaxation of the patient.²⁴ Relaxation was very difficult for most due to involuntary muscle contractions and movements which characterize dyskinesia, and therefore spasticity measures might be less reliable in this group.

Changes in range of motion (ROM) were similar between groups which corresponds with previous studies.²⁵ This is not surprising because when contractures occur, due to changes in bony structures or to muscle shortening, these problems will not improve on the short term with a decrease of muscle tone established by ITB. On the long term however, contractures might be less progressive or stabilized with ITB, preventing (worsening of) contracture related problems in daily life.

Pain and comfort scores at three months were similar between groups. As most patients in our study did not report pain or comfort as a problem, improvement was not expected for these patients. However, 3 out of 4 patients on ITB who did report pain or discomfort as treatment goals, (partially) achieved this goal compared to non in the placebo group. These last results correspond with previously published studies.²⁵⁻²⁸

Most parents were correct about group allocation which shows that changes due to ITB are indeed favorable, and significantly noticeable. Parents' thoughts on group allocation might bias outcome for goal attainment scaling. We limited this effect by asking parents to describe the current situation without telling them how the baseline situation was.

Baclofen did not have an additional adverse effect on the known serious complications of pump implantation since AE and SAE were similar between groups in types and number. In comparison with previous studies, the frequency of adverse drug effects as constipation and nausea or vomiting were low in our study.²⁹ The majority of severely affected CP patients already experience constipation before pump implantation. Bedrest after surgery can increase this problem further. In our study, constipation worsened in one of the patients on ITB requiring prolonged hospital admission. Nausea and vomiting occurred in three patients (two ITB, one placebo). Anesthesia, surgery, the period of bedrest after surgery in patients prone to reflux might all elicit nausea and vomiting in addition to the possible adverse drug effect of baclofen. Several surgery-related complications were seen, with symptomatic CSF leak and infection being the most frequent. We found a higher occurrence of CSF leak compared to literature and a lower incidence for infection.⁶ We observed one catheter related problem. Comparison with previous studies is not possible since catheter related problems often occur after a longer follow up time.

No difference was found between ITB and placebo in parent reported symptoms related to risk of sleep related breathing disorders. Apnea or hypopnea during sleep is increased with ITB in adult patients with spinal cord injury and multiple sclerosis.¹⁰ Children with CP already have a higher risk of sleep-related disorders than typically developing children.¹⁰ Hypothetically, this risk might increase even more by placing the catheter at the mid cervical level, close to the breathing centre. To be able to fully reject this hypothesis, studies using polysomnography are needed.

Total dosage was similar between groups. This was to be expected as both groups were required to have at least 10 dose increments during the study. More patients in the ITB group were on a bolus schedule. This can be explained by the dosing schedule we used (table 1, figures 1 and 2). If no effect is noticed, the daily dose is increased in the simple continuous mode, providing a fixed infusion rate throughout the day. Only when effect is noticed, but the effect decreases over time, a bolus will be added. Consequently, some of the patients on ITB were on the simple continuous mode, while others were on a "flex mode" in which a bolus can be included.

The study has several limitations. First, the study was powered on the primary outcome measure resulting in a relatively low number of patients needed. This limits the possibility for subgroup analysis on effect modifying factors. It was not possible to place the catheter tip at the aimed C4 level in some patients due to technical surgical issues. Considering the hypothesis that ITB in dyskinetic patients works intracranially, the effect in these patients might be less than when placed on a higher level. Another limitation is the short follow up period of three months. In our clinical experience it takes at several weeks up to several months to find the right dosage for the individual patient. We aimed to approach the optimum dosage by requiring at least 10 increments. Still, some patients might not be on the most adequate dosage yet and others only for a short period of time. However, we did find a clear difference for our primary outcome measure. Furthermore, a longer period of placebo was felt to be unethical. We will perform an additional 9 month open label follow-up period for all patients. Last, assessors were blinded for treatment allocation, but we did not test whether this was successful (i.e. whether they personally had an idea about treatment allocation which could have influenced their judgement). Retrospectively we asked assessors to recall whether they had thoughts on group allocation. They responded that they had no opinion on group allocation during measurements, including scoring of the GAS. During scoring of the videos, which were provided coded and in random order (mixing patients and measuring moments), they also had no idea on group allocation.

Future studies should assess long-term effects and complications using a prospective longitudinal cohort study design. An (inter)national register, like the Australian ITB audit, could provide a good basis for such a study, ensuring sufficient patient numbers and harmonization of outcome measures.³⁰ The achievement of individual treatment goals should have the primary focus. With a larger sample size, patient- and treatment characteristics and factors on the level of body functions and structures, influencing goal attainment, can hopefully be identified. Furthermore, it might provide us with additional insights in optimal dosage and provide evidence for the usefulness of our dosing schedule designed for patients with dyskinetic CP, which is now based on clinical observation.

Considering the described problems in reliable measurement of dystonia, test-retest reliability studies of the DIS and BADS are needed, and furthermore, it would be useful to examine whether a shortened version of the DIS, perhaps limiting to the most responsive and clinically relevant items, makes the DIS more feasible for severely affected patients with dyskinetic CP in both research and clinical practice. In addition, other measurement methods of dystonia should be explored, such as instrumented measures, which are not dependent on assessors, and which can easily be applied at home or at day-care/school to decrease provocation of stress and ensure correspondence with the daily situation.

In conclusion, we were able to provide level II evidence for the effect of intrathecal baclofen in paediatric and adolescent patients with severe dyskinetic CP (GMFCS IV and V) on the achievement of individual treatment goals. ITB should be considered as a treatment option in patients with severe dyskinetic CP in whom oral medication is insufficient. Treatment goals should be on the level of body functions and structures (such as pain or discomfort), on the level of activities and participation for mobility (such as transfers and sitting), and/or goals on the level of environmental factors (caregiving by others). Studies into the long term effects of ITB and factors influencing outcome, and on reliable measurement of dystonia in severe dyskinetic CP are future perspectives.

ACKNOWLEDGEMENTS

This study was funded by the Phelps Stichting voor Spastici (project number 2011037 and project number 99.047), het Revalidatiefonds (project number R2011032/), Kinderrevalidatie Fonds de Adriaanstichting (project number 11.02.17-2011/0035) and het Johanna KinderFonds (project number 2011/0035-357). We thank Sanne van der Vossen, MD for her help in conducting measurements during the absence of LB, Tamara van Ams and Sharony de Vries for taking care of the logistics around study medication,



appointments and pump fillings, and Helga Haberfehlner for her enthusiastic and innovative ideas.

CONFLICTS OF INTEREST

There are no potential competing interests to report.

AUTHOR CONTRIBUTIONS

The members of the IDYS study group are, from the Amsterdam UMC, Vrije Universiteit, Amsterdam, the Netherlands: Laura A. Bonouvrié MD, Prof Jules G. Becher MD, Annemieke I. Buizer MD, Karin Boeschoten, Johanna J.M. Geytenbeek PhD, Prof Vincent de Groot MD (Rehabilitation Medicine), Laura A. van de Pol MD (Child Neurology), Willem J.R. van Ouwerkerk MD, K M Slot MD, Prof S M Peerdeman MD (Neurosurgery), Rob L.M. Strijers MD (Clinical Neurophysiology), Elisabeth M.J. Foncke MD (Neurology), Jos W.R. Twisk PhD, Peter van de Ven PhD (Epidemiology and Biostatistics), and from Maastricht University Medical Center, Maastricht, the Netherlands: Prof R Jeroen Vermeulen MD, Prof Johan S.H. Vles MD, Dan Soudant MANP, Sabine Fleuren (Neurology) and Onno P. Teernstra MD (Neurosurgery).

LB, JB, JV, RJV, KB, JG, VG, WO, SP, RS, EF, and PV contributed to the conception and design of the study; LB, JB, RJV, AB, KB, DS, SF, JG, RS, LP, and JT contributed to the acquisition and analysis of data; LB, JB, JV, JT, RJV, and AB contributed to drafting the text and preparing the figures.

REFERENCES

- Rosenbaum P, Paneth N, Leviton A, Goldstein M, Bax M, Damiano D, et al. A report: the definition and classification of cerebral palsy April 2006. Dev Med Child Neurol. 2007;109 (Suppl):8-14.
- Oskoui M, Coutinho F, Dykeman J, Jetté N, Pringsheim T. An update on the prevalence of cerebral palsy: a systematic review and meta-analysis. Developmental Medicine & Child Neurology. 2013;55:509–19.
- Murthy GV, Mactaggart I, Mohammad M, Islam J, Noe C, Khan AI, et al. Assessing the prevalence of sensory and motor impairments in childhood in Bangladesh using key informants. Arch Dis Child. 2014;99(12):1103-8.
- Kakooza-Mwesige A, Andrews C, Peterson S, Wabwire Mangen F, Eliasson AC, Forssberg H. Prevalence of cerebral palsy in Uganda: a population-based study. Lancet Glob Health. 2017;5:e1275-82.
- Krägeloh-Mann I, Petruch U, Weber PM. SCPE Reference and Training Manual (R&TM). Grenoble: Surveillance of Cerebral Palsy in Europe; 2005.
- Monbaliu E, Himmelman K, Lin JP, Ortibus E, Bonouvrié L, Feys H, et al. Clinical presentation and management of dyskinetic cerebral palsy. Lancet Neurol. 2017;16(9):741-9.
- Fehlings D, Brown L, Harvey A, Himmelmann K, Lin JP, Macintosch A, et al. Pharmacological and neurosurgical interventions for managing dystonia in cerebral palsy: a systematic review. Dev Med Child Neurol. 2018;epub before print.
- Buizer AI, Martens BHM, Grandbois van Ravenhorst C, Schoonmade LJ, Becher JG, Vermeulen RJ. Effect of continuous intrathecal baclofen therapy in children: a systematic review. Dev Med Child Neurol. 2018;epub ahead of print.

- Liew PY, Stewart K, Khan D, Arnup SJ, Scheinberg A. Intrathecal baclofen therapy in children: an analysis of individualized goals. Dev Med Child Neurol. 2018;60(4):367-73.
- Bonouvrié LA, Becher JG, Vles JSH, Boeschoten K, Soudant D, de Groot V, et al. Intrathecal baclofen treatment in dystonic cerebral palsy: a randomized clinical trial: the IDYS trial. BMC Pediatrics. 2013;131:175-83.
- Albright AL, Barry MJ, Shagron DH, Ferson SS. Intrathecal baclofen for generalized dystonia. Dev Med Child Neurol. 2001;43:652-7.
- Albright AL, Turner M, Pattisapu JV. Best-practice surgical techniques for intrathecal baclofen therapy. J Neurosurg. 2006;104:233-9.
- Konstantelias AA, Vardakas KZ, Polyzos KA, Tansarli GS, Falagas ME. Antimicrobialimpregnated and -coated shunt catheters for prevention of infections in patients with hydrocephalus: a systematic review and meta-analysis. J Neurosurg. 2015;122(5):1096-112.
- 14. Geytenbeek JJ, Mokkink LB, Knol DL, Vermeulen RJ, Oostrom KJ. Reliability and validity of the C-BiLLT: A new instrument to assess comprehension of spoken language in your children with cerebral palsy and complex communication needs. Augmentative and Alternative Communication. 2014;30(3):252-66.
- Organization WH. International classification of functioning, disability and health: children & youth version: ICF-CY. 2007.
- Monbaliu E, Ortibus E, Prinzie P, Dan B, De Cat J, De Cock P, et al. Can the Dyskinesia Impairment Scale be used by inexperienced rater? A reliability study. European Journal of Neurology. 2012;17(3):238-47.

- 17. Stewart K, Harvey A, Johnston LM. A systematic review of scales to measure dystonia and choreoathetosis in children with dyskinetic cerebral palsy. Dev Med Child Neurol. 2017;59(8):786-95.
- Hoving MA, van Kranen-Mastenbroek VHJM, van Raak EPM, Spincemaille GHJJ, Hardy ELM, Vles JSH. Placebo controlled utility and feasibility study of the H-reflex en flexor reflex in spastic children treated with intrathecal baclofen. Clinical Neurophysiology. 2006;117:1508-17.
- 19. Feldman AB, Haley SM, Coryell J. Concurrent and construct validity of the Pediatric Evaluation of Disability Inventory. Phys Ther. 1990;70(602-610).
- Chervin RD, Weatherly RA, Garetz SL, Ruzicka DL, Giordani BJ, Hodges EK, et al. Pediatric Sleep Questionnaire. Archives of Otolaryngology - Head & Neck Surgery. 2000;133:216-22.
- Gimeno H, Tustin K, Lumsden D, Ashkan K, Selway R, Lin JP. Evaluation of functional goal outcomes using the Canadian Occupational Performance Measure (COPM) following Deep Brain Stimulation (DBS) in childhood dystonia. European Journal of Paediatric Neurology. 2014;18(3):308-16.
- 22. Monbaliu E, de Cock P, Ortibus E, Heyrman L, Klingels K, Feys H. Clinical patterns of dystonia and choreoathetosis in participants with dyskinetic cerebral palsy. Dev Med Child Neurol. 2016;58(2):138-44.
- Monbaliu E, De Cock P, Mailleux L, Dan B, Feys H. The relationship of dystonia and choreoathetosis with activity, participation and quality of life in children and youth with dyskinetic cerebral palsy. European Journal of Paediatric Neurology. 2017;21(2):327-35.

- Chen YS, Zhou S, Cartwright C, Crowley Z, Baglin R, Wang F. Test-retest reliability of the soleus H-reflex is affected by joint positions and muscle force levels. J Electromyogr Kinesiol. 2010;20(5):980-7.
- Eek MN, Olsson K, Lindh K, Askljung B, Pahlman M, Corneliusson O, et al. Intrathecal baclofen in dyskinetic cerebral palsy: effects on function and activity. Dev Med Child Neurol. 2018;60(1):94-9.
- 26. Bonouvrie L, Becher J, Soudant D, Buizer A, van Ouwerkerk W, Vles G, et al. The effect of intrathecal baclofen treatment on activities of daily life in children and young adults with cerebral palsy and progressive neurological disorders. European Journal of Paediatric Neurology. 2016;20:538-44.
- 27. Bonouvrié LA, van Schie PEM, Becher JG, van Ouwerkerk WJR, Reeuwijk A, Vermeulen RJ. Effects of intratheal baclofen on daily care in children with secondary generalized dystonia: a pilot study. European Journal of Paediatric Neurology. 2011;15:539-43.
- Motta F, Stignani C, Antonello CE. Effects of intrathecal baclofen on dystonia in children with cerebral palsy and the use of functional scales. Journal of Pediatric Orthopaedics. 2008;28:213-7.
- Ward A, Hayden S, Dexter M, Scheinberg A. Continuous intrathecal baclofen for children with spasticity and/or dystonia: Goal attainment and complications associated with treatment. Journal of Paediatrics and Child Health. 2009;45:720-6.
- Stewart K, Hutana G, Kentish M. Intrathecal baclofen therapy in paediatrics: a study protocol for an Australian multicentre, 10-year prospective audit. BMJ Open. 2017;7(6):e015863.

IDYS trial: the results of a randomized clinical trial

CHAPTER 6

Intrathecal baclofen for progressive neurological disease in childhood: a systematic review of literature

LA Bonouvrié, PEM van Schie, JG Becher, WJR van Ouwerkerk, RJ Vermeulen

Eur J Paediatr Neurol 2012; 16(3): 279-284



ABSTRACT

Background Intrathecal baclofen (ITB) treatment is frequently used for individuals with severe, but non-progressive, spasticity refractory to oral treatment. However, experiences with ITB in patients with progressive neurological disorders of childhood causing spasticity are limited.

Aim To investigate whether ITB is an option in patients with progressive neurological disorders causing spasticity in childhood.

Design A systematic literature search in Embase, Pubmed and the Cochrane Library was performed.

Results We identified six eligible studies considering patients with progressive neurological disease in childhood and receiving ITB treatment. The studies included a total of seven paediatric patients and four adult patients. Improvement was reported in spasticity, spasms, pain, gait, activities of daily life and providing care. High satisfaction is described.

Conclusions ITB has beneficial effects in paediatric patients with progressive neurological disease. However, the level of evidence is limited due to the small number of available studies and due to the poor quality of these studies

Keywords Intrathecal baclofen, Progressive neurological disease, Spasticity

INTRODUCTION

Intrathecal baclofen treatment (ITB) has been used since the 1980s for the treatment of spasticity.¹ Baclofen is a GABAagonist^{2,3} and inhibits neuronal transmission³⁻⁵ at the level of the spinal cord.

ITB is an effective treatment for individuals with severe spasticity refractory to oral, or other non-invasive, spasticity reducing treatments.^{3,6-8} ITB is effective especially in patients with stable underlying conditions such as cerebral palsy (CP),^{3,4,6,8} traumatic brain injury^{4,6} and spinal cord injury.^{1,3,5} In these patients ITB decreases spasticity and spasticity related pain, improves functional activities related to activities of daily life^{3,9-11} and facilitates ease of care.⁸⁻¹⁰ Although complications occur frequently,^{6,8,9} most individuals and their caregivers are satisfied with ITB treatment.¹¹

The experiences with ITB in patients with progressive neurological disorders of childhood causing spasticity are limited. We will therefore present a systematic review of literature to investigate if ITB is a treatment option in patients with progressive neurological disorders causing spasticity in childhood.

METHODS

Search strategy

A systematic literature search in Pubmed, Embase and the Cochrane Library, was conducted on April 15th 2010. The search strategy used for Pubmed is shown in Table 1. For Embase and the Cochrane Library comparable search terms are used (available as additional material). Titles and available abstracts identified in the searches were reviewed to identify relevant studies. References of relevant studies were searched to identify studies not found in the electronic searches.

Study selection

We included all studies in which: 1) spasticity was caused by progressive neurological disorders with a presentation in childhood (<18 years), 2) the intervention was ITB administration by either a bolus injection by lumbar punction (LP) or continuous infusion by an implanted pump.

We excluded the following studies in which: 1) non-progressive diseases causing spasticity were described; 2) mixed progressive and non-progressive disease in which progressive neurological diseases causing spasticity were not

further specified or not separately analysed.

Table	1	Search	strategy	PubMed
Iavie	- 1	Jearch	Suaregy	rubivieu

Search number	Search terms	Results
#1	"baclofen"[Mesh]	4232
#2	"baclofen"[tiab] OR "pcp gaba"[tiab] OR "beta p chlorophenyl gamma aminobutyric acid"[tiab] OR "chlorophenyl gaba"[tiab] OR "beta aminomethyl 4 chlorobenzenepropanoic acid"[tiab] OR clofen[tiab] OR genbaclofen[tiab] OR atrofen[tiab] OR lioresal[tiab] OR "nubaclo"[tiab] OR nubaclo[tiab] OR apobaclofen[tiab] OR "ba 34 647"[tiab] OR "ba34 647"[tiab] OR "ciba 34 647 ba"[tiab] OR "ciba34 647ba"[tiab OR "ba 34647"[tiab] OR ba34647[tiab] OR baclospas[tiab]	4987
#3	#1 OR #2	5723
#4	"Injections, Spinal"[Mesh]	11400
#5	(Intrathecal[tiab] OR intraspinal[tiab] OR spinal[tiab]) AND (injection*[tiab] OR infusion*[tiab])	15525
#6	#4 OR #5	23652
#7	#3 AND #6	751
#8	Child*[tw] OR schoolchild*[tw] OR infan*[tw] OR adolescen*[tw] OR pediatri*[tw] OR paediatr*[tw] OR neonat*[tw] OR boy[tw] OR boys[tw] OR boyhood[tw] OR girl[tw] OR girls[tw] OR girlhood[tw] OR youth[tw] OR youths [tw] OR baby[tw] OR babies[tw] OR toddler*[tw] OR "Mental Disorders Diagnosed in Childhood"[MeSH] OR teen [tw] OR teens[tw] OR teenager*[tw] OR newborn*[tw] OR postneonatal*[tw] OR postnat*[tw] OR OR puberty[tw] OR preschool*[tw] OR suckling*[tw] OR picu[tw] OR nicu[tw] OR	2884776
#9	#7 AND #8	239

Assessment

Two researchers (RJV, LAB) independently reviewed the selected studies and produced summary data tables (level of evidence,¹² study-type, patient-characteristics, intervention, outcome measurements and results). Findings were discussed and consensus was achieved for final analysis of the data. Outcome data extracted from the selected studies included spasticity scores (Ashworth scale, Penn scores), range of motion (ROM), spasticity associated pain (no specified method), gait (velocity, stride length, step width), functioning (no specified method), satisfaction and ease of providing care (no specified method).

RESULTS

Description of studies

Six eligible studies were found in the electronic databases.¹³⁻¹⁸ One study was level 4 evidence (case-series),¹⁷ five studies were level 5 evidence (expert opinion/case report)^{13-16,18} as defined by the Oxford Centre of Evidencebased Medicine levels of evidence.¹² Study characteristics and outcomes are presented in Table 2.

Description of results

Spasticity, spasms and pain

Three studies described spasticity (Ashworth score) (all level 5 evidence).^{14,15,18} ITB treatment decreased spasticity in all studies.^{14,15,18} ITB also decreased disabling extensor spasms in one patient, This patient noticed subsequent improvement of pain and sleep.¹³ Two other patients reported subjective pain relief.^{14,18}

Mobility

In one patient improvement of gait velocity, step length and a decrease in step width was reported (level 5 evidence).¹⁶ Subjective improvement in ease of assisted transfer was reported in one paediatric patient (level 5 evidence).¹⁴

Activities of daily living and providing care

Four studies (level 4 and 5) described patients who were reliable on others for personal care.^{13,14,17,18} In three studies, caregivers report improvement in dressing, hygienic care and positioning in a wheelchair.^{14,17,18} One adult patient experienced facilitation of positioning in his wheelchair.¹³

Furthermore, after starting ITB two patients were reported to be more alert and more interactive during the day.^{14,18} Caregivers of one patients reported a subsequent improved quality of life.¹⁸

Side effects and complications

Complications were not reported in the level 5 studies (n = 5). In one study, two of six patients experienced cerebrospinal fluid (CSF) leak during the test phase of treatment. One patient experienced catheter related problems requiring surgery.¹⁷

ŝ
ŭ
·
÷
S.
<u> </u>
Ð
÷
C
B
-
10
_
()
_
~
کم م
A of
ndy o
tudy o
Study o
. Study o
2. Study o
2. Study o
e 2. Study o
le 2. Study o
ole 2. Study o
able 2. Study o
able 2. Study o
Table 2. Study o

		ent) t)		elf-	ts	
tcome measures and results	Decreased extensor spasms (subjectively and electrophysiologically measured) Decreased associated pain (subjectively) Improved sleep (subjectively) Facilitated positioning in wheelchair (subjectively) High satisfaction (subjectively)	Decreased spasticity (2 points ASH) Increased ROM UE and LE Elimination of pain (parental assessment) Improved hygienic care and dressing (parental assessmen Improved transfer and positioning (parental assessmen	Decreased spasticity (2 points ASH)	Improvement in gait (Increased step length, increased s selected speed in stride, decreased step width)	Four caregivers were satisfied with ITB treatment, one was partially satisfied, one was not satisfied (caregiver questionnaire by VAS) Dissatisfaction due to unfulfilled goals or negative effec (caregiver questionnaire)	Increased range of motion LE Decreased spasticity UE and LE (1point ASH) Improvement in ease of care including hygienic care and positioning (caregivers assessment) Decreased pain and increased quality of life (caregivers assessment)
Ou	• • • • •	• • • • •	•	•	• •	••••
Duration FU	6 months	10 months	n.a.	24 months	6 months to 6.7 year	6 months
Intervention	Bolus ITB ^a (50, 75, 100 μg) Continuous ITB ^b (100 μg/24u)	Continuous ITB ^b	Bolus ITB ^b (75 μg)	Continuous ITB ^b	Continuous ITB ^b	Continuous ITB ^b
Ambulation	Non- ambulatory	Non- ambulatory	Non- ambulatory	Ambulatory	Ambulatory and non- ambulatory	Non- ambulatory
Patient characteristics (Diagnosis, age)	Friedreich's ataxia 39-yr-old male	X-linked adrenoleukodystrophy 8-yr-old male	FSP 41-yr-old male	HSP 31-yr-old male	Multiple progressive neurological diseases 4 to 18-yr old 50% male	Rett syndrome 32-yr-old woman
Number of patients	-	1	Ч	1	ω	
itudy ype	ж,	ж,	ĸ	ĸ	S	<u>ج</u>
DE S		0	0	0	0	
<i>if</i> 10	ц	Ы	ы	ы	4	ں
Re	13	14	15	16	17	18

cohort and case-control studies). Level 5 evidence: expert opinion. LE, lower extremity. OL, open label. Ref, reference number. ROM, range of motion.

UE, upper extremity. VAS, visual analog scale. Yr, year.

Baclofen related side effects were also reported in this study. In two patients, caregivers reported progressive scoliosis. In two other patients caregivers reported ITB-tolerance. Other reported side effects (all reported once) included constipation, hypotension, somnolence, increased drooling, decreased head control, swallowing difficulties, abdominal discomfort, back pain, withdrawal symptoms due to a nearly empty pump and iatrogenic overdose.¹⁷

Satisfaction

One level 4 evidence study scored satisfaction in six paediatric patients. Satisfaction was measured with a modified visual analog scale (VAS) with 0 being the poorest score and 10 the most optimal score. Four of six caregivers were satisfied with

the treatment. In one patient the caregiver was only partially satisfied due to improvements but also side effects and complications. In one patient the caregiver scored satisfaction as a score of 7.0, however he was not satisfied with the ITB treatment due to unfulfilled goals.¹⁷

DISCUSSION

Data on ITB in patients with spasticity due to progressive neurological disorders with clinical presentation during childhood are limited. In our systematic review of literature, we found six studies reporting seven paediatric patients and four adult patients with manifestation of their disease during childhood. The level of evidence was low in all studies (one level 4 case-series,¹⁷ five level 5 expert opinion/case report).^{13-16,18}

ITB treatment decreases spasticity caused by non-progressive diseases such as cerebral palsy.^{1-6,8,19} This review shows ITB also decreases spasticity and spasms due to progressive diseases.^{13-15,18}

In one ambulant adult patient, improvement of gait was reported during continuous ITB treatment.¹⁶ Other studies showed similar effects on gait in adult patients with progressive neurological disease.²⁰⁻²² Improvement in gait may lead to functional improvement as one study showed a decreased need for walking aids.²²

Most patients in this review were non-ambulatory. Also, in spastic CP patients, the positive effect of ITB is most frequently described in non-ambulatory patients (GMFCS IV or V). For ambulatory CP patients (GMFCS levels I-III) only limited data on improvement of mobility are available.⁸

In this study most patients were dependent on caregivers for mobility and personal care (seven paediatric patients and one adult patient). Three studies reported improvement of ease of care, including dressing, hygienic care, positioning and transfer.^{14,17,18} These findings are in line with the goals and effects of ITB treatment on function and caregiver assistance in GMFCS IV and V CP patients.^{8,11}

Side effects and complications of ITB treatment are common.^{2,4,6,8,23} The most frequent complications are infections, cerebrospinal fluid (CSF) leaks and catheter problems.^{2-4,8} In this review, only one study mentioned a total of three complications in six patients¹⁷ comparable to the most frequent reported side effects in CP patients receiving ITB treatment.

Constipations is the most common chronic side effect of ITB in CP patients.²⁻⁴ Though constipation was reported only once in the progressive disease patient population. In two patients progressive scoliosis was reported during ITB treatment.¹⁷ Since ITB treatment causes muscle weakness, including weakness of the paraspinal muscles, it is hypothesized that ITB might elicit scoliosis.²⁴ In addition to others, we think that scoliosis is a common complication in severe neurological disease, also without ITB treatment.^{25,26} Therefore, as well in patients with non-progressive disease as in patients with progressive neurological disease and spasticity, the development and progression of scoliosis should be monitored closely.

Tolerance to ITB is an important complication and results in increased muscle tone over time.⁶ We found two patients who suffered from tolerance.¹⁷ We must realize that tolerance is not common in patients with CP.^{2,4} Presumably, unresponsiveness to ITB in these patients is usually caused by an underlying system malfunction. Physiological adaptation is common in the first years of treatment, Albright et al. showed increasing baclofen dosages during the first one to three years of ITB treatment in children with cerebral palsy.⁶ In our review population, patients had spasticity due to progressive neurological disease. It is not inconceivable that in these patients spasticity worsens as disease progresses, requiring increasing dosages of ITB.

Side-effects and complications can lead to dissatisfaction with ITB treatment. Other reasons for being unsatisfied or being only partially satisfied with continuous ITB treatment are unfulfilled goals. Fortunately, most patients with spasticity due to progressive neurological disease and their caregivers are satisfied with continuous ITB treatment as are patients with stable underlying disease.^{11,13,17,18,27}

CONCLUSION

In patients with progressive neurological disease causing spasticity, ITB improves not only spasticity, spasms and pain but furthermore has positive effects on mobility, facilitation of care including dressing, hygienic care, positioning and transfer. Despite the widespread use of ITB, the few available studies and the poor quality of these studies (level 4 and mostly 5) is low, limits our conclusion.

We think that ITB should be considered as a treatment option in patients with progressive neurological disease when other treatment options fail. Further research is needed to establish the beneficial effect of ITB in patients with progressive neurological disease using a range of outcome measures addressing multiple domains of the International Classification of Functioning, Disability and Health (ICF) and furthermore quality of life and long term outcomes.

ACKNOWLEDGEMENTS

We are grateful to J.C.F. (Hans) Ket, medical information specialist of the VU University Medical Center Library for his assistance with the literature search.

REFERENCES

- Penn R, Kroin J. Intrathecal baclofen alleviates spinal cord spasticity. Lancet 1984;1:1078.
- Albright AL, Ferson SS. Intrathecal baclofen therapy in children. Neurosurg Focus 2006;21. e3.
- Guillaume D, Van Havenbergh A, Vloeberghs M, Vidal J, Roeste G. A clinical study of intrathecal baclofen using a programmable pump for intractable spasticity. Arch Phys Med Rehabil 2005;86:2165e71.
- Albright AL. Intrathecal baclofen for childhood hypertonia. Childs Nerv Syst 2007;23:971e9.
- Latash M, Penn R. Changes in voluntary motor control induced by intrathecal baclofen in patients with spasticity of different etiology. Physiother Res Int 1996;1:229e46.
- Albright AL, Gilmartin R, Swift D, Krach LE, Ivanhoe CB, McLaughlin JF. Long-term intrathecal baclofen therapy for severe spasticity of cerebral origin. J Neurosurg 2003;98:291e5.
- Sampson F, Hayward A, Evans G, Morton R, Collet B. Functional benefits and cost/ benefit analysis of continuous intrathecal baclofen infusion for the management of severe spasticity. J Neurosurg 2002;96:1052e7.
- Dan B, Motta F, Vles JSH, Vloeberghs M, Becher JG, Eunson P, et al. Consensus on the appropriate use of intrathecal baclofen (ITB therapy in paediatric spasticity. Eur J Paediatr Neurol 2010; 14:19e28.
- Hoving MA, van Raak EPM, Spincemaille GHJJ, van Kranen-Mastenbroek VHJM, van Kleef M, Gorter JW, Vles JSH. Safety and one-year efficacy of intrathecal baclofen therapy in children with intracable spastic cerebral palsy. Eur J Paediatr Neurol 2009;13:247e56.

- Hoving MA, van Raak EPM, Spincemaille GHJJ, Palmans LJ, Becher JG, Vles JSH. Efficacy of intrathecal baclofen therapy in children with intracable spastic cerebral palsy: a randomized controlled trial. Eur J Paediatr Neurol 2009;13:240e6.
- Krach LE, Nettleton A, Klempka B. Satisfaction of individuals treated long-term with continuous infusion of intrathecal baclofen by implanted Programmable pump. Pediatr Rehabil 2006;9:210e8.
- Philips B, Ball C, Sackett D, Badenoch D, Straus S, Haynes B, et al. Oxford centre of evidence-based medicine levels of evidence. Oxford Centre of Evidencebased Medicine; 2001. 10.08.07. Ref Type: Electronic Citation.
- Ben Smail D, Jacq C, Denys P, Bussel B. Intrathecal baclofen in the treatment of painful, disabling spasms in Freidreichs ataxia. Mov Disord 2005;20:758e9.
- Chu MLY, Sala DA, Weiner HL. Intrathecal baclofen in Xlinked adrenoleukodystrophy. Pediatr Neurol 2001;24:156e8.
- Dan B, Cheron G. Intrathecal baclofen normalizes motor strategy for squatiing in familial spastic paraplegia: a case study. Neurophysiologie Clinique 2000;30:43e8.
- Molteni F, Carda S, Cazzaniga M, Magoni L, Rossini M, Caimmi M. Instrumental evaluation of gait modifications before and during intrathecal baclofen therapy: a 2-year follow-up case study. Am J Phys Med Rehabil 2005;84:303e6.
- Bonouvrié LA, van Schie PE, Becher JG, van Ouwerkerk WJ, Vermeulen RJ. satisfaction with intrathecal baclofen treatment in paediatric patients with progressive neurological disease. Dev Med Child Neurol 2008;50:636e8.

- Kadyan V, Clairmont AC, George RJ, Johnson EW. Intrathecal baclofen for spasticity management in Rett syndrome. Am J Phys Med Rehabil 2003;82:560e2.
- 19. Hoving MA, Raak van EPM, Spincemaille GHJJ, Palmans LJ, Sleypen FAM, Vles JSH. Intrathecal baclofen in children with spastic cerebral palsy: a double-blind, randomized, placebocontrolled, dosefinding study. Dev Med Child Neurol 2007;49:654e9.
- Dan B, Bouillot E, Bengoetxea A, Cheron G. Effect of intrathecal baclofen on gait control in human hereditary spastic paraparesis. Neurosci Lett 2000;280:175e8.
- Klebe S, Stolze H, Kopper F, Lorenz D, Wenzelburger R, Deuschl G, et al. Objective assessment of gait after intrathecal baclofen in hereditary spastic paraplegia. J Neurol 2004;252:991e3.
- 22. Meythaler JM, Steers WD, Tuel SM, Cross LL, Sesco DC, Haworth CS. Intrathecal baclofen in hereditary spastic paraparesis. Arch Phys Med Rehabil 1992;73:794e7.
- Albright AL, Turner M, Pattisapu JV. Best-practice surgical techniques for intrathecal baclofen therapy. J Neurosurg (Suppl Pediatrics) 2006;104:233e9.
- Sansone JM, Mann D, Noonan K, Mcleish D, Ward M, Iskandar BJ. Rapid progression of scoliosis following insertion of intrathecal baclofen pump. J Pediatr Orthop 2006;26:125e8.
- 25. Shilt JS, Lai LP, Cabrera MN, Frino J, Smith BP. The impact of intrathecal baclofen on the natural history of scoliosis in cerebral palsy. J Pediatr Orthopaedics 2008;28:684e7.
- Senaran H, Shah SA, Presedo A, Dabney KW, Glutting JW, Miller F. The risk of progression of scoliosis in cerebral palsy patients after intrathecal baclofen therapy. Spine 2007;32:2348e54.

 Lambrecq V, Muller F, Joseph PA, Cuny E, Mazaux JM, Barat M. Intétêts et limites du baclofène dans les paraparésies spastiques héréditaires. Ann Readapt Med Phys 2007;50:577e81

CHAPTER 7

Satisfaction with intrathecal baclofen treatment in paediatric patients with progressive neurological disease

LA Bonouvrié, PEM van Schie, JG Becher, WJR van Ouwerkerk, RJ Vermeulen

Dev Med Child Neurol 2008; 50(8): 636-638

Intrathecal baclofen (ITB) treatment could be a beneficial treatment option in paediatric patients with progressive neurological disorders causing spasticity. Since its introduction in the 1980s,¹ ITB has been mainly recommended for individuals with severe spasticity refractory to oral treatment²⁻⁴ and is proven to be effective for reduction of spasticity, especially in patients with stable underlying conditions such as cerebral palsy,^{2,3,5-7} traumatic brain injury,^{2,5,6} and spinal cord injury.^{1,3,6} We noticed that the experiences with ITB in paediatric patients with progressive neurological disorders are limited. In addition, although minor complications occur frequently,^{2,7} most individuals and their caregivers are satisfied with ITB treatment.^{7,8}

We retrospectively studied the overall satisfaction of caregivers with ITB treatment in a group of children and adolescents from our center with progressive neurological disorders causing spasticity. For this purpose, we selected six patients from a total cohort of nine paediatric patients with ITB treatment (Medtronic Inc, Minneapolis, MN, USA; three males, three females; mean age 11y 8mo, [SD 4y 5mo]), with progressive neurological disorders (Table I) starting in childhood. The remaining three patients from the total cohort were not using ITB at the time of the survey. We analyzed whether the treatment effects met the expectations of the caregivers and how they would score the level of overall satisfaction on a scale from 0 to 10 (0 being the poorest score and 10 being the most optimal score). For this purpose we used standard questionnaires.⁹ Medical records were checked for complications related to surgery and ITB treatment and for treatment goals.

The mean time since pump implantation was 3 years 4 months (SD 2y 11mo, range 6mo-6y 8mo) and the mean age at time of pump insertion was 8 years 4 months (SD 3y, range 4–13y).

Reduction of spasticity or spasms, facilitation of care, and reduction of pain were the main goals for initiating ITB treatment. Medical records of the six patients still using ITB reported several complications related to ITB hardware, and surgery, including catheter migration in two patients (Table I). One of the patients experienced an overdose of ITB due to a mistake in programming the pump while having his dosage adjusted. The same patient experienced increase of spasticity due to a nearly empty ITB 10ml pump.

The following side effects of ITB treatment were reported in the questionnaires: increased drooling, increased swallowing difficulties, decreased head balance, abdominal discomfort, and constipation (Table I). Each of these complaints was reported once.

Back pain due to worsening scoliosis was reported twice. In two patients, the caregivers reported increased tolerance of ITB with frequent pump adjustments consequently.

Overall satisfaction scored 7.5 (SD 1.6, range 6–10). For detailed results, see Table I. The caregivers of four patients were satisfied with ITB (nr 1–4), mean satisfaction score 8.0 (SD 1.8, range 6–10). The caregiver of patient 5, stated that they were not satisfied although a score of 7.0 was reported. This dissatisfaction was due to not being able to see full results of ITB yet, because of a short follow-up since pump implantation. The caregiver of patient 6 was partially satisfied due to improvement in dressing, personal hygiene spasms, and pain; but on the other hand noted worsening of swallowing, constipation, head balance, trunk stability, back pain, and worsening scoliosis (satisfaction score 6.0).

We are aware of the fact that this is a retrospective study and we only present a small number of patients. Still, we think this study shows clearly and similar to other studies^{3,8,10,11} that most caregivers are overall satisfied with the effects of ITB treatment. Dissatisfaction was related to unfulfilled goals or related to negative effects of ITB on functioning.

Considering the present results with moderate to good satisfaction, ITB is, in our opinion, a valuable treatment option in patients with progressive neurological diseases.

REFERENCES

- Penn R, Kroin J. Intrathecal baclofen alleviates spinal cord spasticity. Lancet 1984; 1: 1078.
- Albright AL, Gilmartin R, Swift D, Krach LE, Ivanhoe CB, McLaughlin JF. Long-term intrathecal baclofen therapy for severe spasticity of cerebral origin. J Neurosurg 2003; 98: 291–5.
- Guillaume D, Van Havenbergh A, Vloeberghs M, Vidal J, Roeste G. A clinical study of intrathecal baclofen using a programmable pump for intractable spasticity. Arch Phys Med Rehabil 2005; 86: 2165–71.
- Sampson F, Hayward A, Evans G, Morton R, Collet B. Functional benefits and cost/ benefit analysis of continuous intrathecal baclofen infusion for the management of severe spasticity. J Neurosurg 2002; 96: 1052–57.
- Albright AL. Intrathecal baclofen for childhood hypertonia. Childs Nerv Syst 2007; 23: 971–79.
- Latash M, Penn R. Changes in voluntary motor control induced by intrathecal baclfen in patients with spasticity of different etiology. Physiother Res Int 1996; 1: 229–46.
- Hoving MA, Raak van EPM, Spincemaille GHJJ, Palmans LJ, Sleypen FAM, Vles JSH. Intrathecal baclofen in children with spastic cerebral palsy: a double-blind, randomized, placebocontrolled, dosefinding study. Dev Med Child Neurol 2007;49: 654–59.
- Krach LE, Nettleton A, Klempka B. Satisfaction of individuals treated long-term with continuous infusion of intrathecal baclofen by implanted programmable pump. Pediatr Rehabil 2006; 9: 210–18.

- Schneider JW, Gurucharri LM, Gutierrez AL, Gaebler-Spira DJ. Health related quality of life and functional outcome measures for children with cerebral palsy. Dev Med Child Neurol 2001;43: 601–8.
- Ben Smail D, Jacq C, Denys P, Bussel B. Intrathecal baclofen in the treatment of painful, disabling spasms in Freidreichs ataxia. Mov Disord 2005; 20: 758–59.
- Lambrecq V, Muller F, Joseph PA, Cuny E, Mazaux JM, Barat M. Intérêts et limites du baclofène dans les paraparésies spastiques héréditaires. Ann Readapt Med Phys 2007; 50: 577–81.

ž ž			<i>(X</i>)	since	6000	(µg/24	(Yt			Score
1 Ii				(ou:k)		start	3			Satisfied
	euronal ceroid oofuscinosis type1	Σ	×	0:6	Reduction of contractures UE and spasms facilitation of care	100	182		Tolerance	9 Yes
2 o di or	rogressive eurodegenerative ^s sease of unknown 'igin	щ	10	1:4	Reduction of spasticity and hyperkinesias	100	202	Temporary hypotension (trial) Catheter migration and re-implantation	Scoliosis Tolerance	No
e ui	pastic tetraplegia of nknown origin	ш	×	4:1	Reduction of spasticity LE	50	150	CSF leak (trial)		6 Yes
4 dí	pinocerebellar egeneration	Σ	4 °o	6:8	Reduction of spasticity LE and ataxia	75	155	CSF leak (trial) Back pain (trial)	Abdominal discomfort Increase drooling	7 Yes
с Г Б	rogressive :ukoencephalopathy	Σ	13 18^{*}	6:6	Reduction of pain	135	367	Catheter migration Withdrawal symptoms (early empty pump) Overdose of ITB Pneumonia	Progressive scoliosis Loss of head balance Swallowing difficulties Constipation	6 Partially
6 ≥ =	letachromatic ukodystrophy	ш	2	0:8	Reduction of pain Facilitation of care	65	70	Temporary somnolence (trial) Hypotension (trial)		10 Yes

Table 1. ITB treatment: complications, side effects and satisfaction scores

male; mo, month; PEG, percutaneous endoscopic gastrostomy; UE, upper extremities y, year; * pump replacement. Abbreviations used: CSF, cerebrospinal fluid; F, temale; FU, last follow up during clinic visit; n, nour; H b, intratriecal

7

CHAPTER 8

The effect of intrathecal baclofen treatment on activities of daily life in children and young adults with cerebral palsy and progressive neurological disorders

LA Bonouvrié, JG Becher, Al Buizer, D Soudant, WJR van Ouwerkerk, GF Vles, RJ Vermeulen

Eur J Paediatr Neurol 2016; 20: 538-543

ABSTRACT

Introduction Intrathecal baclofen (ITB) treatment is applied in patients with spastic cerebral palsy (SCP), dystonic cerebral palsy (DCP) and progressive neurological disease (PND). Our aim was to investigate whether ITB treatment has a different effect on activities of daily life (ADL) in these groups.

Method A retrospective and cross-sectional survey was conducted using a questionnaire to assess the qualitative effect of ITB (Likert scale) on different domains of functioning (mobility, personal care, communication, comfort) and satisfaction with the results. Groups were compared using non-parametric statistics.

Results Questionnaires were completed for 68 patients (39 SCP, 13 DCP, 16 PND). Satisfaction scores were relatively high in all groups (7e8) and the positive effect on personal care and communication was similar in all groups. The PND group had the shortest follow-up and scored significantly less favorably for the effect on mobility and comfort.

Discussion This is the first study to show that ITB treatment has similar effects on personal care and communication in stable and progressive neurological disease. The decrease in mobility in the PND group is likely due to the progressive nature of the disease. The different effect on comfort between groups is mainly due to the smaller effect on startles in the PND group.

Keywords Baclofen, Child, Cerebral palsy, Nervous system diseases, Muscle spasticity, Dystonia

INTRODUCTION

Intrathecal baclofen (ITB) treatment is a well-known and widely used treatment for individuals with severe spasticity due to stable underlying conditions. Studies describe a profound decrease of spasticity in patients with spastic cerebral palsy (SCP) during ITB treatment.^{1,2} ITB also seems to have an effect in patients with secondary dystonia.³

The effect of ITB on spasticity CP and other neurological disorders is mediated by inhibition of the GABA receptor located on the spinal interneuron. In contrast, how ITB affects dystonia is unclear.

In conjunction with the reduction of spasticity and/or dystonia, it has been shown that individual problems of daily life, such as pain, transfers, sitting and ease of care improve with ITB treatment.⁴⁻⁶ In 70% of patients with SCP or dystonic cerebral palsy (DCP), with predominantly Gross Motor Function Classification System (GMFCS) level IV and V, goals on activity and participation level are attained.⁷

Other effects of ITB treatment in patients with DCP are also comparable with the effects of ITB treatment in patients with SCP but fewer studies are available. Studies show that care is facilitated and that individual problems of daily life improve.⁷⁻⁹ Additionally, ITB treatment decreases dystonia in patients with DCP.^{8,10}

The experiences with ITB treatment in patients with progressive neurological disorders of childhood (PND) are limited. A previously published review found only six case studies which describe ITB treatment in patients with spasticity caused by PND¹¹. The review reported a decrease of spasticity and pain and an improvement in several activities of daily care.

Caregivers of both CP and PND patients are generally satisfied with ITB treatment.^{1,5,9,12} It is, however, unknown if patients with these different conditions experience the same problems in daily life and if they benefit from ITB treatment in the same way. Therefore, the aim of this study was to compare the effect of ITB treatment on activities of daily life in patients with SCP, DCP and PND.

METHOD

Design

Combined retrospective and cross-sectional survey.

Patients

In 2007 and 2011 patients were selected from a list of patients who underwent ITB pump implantation since 2001. Patients who were using ITB treatment for less than two years at the time of the first survey in 2007 were again sent the questionnaires in 2011. They were included if they: 1) had undergone pump implantation (Medtronic, Inc., Minneapolis, MN, USA) in the VU University Medical Center in Amsterdam or the Maastricht University Medical Center in Maastricht, the Netherlands and/or were followed up in one of these centers; 2) were using ITB treatment at the time of the survey; and 3) suffered from SCP, DCP (both with onset before the first birthday) or PND (onset before the age of 18 years). Patients were excluded if their knowledge of the Dutch language was insufficient to complete the questionnaires or if >10% of the questions were left unanswered. Patients were contacted by phone (by LB and DS) and after agreeing to participate they were sent the questionnaires by mail.

Medical charts were checked for gender, diagnosis, GMFCS level, age at pump implantation and time since pump implantation. For PND a GMFCS level was estimated on basis of chart information about functioning.

The medical ethical committee of the VU University Medical Center in Amsterdam approved the use of the current questionnaires, without the use of a formal informed consent.

Questionnaire

A questionnaire (Appendix I and II) was sent to caregivers of the included patients. The questionnaire consists of several items (questions), which are divided into four domains of functioning: mobility (fine and gross motor skills), personal care, communication and comfort. Each domain consists of multiple questions focusing on tasks within that domain. For example, within the domain mobility one is asked about transfer, sitting and hand use, within the domain of personal care there are questions about dressing, hygienic care and eating/feeding, for communication about the interaction with other children and the ease of being understood. For comfort there are questions about pain and wellbeing. Since we had a clinical observation that startle reactions often occur,

we also included a question about startles. We defined a startle as generalized motor reaction upon auditory, tactile or visual stimulation.¹³

The caregivers were asked to compare the current situation (with ITB treatment) with the situation before pump implantation (without ITB treatment). Changes due to ITB treatment were classified using a five point Likert scale. Possible outcomes were: much better, somewhat better, no change, somewhat worse or much worse. Furthermore, satisfaction was scored on a Visual Analogue Scale (VAS), in which "0" was the worst score and "10" was the best score.

We chose not to use existing scales as the Pediatric Evaluation of Disability Inventory (PEDI) or the WeeFIM for two reasons: 1. Our study is a retrospective study looking into the effect of ITB treatment, therefore we cannot compare pre and post inventories for our patients. 2. Our patients are mostly GMFCS 4 and 5. The PEDI and WeeFIM are not suitable to evaluate the current situation for these severely affected patients and compare between groups since they will have minimal scores (floor effect).

Statistics

Group characteristics were compared for mean age, time of follow up, age of pump implantation, gender and GMFCS level using a Chi-squared test. Scores for the effect of ITB treatment were divided in three categories: improvement (much better, somewhat better), no change, or deterioration (much worse, somewhat worse). Scores were compared between groups per domain and per item using a non-parametric independent samples (Kruskal-Wallis) test. If significant differences were found with the non-parametric independent samples (Kruskal-Wallis) test, a Mann-Whitney test was used to determine individual group differences. p<0.05 was considered statistically significant. All analyses were performed with IBM SPSS version 20.

RESULTS

Patients

Eighty-three patients met the inclusion criteria in 2011. Four patients refused to participate when contacted by phone. Therefore, questionnaires were sent to 79 patients. Seventy patients returned the questionnaires. Two questionnaires were incomplete (>10% missing answers) and were excluded from analysis. Questionnaires of 68 patients (82% overall response rate) were included in the analysis. The mean age of the patients was 17.3 years (standard deviation [SD] 5.9; range 5.5-31.9), GMFCS II n=1, GMFCS III n=7, GMFCS IV n=15, GMFCS V n=45, mean age at implantation was

12.9 years (SD 5.0; range 4.6-28.2), follow-up 4.4 years (SD 2.7; range 0.3-9.3) male to female ratio, 40:28. In 2007 the same questionnaires had been sent to six patients with PND. Three of these patients were treated with ITB for less than two years at that time. At follow up two of these three patients deceased in the previous year due to respiratory problems. A new questionnaire was sent to the other patient. For this patient the most recent questionnaires were used in data analysis. For the other three previously included patients the questionnaire available from 2007 was used. Of those three patients, two were still using ITB at the time of the new survey. One pump was explanted after nine years due to insufficient effect of ITB treatment.

Of the 68 patients, 39 were diagnosed with SCP, 13 with DCP and 16 with PND. The PND group included patients with hereditary spastic paraparesis (HSP, n=5), metachromatic leukoencephalopathy (n=2), Leigh's disease (n=1) neuronal ceroid lipofuscinosis type 1 (n=1), mitochondrial encephalopathy (n=1), progressive primary dystonia (n=1), spinocerebellar syndrome (n=1), neurodegenerative disease (n=1), progressive bilateral spastic paresis (n=1), motor neuron disease (n=1). Patient characteristics are presented in Table 1.

Diagnosis	Ν	Mean age, yrs (SD; range)	GM II II	IFCS I IV \	/		Mean age implantation, yr (SD; range)	Mean follow up, yrs (SD; range)	Gender M:F
SCP	39	17.4 (5.0; 8.4-28.5)	0	5	12	22	12.4 (4.0; 5.0-24.1) ^{*2}	5.0 (2.5; 0.5-9.3) ^{*3}	25:14
DCP	13	20,5 (6.5; 8.4-31.9)*3	1	0	1	11	16.0 (5.3; 7.0-24.4) ^{*1,3}	4.5 (3.2; 1.2-9.2)	6:7
PND	16	14,4 (5.9; 5.5-30.4)*2	0	2	2	12	11.7 (6.3; 4.6-28.2) ^{*2}	2.8 (1.9; 0.3-6.6) *1	9:7
Total	68	17.3 (5.9; 5.5-31.9)	1	7	15	45	12.9 (5.0; 4.6-28.2)	4.4 (2.7; 0.3-9.3)	40:28

Table 1. Patient characteristics

N, number; yrs, years; SD, standard deviation; GMFCS, gross motor functioning classification system; M, male; F, female; SCP, spastic cerebral palsy; DCP, dystonic cerebral palsy; PND, progressive neurological disease; *, p<0.05; ¹, compared to spastic CP; ², compared to dystonic CP; ³, compared to progressive neurological disease.

Questionnaire

The effect of ITB treatment on the level of functioning

For the four domains (mobility, personal care, communication and comfort) the SCP and DCP patients show significantly more improvement than the PND patients for mobility and comfort (Fig. 1a-d). When looking into the different items, the only significant difference is found in the effect on startles (domain comfort). The SCP and DCP groups score significantly better for the effect on startles than the PND group. In contrast to the
SCP and DCP group where startles improve in some patients, startles remain unchanged for all PND patients (Fig. 2). Within the PND group, subgroup analysis was done for the HSP group which did not reveal differences for HSP with the rest of the group for any of the outcome measures (data not shown).

Expectations are met in 82.1% of the SCP patients, compared to 84.6% of the DCP patients and 78.6% of the PND patients, which is not significantly different. The SCP and PND groups scored satisfaction as 7.0 on a 10 point VAS scale. The DCP group scored 8.0. Satisfaction scores do not differ significantly between groups.

DISCUSSION

The aim of this study was to compare the effect of ITB treatment on ADL in patients with SCP, DCP and PND. We used a broad questionnaire for these relatively severely affected patients in order to get an impression on the extent of the effect on all relevant domains.

The effect of ITB treatment on the level of functioning

Previous studies showed positive effects of ITB treatment on individual problems of daily life.^{4,5,7-9} Treatment goals are often met.⁷ The experiences with ITB treatment in PND patients in childhood are limited11 but seem to be comparable to ITB effects in CP patients.^{12,14,15} ITB treatment decreases both dystonia and spasticity.^{1,2,8,10,16} We compared the effect of ITB treatment on activities of daily life between SCP, DCP and PND and found a significant difference in the effect on mobility and comfort between the PND group and the two CP groups. The PND group is rather heterogeneous. Subgroup analysis was not feasible due to the small numbers, except for the HSP group (n=5), which did not differ from the rest of the PND group on any of the outcome measures.



Figure 1. (a) Changes in mobility during intrathecal baclofen treatment (percent of total number of items within domain). There is a statistically significant difference between the CDP group and the PND group with p = 0.047 and Z-score = -1.988 and furthermore between the spastic group and the progressive group with p = 0.008 and Z-score = -2.649 (Mann Whitney). (b) Changes in personal care during intrathecal baclofen treatment (percent of total number of items within domain). There are no significant differences between groups. (c) Changes in communication

Chapter 8





during intrathecal baclofen treatment (percent of total number of items). There are no significant differences between groups. (d) Changes in comfort during intrathecal baclofen treatment (percent of total number of items within domain). There is a statistically significant difference between the DCP group and the PDN group with p = 0.000 and Z-score = -3,857 and furthermore between the SCP group and the PDN group with p = 0.001 and Z-score = -3,267 (Mann Whitney).



Figure 2. Effect of intrathecal baclofen treatment on startles. (percent of total number of items). There is a statistically significant difference between the DCP group and the PND group with p = 0.005 and Z-score = -2,831 and furthermore between the SCP group and the PND group with p = 0.008 and Z-score = -2,643 (Mann Whitney).

Mobility

Fewer patients showed improvement and more patients showed deterioration in the PND group compared to both CP groups. Deterioration of mobility is to be expected in patients with progressive neurological disease and ITB does not influence this process. In this study, the follow up time for the PND group was significantly shorter than the follow up time of the SCP group. This is partly due to the progressive nature of the disease with early death in some of these patients. For the patients with a longer survival, it is to be expected that over time, the difference between groups will increase even further.

Personal care

The effect on personal care was not significantly different between groups. As a result of decrease of spasticity and/or dystonia, it is presumable that care will be facilitated in all groups. We presume that longer follow up time for the PND group might demonstrate a significant differential effect in the different patient groups.

Communication

The differences between groups on the domain of communication did not reach statistical significance, however there was improvement in more patients in the DCP group than in the other groups (53.8% vs 30.8% (SCP) and 18.8% (PND)) but. Pueyo et al. showed that DCP patients showed better performance in auditory comprehension, visuospatial abilities, immediate visual memory and working verbal memory than SCP patients.¹⁸ Previous studies found that the comprehension of spoken and written language (receptive communication) was influenced by type of motor disorder, with the spastic CP patient being more affected than the dystonic patients.¹⁷ In addition, in severely affected (GMFCS IV and V) bilateral spastic and dystonic CP patients expressive communication is severely hampered.^{17,19}

Consequently, in our study we expect that the DCP patients will have better cognitive functions but all groups will have impaired expressive communication due to the severity of the motor disorder. The motor disorder is expected to improve with ITB treatment, resulting in an increased ability for expressive communication in all groups. However, comprehension will not change with ITB treatment. Accordingly, the DCP group, with their pre-existing good understanding, will have more capacities for interaction and communication than the SCP and PND groups, for whom the receptive communication will not change, resulting in a less evident change in interaction/communication.

There are several possible explanations why we did not find a significant statistical difference in the effect of ITB treatment on communication. Firstly, the number of patients is low. Secondly, the domain of communication consisted of only one item. In the questionnaire no distinction is made between receptive and expressive communication. It is possible that with more items, the difference would have been clearer. Thirdly, we omitted to ask for the use of communication aids and as a result we were not able to control for this important factor.

Comfort

The PND group showed more deterioration and less improvement in comfort during ITB treatment than the two CP-groups. The only item in this domain that was different

between groups was the effect on startles. This reflex is more profound in CP children than in typically developing children.²⁰ It is thought that perceptual disorders are the underlying cause of the startle reflex in children with cerebral palsy.¹³ Startles did not change in the PND group in contrast to the SCP group and the DCP group where respectively 43.2% and 53.8% of patients improved. For the CP-groups, this is consistent with the findings of Krach et al. who showed a decrease of startles of 54% with ITB.²¹ Patients with significant startles should be counselled before giving ITB treatment that decreasing spasticity does not necessarily mean decreasing startles.

In questioning about satisfaction, some of our patients reported an increase of scoliosis resulting in more discomfort. Scoliosis can occur or progress in patients during ITB treatment but ITB treatment has not been proven to influence the development or progression of scoliosis,²² also when compared to a control group of similar patients without ITB.²³ Scoliosis is a common complication that may develop during growth in severe neurological disease.¹¹ Our study did not ask about scoliosis specifically and only a few patients spontaneously reported scoliosis and related problems. We cannot draw conclusions for our patient group but we feel that counselling before ITB treatment is recommended. Patients must know that scoliosis can deteriorate or occur despite the treatment, and that monitoring for scoliosis is needed.

Satisfaction

There is no difference in satisfaction between groups. Previous studies show that most patients are satisfied with ITB treatment.^{7,12,21} A score of 7.0 on a VAS scale was given for overall satisfaction in PND patients¹² and 88% of patients with CP would choose for ITB treatment again if they had the choice.²¹ In both patient groups daily care improved, which in our opinion resulted in similar satisfaction scores.There are few treatment options for severe spasticity or dystonia and patients and parents might be content with every little improvement they experience, especially in PND where we do not expect much improvement over time given the progressive character of the disease. In PND, complaints are alleviated only temporarily and will worsen as time progression, which is unrelated to ITB treatment. Therefore, they might be able to put things in perspective and see how ITB treatment benefits their daily lives.

Limitations

This study has several limitations. The main limitation is the retrospective setup of the study, which resulted in different numbers of the patient groups and different followup timing. Due to the limited number of patients we were not able to perform a multivariate analysis to correct for multiple factors. However, this is the largest multicenter study addressing these issues.

The parents were responsible for the responses in the majority of questionnaires (66 of 68 questionnaires). Two questionnaires were filled in by caregivers other than the parents. We do not know if caregivers incorporated the patient's opinion in their answers. Furthermore we asked caregivers to recall the situation at the time of the pump implantation, a few weeks to a few years before this survey. Nevertheless, caregivers reported detailed information on both amelioration and deterioration after starting ITB treatment.

Scales used in the questionnaires have not been validated. No other validated scales were available at the time of this survey to measure ADL in GMFCS level 4 and 5 patients. Outcome measures of the questionnaire were chosen on basis of previous papers^{10,24,25} and the expertise of our clinic's health care professionals with the main problematic areas and therapeutic goals for children and young adults with spasticity.

CONCLUSION

With regard to the effect of ITB treatment on the level of functioning, ITB treatment results in almost similar improvement of ADL for patients with SCP, DCP and PND. The PND group shows a less favorable effect on mobility and comfort. In all groups we noticed variation of the effects, some show improvement, others deterioration and many show no change. Current residual difficulties are comparable for all groups. We would expect to find a larger difference between the PND group and the two CP groups since the PND group is expected to worsen over time. It will be interesting to evaluate these patients after a similar follow up time as the CP groups even though this will not be possible for some of the patients due to the progressive nature of their disease with an early death. We expect that there will be more significant differences between the PND group and the CP groups over time.

Prospective, maybe even randomized controlled, trials are needed to obtain higher scientific evidence about the effect of ITB treatment on activities and participation and furthermore to determine which patients are most likely to benefit.

ACKNOWLEDGEMENTS

Patients and caregivers are gratefully thanked for their cooperation.

REFERENCES

- Albright AL, Gilmartin RC, Swift D, Krach LE, Ivanhoe CB, McLaughlin J. Long-term intrathecal baclofen therapy for severe spasticity of cerebral origin. J Neurosurg 2003;98:291e5.
- Guillaume D, Van HA, Vloeberghs M, Vidal J, Roeste G. A clinical study of intrathecal baclofen using a programmable pump for intractable spasticity. Arch Phys Med Rehabil 2005;86:2165e71.
- Albright AL, Barry MJ, Painter MJ, Shultz B. Infusion of intrathecal baclofen for generalized dystonia in cerebral palsy. J Neurosurg 1998;88:73e6.
- Hoving MA, van Raak EPM, Spincemaille GHJJ, Palmans LJ, Becher JG, Vles JSH. Efficacy of intrathecal baclofen therapy in children with intractable spastic cerebral palsy: a randomized controlled trial. Eur J Paediatr Neurol 2009;13:240e6.
- Hoving MA, van Raak EP, Spincemaille GH, van Kranen-Mastenbroek VH, van KM, Gorter JW, Vles JS. Safety and one year efficacy of intrathecal baclofen therapy in children with intractable spastic cerebral palsy. Eur J Paediatr Neurol 2009 May;13:247e56.
- Vles GF, Soudant DL, Hoving MA, Vermeulen RJ, Bonouvrié LA, van Oostenbrugge RJ, Vles JS. Long-term follow-up on continuous intrathecal baclofen therapy in non-ambulant children with intractable spastic cerebral palsy. Eur J Paediatr Neurol 2013;17:639e44.
- Ward A, Hayden S, Dexter M, Scheinberg A. Continuous intrathecal baclofen for children with spasticity and/or dystonia: goal attainment and complications associated with treatment. J Paediatr Child Health 2009;45:720e6.

- Bonouvrié LA, van Schie PEM, Becher JG, van Ouwerkerk WJ, Reeuwijk A, Vermeulen RJ. Effects of intrathecal baclofen on daily care in children with secondary generalized dystonia: a pilot study. Eur J Paediatr Neurol 2011;15:539e43.
- Motta F, Stignani C, Antonello CE. Effect of intrathecal baclofen on dystonia in children with cerebral palsy and the use of functional scales. J Pediatr Orthop 2008 March;28(2):213e7.
- Albright AL, Barry MJ, Shafron DH, Ferson SS. Intrathecal baclofen for generalized dystonia. Dev Med Child Neurol 2001;43(10):652e7.
- Bonouvrié LA, van Schie PEM, Becher JG, van Ouwerkerk WJ, Vermeulen RJ. Intrathecal baclofen for progressive neurological disease in childhood; a systemic review of literature. Eur J Paediatr Neurol 2012;16:279e84.
- Bonouvrié LA, van Schie PEM, Becher JG, van Ouwerkerk WJ, Vermeulen RJ. Satisfaction with intrathecal baclofen treatment in paediatric patients with progressive neurological disease. Dev Med Child Neurol 2008;50:636e8.
- Ferrari A, Sghedoni A, Alboresi S, Pedrni E, Lombardi F. New definitions of 6 clinical signs of perceptual disorders in children with cerebral palsy: an observational study though reliability measures. Eur J Phys Rehabil Med 2014;50:709e16.
- Ben Smail D, Jacq C, Denys P, Bussel B. Intrathecal baclofen in the treatment of painful, disabling spasms in Freidreich's ataxia. Mov Disord 2005;20:758e9.
- Chu MLY, Sala DA, Weiner HL. Intrathecal baclofen in X-linked adrenoleukodystrophy. Pediatr Neurol 2001;24:156e8.

- Albright AL. Intrathecal baclofen for childhood hypertonia. Childs Nerv Syst 2007;23:971e9.
- Vos R, Dallmeijer AJ, Verhoef M, van Schie PE, Voorman JM, Wiegerink DJ, Geytenbeek JJM, Roebroeck ME, Becher JG, PERRIN Study Group. Developmental trajectories of receptive and expressive communication in children and young adults with cerebral palsy. Dev Med Child Neurol 2014;56:951e9.
- Pueyo R, Junqu_e C, Vendrell P. Neuropsychologic differences between bilateral dyskinetic and spastic cerebral palsy. J Child Neurol 2003;18:845e50.
- Geytenbeek JJ, Vermeulen RJ, Becher JG, Oostrom KJ. Comprehension of spoken language in non-speaking children with severe cerebral palsy: an explorative study on associations with motor type and disabilities. Dev Med Child Neurol 2014;57:294e300.
- Goldberg J, Anderson DW, Wilder S. Startle reflex habituation in children with cerebral palsy. Percept Mot Skills 1979;48:1135e9.

- 21. Krach LE, Nettleton A, Klempka B. Satisfaction of individuals treated long-term with continuous infusion of intrathecal baclofen by implanted programmable pump. Pediatr Rehabil 2006;9:210e8.
- 22. Senaran H, Shah SAM, Presedo A, Dabney KWM, Glutting JWP, Miller F. The risk of progression of scoliosis in cerebral palsy patients after intrathecal baclofen therapy. Spine 2007;32:2348e54.
- 23. Shilt JS, Lai LPM, Cabrera MNM, Frino J, Smith BPP. The impact of intrathecal baclofen on the natural history of scoliosis in cerebral palsy. J Pediatr Orthop 2008;28:684e7.
- 24. Schneider JW, Gurucharri LM, Gutierrez AL, Gaebler-Spira DJ. Health related quality of life and functional outcome measures for children with cerebral palsy. Dev Med Child Neurol 2001;43:601e8.
- 25. Staal C, Arends A, Ho S. A self-report of quality of life of patients receiving intrathecal baclofen therapy. Rehabil Nurs 2003;28:159e63.

APPENDIX I. Questionnaires Part 1

De effecten van intrathecale baclofen behandeling

VOORBEELD INGEVULDE VRAGENLIJST

Naam van kind: Jan

Deze vragenlijst is ingevuld door: de moeder van Jan

		\frown	\frown	
1. Kan uw kind zich verplaatsen?	Zelfstandig	(Met hulp)	In rolstoel	Anders
		\bigcirc	\smile	
 Zo ja, doet hij/zij dat ** 				
Evt. nadere toelichting met hulp	o/ondersteunín	g van één pi	ersoon	
			\frown	<
 Is dit veranderd door de baclofer 	n behandeling v	ia de pomp	? (Nee
	\frown			
• Zo ja, gaat het verplaatsen *	eel beter 🛛 lets be	eter) lets sle	chter 🔥	/eel slechter

2. Zijn onderstaande verzorgingspunten veranderd door de baclofen behandeling?

Aankleden [*]	Ja		Nee			
Zo ja gaat dit nu*	Veel beter	lets beter	lets slechter Veel slechter			
Voeden/eten*	Ja		Nee			
Zo ja, gaat dit nu [*]	Veel beter	lets beter	lets slechter Veel slechter			

Naam van kind:				
Deze vragenlijst is ingevuld door:				
Huidige dosering (zie uitdraai pomp: d	aily dose): .			
1. Kan uw kind zich verplaatsen?*	Ja	Nee		
• Zo ja, doet hij/zij dat ^{**}	standig I	Met hulp	In rolstoel	Anders
Evt. nadere toelichting				
• Is dit veranderd door de baclofe	n behandel	ing via de	pomp ?*	Ja Nee
• Zo ja, gaat het verplaatsen nu [*] _	Veel beter	lets beter	lets slechter	Veel slechter
2. Kan uw kind zelfstandig zitten? [*]	Ja N	ee		
• Is dit veranderd door de baclofe	n behandel	ing via de	pomp?*	a Nee
• Zo ja, is dit nu [*] Veel beter let	s beter let	s slechter	Veel slechter	_
3. Gebruikt uw kind zijn of haar hande	n?* Ja	Nee		
Is dit veranderd door de baclofe	n behandel	ing via de	pomp ?*	Ja Nee
• Zo ja, is dit nu [*]	Veel beter	lets beter	lets slechter	Veel slechter
4. Kan uw kind zijn/haar hoofd zelfstar	ndig goed re	echtop ho	uden?* Ja	Nee
 Is dit veranderd door de baclofe 	n behandel	ing via de	pomp?*	Ja Nee
• Zo ja, is dit nu [*]	Veel beter	lets beter	lets slechter	Veel slechter

* omcirkelen wat van toepassing is.

** omcirkelen wat van toepassing is, meerdere antwoorden mogelijk.

Ch	ap	oter	. 8
----	----	------	-----

5. Krijgt uw kind eten v	via de mond	?* Ja	Nee			
• Zo nee, hoe krijg nl	t uw kind vc	eding binn	en? Is dat via	een PE	G-sonde	/anders*
 Is het eten/voed behandeling via de 	Ja	Nee				
• Zo ja, is dit nu [*]	Veel beter	lets beter	lets slechter	Veel	slechter	
6. Is uw kind bij de vera	zorging volle	edig afhank	elijk van ande	ren?*		
• Zo niet, wat kan	hij of zij zelf	doen?				
• Vindt u de verzo	 rging makke	liiker gaan s	sinds de —			

baclofen behandeling?*

• Zijn onderstaande verzorgingspunten veranderd door de baclofen behandeling?

Ja

Nee

Aankleden [*]	Ja			Nee
Zo ja gaat dit nu *	Veel beter	lets beter	lets slechter	Veel slechter
Voeden/eten*	Ja			Nee
Zo ja, gaat dit nu*	Veel beter	lets beter	lets slechter	Veel slechter
Wassen/baden/douchen*	Ja			Nee
Zo ja, gaat dit nu [*]	Veel beter	lets beter	lets slechter	Veel slechter
Ontlasting (poepen)*	Ja			Nee
Zo ja, gaat dit nu*	Veel beter	lets beter	lets slechter	Veel slechter

*omcirkelen wat van toepassing is

The effect of ITB: CP vs PND

Blaascontrole (plassen)*		Ja		Nee			
Zo ja, gaat dit nu	* Veel bet	er let	s beter	lets sle	echter	Veel slechter	
7. Is de mogelijkheid tot tra	nsfer (verpl	aatsen va	anuit be	d naar	(rol)stoel e	en andersom)	
veranderd door de baclofer	n behandeli	ng?*		Ja	Nee		
• Zo ja, is transfer*	'eel beter I	ets beter	lets sle	echter	Veel slech	ter	
• Kunt u toelichten wat	er precies	verander	d is?				
3. Is er communicatie moge	lijk tussen i	u en uw k	ind?*	Ja	Nee		
• Zo ia hoe communice	ert 11 met 1	w kind?	-				
• Is dit veranderd door	de baclofer	n behand	eling?*	Ja	Nee		
• Zo ja, gaat de commu	nicatie nu *	Veel bet	er lets	s beter	lets slecht	er Veel slech	
9. Hoe zou u de stemming v	an uw kind	over het	algeme	en om	schrijven?		
 Is de stemming van uv 	w kind vera	nderd do	or de			Nee	
baclofen behandeling?	*						
• Zo ja, is de stemming	van uw kin	d*					
	Veel beter	lets be	ter le	ets slecht	er Veels	slechter	

*omcirkelen wat van toepassing is.

10. Heeft uw kind last van plotselinge spierschokken (spasmen)?* Ja Nee • Zijn deze spierschokken veranderd in hoeveelheid of Ja Nee ernst door de baclofen behandeling?* • Zo ja, is het * Veel beter lets beter Veel slechter lets slechter 11. Heeft uw kind last van schrikken (bv hevige reactie op geluid)?* Ja Nee Is dit veranderd door de baclofen behandeling?* Ja Nee • Zo ja, is het schrikken^{*} Veel beter lets beter lets slechter **Veel slechter** 12. Heeft uw kind problemen met slapen s'nachts?* Ja Nee Is daar een bekende reden voor?* Ja Nee • Zo ja, welke? • Is de slaap van uw kind veranderd door de Ja Nee baclofen behandeling?* Zo ja, is de slaap^{*} Veel beter lets beter lets slechter **Veel slechter** Kunt u vertellen wat er precies aan veranderd is?

*omcirkelen wat van toepassing is.

Chapter 8

The effect o) of	ITB:	СР	VS	PND
--------------	------	------	----	----	-----

13. Heeft u het idee dat	uw kind las	t heeft var	n pijn? *	J	la		Nee			
• Zo ja, waardoor he	eeft uw kinc	l pijn?								
• Is dit veranderd do	oor de baclo	ofenbehan	deling?*	c	Ja		Nee			
• Zo ja, is de pijn [*]	Veel beter	lets beter	lets si	lechte	?r	Veel	slech	nter	-	
14. Heeft u het idee dat	uw kind lek	ker in zijn/	'haar ve	l zit?	*	Ja		N	ee	
• Is dit veranderd do	oor de baclo	ofenbehan	deling?*	¢	-	Ja		N	ee	
• Zo ja, hoe zit uw k in zijn/haar vel*	ind nu 	l beter le	ts beter	let	s slec	hter	v	eel sl	echte	er
15. Ondervindt uw kind	ook problei	men van d	e baclof	enbe	ehan	delir	۱g? *	J	a	Nee
• Zo ja, welke proble	emen?									
16. Heeft de baclofen be die u verwacht/gehoopt	ehandeling o had?*	de verbete	rig gebr	acht	-	Ja		N	ee	
			abandal	ing?						
• Hoe tevreden ben	t u over de	baclofenb	enanuei	0						
 Hoe tevreden ben Geef een cijfer van 2 	t u over de L tot 10	baclofenbe	2 3	4	5	6	7	8	9	10
 Hoe tevreden ben Geef een cijfer van 2 Kunt u dit toelichter 	t u over de L tot 10 en?	baclofenbo	2 3	4	5	6	7	8	9	10

* omcirkelen wat van toepassing is.

APPENDIX II. Questionnaires Part 2

Probleem scoring

VRAGENLIJST DEEL 2: PROBLEEMSCORING

Naam van kind:	
Naam van verzorger:	

Datum:....

Wilt u hieronder aangeven hoe gemakkelijk of moeilijk het voor u of uw kind is om elk van de volgende taken uit te voeren. Op de onderstaande balken staan cijfers. Omcirkel het cijfer dat volgens u overeenkomt met de pijn, beperkingen of klachten **op dit moment**. Als een vraag niet van toepassing is, omcirkel dan NVT.

VOORBEELD INGEVULDE VRAGENLIJST

Luiers	verwiss	elen?									
Niet m	oeilijk								He	eel erg r	noeilijk
0	1	2	3	4	5	6	7	8	9	10	(NVT)
Uitleg:	Dit kin	d draa	gt geen	luiers							\bigcirc
Heeft u	ıw kind	pijn?									
Geen p	oijn									Onhoud	lbarepijn
0	1	(2)	3	4	5	6	7	8	9	10	NVT
Uitleg:	Dit kin	d heeft	geen e	erge pij	n						
Het gei	mak var	n transf	fers (ve	rplaats	en vanı	it bed	in de (r	ol)stoe	l en anc	dersom)?
Niet m	oeilijk							_	He	eel erg	moeilijk
0	1	2	3	4	5	6	7	$\left(8 \right)$	9	10	NVT
Uitleg:	Het is e	erg mo	eilijk oı	<mark>n dit k</mark> i	i <mark>nd te v</mark>	erplaat	tsen m	aar het	lukt no	og wel	

Persoonlijke verzorging

Niet moeilijk

1. Het a	aan- en	uitkled	len?								
Niet m	Niet moeilijk Heel erg moeilijk										
0	1	2	3	4	5	6	7	8	9	10	NVT
2. Luiers verwisselen?											
Niet mo	oeilijk									Heelerg	gmoeilijk
0	1	2	3	4	5	6	7	8	9	10	NVT
3. Bips schoon maken?											
Niet mo	oeilijk									Heelerg	gmoeilijk
0	1	2	3	4	5	6	7	8	9	10	NVT
4. Gem	ak van	toiletga	ang (po	epen)?							
Niet mo	oeilijk									Heelerg	gmoeilijk
0	1	2	3	4	5	6	7	8	9	10	NVT
5. Gem	ak van	baden/	douche	en/was	sen?						
Niet mo	oeilijk									Heelerg	gmoeilijk
0	1	2	3	4	5	6	7	8	9	10	NVT
6. Gem	6. Gemak van eten c.q. voeren?										

0 1 2 3 4 5 6 7 8 9 10 N												
	0	1	2	3	4	5	6	7	8	9	10	NVT

Heel erg moeilijk

7. Hoe tevreden bent u met de vooruitgang van uw kind betreffende persoonlijke verzorging door de intrathecale baclofen?

Zeer tevreden

Helemaal niet tevreden

0	1	2	3	4	5	6	7	8	9	10	NVT

Positioneren/ verplaatsen

8. Het gemak van positioneren in de rolstoel?

Niet m	oeilijk									Heeler	gmoeilijk
0	1	2	3	4	5	6	7	8	9	10	NVT

9. Het gemak van positioneren buiten de rolstoel?

Niet m	oeilijk									Heelerg	gmoeilijk
0	1	2	3	4	5	6	7	8	9	10	NVT

10. Het gemak van transfers (verplaatsen vanuit bed in de (rol)stoel en andersom)?

Niet m	oeilijk									Heeler	gmoeilijk
0	1	2	3	4	5	6	7	8	9	10	NVT

11. Het gemak van het aantrekken van braces of het positioneren van hulpmiddelen?

Niet	m	oeilijk									Heeler	gmoeilijk
0		1	2	3	4	5	6	7	8	9	10	NVT

12. Hoe tevreden bent u met de vooruitgang van uw kind op het gebied van positionering en transfers door de intrathecale baclofen?

Heel te	Ieel tevreden Helemaal niet tevreden 0 1 2 3 4 5 6 7 8 9 10 NVT													
0	1	2	3	4	5	6	7	8	9	10	NVT			
Comfo	rt													
13. Mij	n kind i	s mees	tal gezo	ond en	actief									
Mee ee	ens									Mee	oneens			
0	1	2	3	4	5	6	7	8	9	10	NVT			
14. Hoe is de stemming van uw kind? Zeer goede stemming Zeer slechte stemming														
O 1 2 3 4 5 6 7 8 9 10 NVT														
15. Hee Geen p	0 1 2 3 4 5 6 7 8 9 10 NV1 15. Heeft uw kind pijn? Geen pijn Onhoudbarepijn													
0	1	2	3	4	5	6	7	8	9	10	NVT			
16. Hee Nooit	16. Heeft uw kind last van plotselinge spierschokken Nooit Zeer vaak													
0	1	2	3	4	5	6	7	8	9	10	NVT			
17. Hee	.7. Heeft uw kind last van schrikken (bv van geluiden)													

Nooit										Ze	er vaak
0	1	2	3	4	5	6	7	8	9	10	NVT

Chapter 8

Interactie/ communicatie

18. Hoe gemakkelijk is het voor uw kind om met andere kinderen te spelen?

Zeer ge	makke	lijk								Onr	nogelijk
0	1	2	3	4	5	6	7	8	9	10	NVT

19. Hoe gemakkelijk is het voor uw kind om helemaal begrepen te worden door anderen die uw kind goed kennen (broertjes/zusjes, uzelf)?

Zeer ge	emakke	lijk								Onm	nogelijk
0	1	2	3	4	5	6	7	8	9	10	NVT

20. Hoe gemakkelijk is het voor uw kind om helemaal begrepen te worden door iemand die uw kind niet kent?

Zeer ge	emakke	lijk								Onmog	gelijk
0	1	2	3	4	5	6	7	8	9	10	NVT

21. Hoe tevreden bent u met de mogelijkheden van uw kind voor interactie en communicatie?

0	1	2	3	4	5	6	7	8	9	10	NVT

Helemaal niet tevreden

22. Beschrijf uw kind

Zeer tevreden

Zeer gelukkig										Zeer ongelukkig		
	0	1	2	3	4	5	6	7	8	9	10	NVT

Hartelijk dank voor het invullen van de vragenlijst.

U kunt deze nu in de bijgaande envelop aan ons terugsturen.

The effect of ITB: CP vs PND



The studies in this thesis focus on the effect of intrathecal baclofen (ITB) in severely affected children on different levels of the International Classification of Functioning, disability and health for Children and Youth (ICF-CY).

- The first aim was to investigate the effect of ITB in the treatment of dyskinetic cerebral palsy (CP). The focus was primarily on the effect on individual treatment goals, mostly on the levels of activities and participation, and environmental factors. The secondary focus was on the level of body functions and structures (dystonia, choreoathetosis, spasticity, pain, comfort).
- The second aim was to describe the effect and the current level of evidence of ITB treatment in progressive neurological disease (PND) of childhood on all ICF-CY levels.

In this chapter the main findings of this thesis are critically appraised. Subsequently, implications for clinical practice and directions for future research are provided.

ITB IN DYSKINETIC CEREBRAL PALSY

The majority of patients with dyskinetic CP are severely disabled with 59-80% of patients being classified in Gross Motor Functioning Classification System (GMFCS) level IV or V.¹⁻³ When pharmacological treatment is considered in order to decrease functional problems, the first step is oral medication such as baclofen, trihexyphenidyl, or gabapentin.⁴ The evidence for the use of oral pharmacological agents for treatment of dystonia and functional problems in dyskinetic CP is limited, and the level of evidence of the reported studies low.⁵⁻⁷ Efficacy of the different agents ranges from low to possibly ineffective.⁵⁻⁷ The second step in treatment of dystonia are advanced treatment options, such as ITB and Deep Brain Stimulation.⁵⁻⁷ These treatment options can possibly be effective, however evidence is limited due to lack of high quality studies.^{5,6,8}

Treatment goals for ITB in patients with severe cerebral palsy (GMFCS IV-V) are mostly to improve activities of daily living (e.g. ease of caregiving and sitting), decrease pain and improve quality of life.^{9,10} There are no selection criteria for ITB specifically formulated for patients with dyskinetic CP. In clinical practice, selection criteria analogous to those for spastic CP can be used.⁹ Following these criteria, ITB can be considered in patients with dyskinetic CP in whom dystonia interferes with activities of daily life or quality of life, and in whom other pharmacological treatment options are insufficiently effective.⁹

Only few previous studies have reported on the effect of ITB in children with severe dyskinetic CP.¹⁰⁻¹³ The scientific quality of these studies is low (one case series study, one small retrospective cohort study, and two (small and/or low quality) prospective cohort studies).¹⁰⁻¹³ In these studies, severity of dystonia is the most commonly used outcome measure.¹⁰⁻¹³ The Barry Albright Dystonia Score (BADS) is the most frequently used outcome measure to report the effect on the severity of dystonia.¹⁰⁻¹² Dystonia decreases in all studies.¹⁰⁻¹³ In addition to the effect on dystonia, structured interviews or questionnaires have been used to evaluate the effect on multiple levels of the ICF-CY.¹⁰⁻¹² On the level of body functions and structures improvement has been reported for quality of life, pain, mood, and sleep.¹⁰⁻¹² On the level of activities and participation improvement has been reported for feeding and swallowing, sitting and posture control, upper limb use and communication/speech.¹⁰⁻¹² In the majority of patients no change in autonomy of carrying out daily activities was seen.¹² Furthermore, there was no change in gross motor function.¹⁰ On the level of environmental factors, ease of care has been reported to improve in most patients.¹⁰⁻¹² Treatment goals were reported to be fully reached in the majority of patients.¹⁰ Furthermore, most patients reported to be satisfied with ITB treatment.¹²

Despite the apparent positive outcomes of these studies for the effect of ITB in dyskinetic CP on dystonia and functional problems (e.g. sitting, burden of care giving), the level of evidence provided by these studies is low. Further high quality studies were required to determine the treatment effects of ITB in this population.

The IDYS trial which is presented in chapter 4 and 5 of this thesis, is the first multicenter, double blind, placebo controlled, randomized trial addressing the effect of ITB in patients with severe dyskinetic CP (GMFCS IV-V), primarily on the attainment of individual treatment goals. Thirty-six patients participated in the trial, providing the study with sufficient power. Considering the complex multi-morbidity of patients with dyskinetic CP, a multi-disciplinary team consisting of (pediatric) physiatrists, (child) neurologists, neurosurgeons, clinical neurophysiologists, pharmacists, occupational therapists, physiotherapists, speech therapists and epidemiologists were involved in designing and executing the study.

In the IDYS trial we found that the ITB group scored significantly better on attainment of individually defined treatment goals compared to the placebo group. Our findings are in line with the previously described studies using structured interviews or questionnaires on different areas of daily life.¹⁰⁻¹² Additionally, this randomized controlled study adds substantially to the level of evidence, which was previously low.^{5,8}

Secondary outcome measures of the IDYS trial included the severity of dystonia (Dyskinesia Impairment Scale (DIS), BADS).¹⁴⁻¹⁶ We found a significant difference between groups in favor of the ITB group for the dystonia subscale and the dystonia in rest subscale of the DIS. The ITB group remained stable on these scores whilst the placebo group showed slight worsening. We did not find a significant difference between groups on the BADS. This is different from the findings of our pilot study (chapter 3) and other previously published studies, where the BADS showed a clinically significant decrease (decrease of 25% or more).¹⁰⁻¹³ These differences might be explained by the double-blind, placebo-controlled, randomized design of the IDYS trial, eliminating bias from the outcome assessments. Furthermore, we used a structured video-protocol for scoring, which was new compared to previous studies.

Individual treatment goals

ITB in dyskinetic CP is a symptomatic, and not curative, treatment to reduce the level of involuntary muscle activity, mainly in order to improve daily functioning and caregiving.^{5,6} Standardized questionnaires and outcome measures, available at the time of this thesis, such as the Pediatric Evaluation of Disability Inventory (PEDI) or Gross Motor Functioning Measure (GMFM), do not adequately capture the individual problems in daily life of severely affected patients with very limited motor skills (GMFCS IV and V), who were the focus in the studies in this thesis. Improvement of autonomy of carrying out daily activities is not likely to be achieved with ITB in most of these patients.¹² Gross motor function, measured with the GMFM, has been reported to improve slightly in some dyskinetic patients receiving ITB but on a group level no significant difference was found for the overall GMFM score.¹⁰ In a group of patients with spastic CP, mostly GMFCS IV-V, the sitting domain of the GMFM was reported to improve after one year of ITB.¹⁷ We would expect this to be similar for dyskinetic CP patients. Considering the severity of motor problems in GMFCS IV-V patients, improvement of other domains of the GMFM (e.g. rolling, crawling, standing and walking/running) are not expected to change with ITB and subsequently, were these motor functions were not reported as individual treatment goals.

ITB treatment goals for patients with dyskinetic CP were on multiple levels of the ICF-CY: the level of body functions and structures (e.g. pain and comfort), activities and participation (e.g. mobility including sitting and transfers), and environmental factors (caregiving by others) (Chapter 5). Goal attainment scaling (GAS) is a meaningful measure to determine treatment effects in severely affected patients with dyskinetic CP. Goals are set from a client centered perspective and can include different levels of the ICF-CY.^{18,19} Due to the inclusion criteria of the IDYS trial, our population was fairly homogeneous, but the GAS can also be used in heterogeneous populations where goals differ between patients in type and magnitude.²⁰ Furthermore, GAS can capture subtle changes that are important for the patient, but difficult to determine using standardized outcome measures. GAS development is, however, time consuming (45 minutes per child) and training is needed before use.^{18,19}

Measurement of dystonia and attainment of goals

A previous study by Monbaliu et al. showed lower functional abilities for children with higher levels of dystonia and in addition poorer scores on participation and quality of life questionnaires.²¹ Since ITB is aimed to reduce dystonia, we expected that a decrease of dystonia would induce attainment of individual treatment goals. In the IDYS trial we found a significant difference between ITB and placebo for the DIS dystonia subscale and dystonia in rest subscale. However, this difference was based on a slight increase of dystonia in the placebo group whereas dystonia the ITB group remained unchanged. We did not find a difference for the BADS between groups. We furthermore did not find a correlation between changes in dystonia scores and attainment of goals. Based on these findings, the hypothesis that treatment goals are attained due to a decrease of the level of dystonia cannot be confirmed.

As previously described, our findings for the BADS are in contrast with previous studies.¹⁰⁻¹³ These studies were not blinded, and as a consequence biased by the knowledge of observer and caregiver, resulting in a lower scientific level.

A limitation of both the DIS and BADS is that there are no test-retest reliability studies available for either scale. In clinical practice we observe that the severity of dystonia fluctuates and can be aggravated by non-specific stimuli such as emotion, stress and intentional movement.^{6,22} Test-retest reliability might therefore be limited and the usability of this type of measurement methods in intervention studies questionable. Furthermore, the situation during a clinical consultation or during measurements can cause stress or anxiety, eliciting an increased level of dystonia. As a consequence, the measured level of dystonia might not necessarily correspond to the overall normal daily situation. We do expect that dystonia decreases in the overall normal daily situation, causing attainment of treatment goals, but at this moment, with the current available outcome measures, we are not able to capture this change.

Clinical implications

Conditions for ITB treatment

Patients with dyskinetic CP receiving ITB are often severely affected. With a higher GMFCS level, accompanying impairments such as epilepsy, cognitive impairments, hearing and visual impairments, are more frequently present.^{1,3} This makes medical care for these patients complex. As a consequence of this complexity, a multi-disciplinary team consisting of a pediatric physiatrist, child neurologist, neurosurgeon, pediatrician, specialized nurse or physician assistant, occupational therapist, physiotherapist, speech therapist, social worker, and psychologist, is needed for selecting patients for ITB, evaluating ITB treatment effects, treating accompanying impairments and ITB emergency problem solving. Specialized members of the team must be available 24 hours a day, seven days a week, for trouble shooting in case of ITB pump emergencies such as sudden catheter dysfunction which can, without adequate care, result in life threatening withdrawal symptoms. As a result of the possible occurrence of lifethreatening complications, patients or caregivers must be able to comply to instructions in case of emergency and adhere to appointments for pump filling. Considering the low prevalence of patients on ITB it might take a long time to travel to a specialized center. When the need for pump filling is frequent, or when traveling is too time consuming and difficult for patients, it should be explored if pump fillings are possible in the home situation, as is the case in the Netherlands. Collaboration between enterprises who perform home-based ITB care and the responsible physicians in specialized clinics must be warranted.

Patient and treatment characteristics

ITB has been shown to be effective for attainment of individual treatment goals in patients with dyskinetic CP (GMFCS IV and V) and is therefore a treatment option in patients in whom oral medication is insufficient and who have goals that can be attained with ITB.

For the treatment of dystonia, baclofen is thought to act on the intracranial level.^{11,23} Clinical findings support this hypothesis: in contrast to spasticity, a single intrathecal lumbar bolus of baclofen mostly is not sufficient in decreasing dystonia.¹³ In line with what is described in literature, it usually takes several days of continuous intrathecal infusion to see improvement on dystonia and individual problems of daily life. Subsequently, for the treatment of dystonia in CP, catheters in our clinic are usually placed at the level of C4. Drawbacks of this high placement are assumed to be decreased head and trunk control and increased breathing disorders.²⁴⁻²⁶ In our study population however, we had only one patient on ITB complaining of decreased trunk control. Furthermore we did not see a difference in change of risk on sleep related breathing disorders while using ITB compared to placebo, showing that high catheter placement can safely be done.

A test period with ITB treatment via an external micro-infusion pump is conducted in many centers before definitive pump implantation. As described above, a single bolus administration of ITB is not sufficient in patients with dystonia.¹³ They require multiple days of continuous ITB infusion via an external lumbar catheter and an external micro-infusion pump. In our clinical experience, we encountered many side effects and major complications during these test periods, including CSF leaks, infection and meningitis (chapter 3). In these cases, ITB treatment had to be stopped prematurely, sometimes before any effect was noticeable. Often, adequate evaluation of the effect of ITB treatment could not be established. For parents and children, it was difficult to separate the effect of complications with the effect of the ITB, making the decision for pump implantation difficult. In the international consensus conference on the appropriate use of ITB, test treatment has been under discussion.⁹ It was decided that pump implantation without test treatment can be an option in teams with experience in ITB treatment. For these reasons we no longer include a test period for patients with dyskinetic CP and directly continue to pump implantation.

Dose finding can be troublesome for patients with dyskinetic CP. We developed a dosing schedule, based on clinical experience. We saw that many patients with dyskinetic CP do not respond enough or only temporary to the simple continuous mode in which they receive a stable dosage of baclofen during the day. Patients who do not respond sufficiently or who experience improvement with increments but fading of the effect after several days, are likely to benefit from bolus dosing 3 to 4 times per day and a lower basal rate in between boluses. We suggest using this dosing schedule as a guideline for finding the optimal dosage in patients with dyskinetic CP (schedule is presented in chapter 5, additional information).

Evaluation of ITB treatment

ITB in dyskinetic CP is aimed to decrease dystonia, but current measurement methods for dystonia severity (DIS, BADS) did not show change after three months of ITB treatment (IDYS trial chapter 5). Furthermore, dystonia fluctuates and the measured severity of dystonia does not necessarily correspond to the overall normal daily situation. Therefore, the severity of dystonia as measured with these scales cannot be used to guide optimal dose finding. As described, GAS is a meaningful measure to evaluate the effect of treatment for the individual patient. Attainment of goals, by interviewing patients and parents, can be used to guide optimal dose finding. For example, when a goal is to facilitate changing a diaper, parents can be questioned about the status of that specific goal. When the goal is attained, further increase might not be needed. However, when the goal is not yet attained, the dosage can be further increased in a stepwise manner, until it is.

During the time of writing this thesis, a new scale was developed, the Caregiver Priorities and Child Health Index of Life with Disabilities (CPCHILD[®]) Questionnaire.²⁷ This questionnaire assesses the health status, functional limitations, comfort, wellbeing and ease of caregiving of children with severe CP (GMFCS IV and V).^{27,28} Six domains are rated: personal care/activities of daily living (9 items), positioning, transferring and mobility (8 items), comfort and emotions (9 items), communication and social interaction (7 items), health (3 items), overall quality of life (1 item). Domain subscores can be evaluated and a total score can be calculated from these scores. The CPCHILD has an excellent test-retest reliability.²⁷ In 2015 the Dutch version (CPCHILD-DV) was found to have sufficient reliability and validity to be a representative outcome measure for health status and well-being of non-ambulatory children with CP.²⁸ A limitation of this scale is that scoring is done from a child-perspective and does not include the burden of care for parents and other caregivers. However, this questionnaire seems to be a useful instrument for children with severe CP, in addition to goal attainment scaling, to evaluate treatment effects.

Future studies

Reliable measurement of dystonia

Current clinical observation scales to rate dystonia are limited in reliability. More reliable measures are needed to measure changes in dystonia in patients on ITB. Future research into the assessment of dystonia should focus on finding reliable outcome measures that are feasible for clinical measurement of dystonia, both in the clinical setting and at home. Instrumented measures of dystonia are future perspectives, which might provide objective assessment during average daily situations. Options to be explored are, for example, automated markerless video tracking, and/or the use of body sensors such as accelerometers and 3D motion trackers with automated pattern recognition. Using these measurements, we can explore the use of machine learning to assess and distinguish dystonia.

Long term effects and characteristics influencing outcome

The IDYS trial evaluated the effect of ITB treatment in dyskinetic CP during the first 3 months of treatment. Several questions on the effect of ITB in dyskinetic CP remain to be answered.

Firstly, the long term effects have yet to be determined. Will the level of dystonia remain stable in the course of time? How often is adjustment of the dose of ITB needed? Will the effect of ITB remain present on the long term? What is the relation between the position of the spinal catheter and the effect of baclofen on dystonia and individual treatment goals?

The use of GAS for evaluation on the long term is questionable. Dystonia will not disappear with ITB. During growth and further into adulthood, it can be expected that other complaints or problems will become more evident for patients. Evaluation of problems from the past will be less relevant. Standardized measurements, sensitive to change, on the level of activities and participation and the degree of external support needed, seem more suitable for long term evaluation.

Secondly, we see that most patients benefit from ITB but not all patients attain their goals. It is important to find out which patients characteristics influence outcome. This knowledge can be used to improve patient selection and counselling.

In order to determine long term effects and which treatment and patient characteristics influence outcome, studies with a large sample size are needed. A randomized clinical trial is not a suitable design to study the effect of ITB over years. A prospective cohort study design is most suitable. National and international networks are important to obtain sufficient sample size. In the Netherlands, a national follow up and treatment register (Nederlands CP Register) has started, providing the opportunity to prospectively register outcomes of ITB on a national level. Another example is the Australian ITB Audit.²⁹ International harmonization of outcome measures is important to provide the opportunity to pool data of this relatively small patient group.

ITB IN PROGRESSIVE NEUROLOGICAL DISORDERS OF CHILDHOOD

Spasticity and dystonia can be symptoms in children with progressive neurological disorders (PND). ITB has been applied for treatment of severe spasticity or dystonia in patients with PND in order to maintain comfort and decrease the burden of caregiving but the effect and the level of evidence were not clear. In the systematic review presented in chapter 6, we found only six studies, which consisted of five case reports

and one small case series. Between publication of this paper and the present, two more papers (one case report, one case series) were published.^{30,31} This means that there is insufficient evidence for the effect of ITB in PND. Despite the low level of evidence of the available studies on the effect of ITB in PND, studies reported improvement of spasticity, spasms, pain, mobility and care giving (chapter 6).^{30,31} In the studies in chapter 7 and 8, we found that most patients are satisfied with ITB treatment. Unfulfilled goals and adverse treatment effects are reasons for being unsatisfied or only partly satisfied.

Comparable to severely affected CP patients, ITB treatment goals for PND patients are set on the different levels of the ICF-CY (chapter 6 and 7).³² On the level of body functions and structures, reduction of spasticity and pain are frequent goals. On the level of activities and participation, goals are mostly related to mobility (e.g. improvement of sitting and transfers). On the level of environmental factors, goals are related to facilitation of care giving. The nature of these goals reflects that most patients with PND receiving ITB are non-ambulatory.

ITB as part of palliative care

Although treatment goals are comparable to those of patients with CP receiving ITB, we questioned whether the effect would be similar. Since PND has a progressive course, we hypothesized that the reported effect would be less evident over time. Chapter 8 presents a cross-sectional questionnaire study. In this study we found that the effect of ITB on caregiving was comparable in spastic or dyskinetic CP and PND. Passive movements of the legs during care giving, transfers and sitting are thought to be facilitated by decreasing spasticity and/or dystonia, independent of the underlying condition, explaining the similar findings between groups.

The effect on comfort and mobility was significantly less favorable for the PND group compared to both CP groups. For CP, spasticity and/or dystonia are expected to be continuously present in more or less similar severity. Thereby the effectiveness of ITB is expected to be maintained on the long term.^{8,32} However, the clinical manifestation of PND can be complicated. For diseases such as metachromatic leukodystrophy, demyelination of the peripheral nervous system and the central nervous system can be present in varying ways leading to a clinical picture that changes between patients and within patients over time. Spasticity can be present at first and increase over time due to central nervous system involvement, but when peripheral neuropathy becomes more evident, spasticity decreases and may even disappear. Motor impairments, epilepsy, visual and cognitive impairments will worsen when disease progresses with death as a final consequence. Despite similarity in goals between CP and PND, ITB in PND is part

of palliative care, with the aim to maintain the best achievable level of comfort, delay progressing of contractures and (hereby) facilitate sitting, transfers and care giving.

Clinical implications and advice

Despite the complicated and varying clinical manifestation of PND, a common clinical finding is that discomfort due to pain and spasticity often increases shortly after wheelchair dependency occurs.^{30,31} Patients' general health status will also decrease from this moment on. It is advisable to consider ITB at this point in time as part of palliative care, since general health status is still sufficient to allow pump implantation and initiation of ITB will be beneficial for the patient.

Taking the decision for ITB is a difficult one, for patients, parents and caregivers. Decision making should therefore be done in a multidisciplinary team including a pediatric physiatrist, child neurologist, neurosurgeon, pediatrician, specialized nurse or physician assistant, occupational therapist, physiotherapist, speech and language therapist, social worker and psychologist, who have experience with ITB and PND. Follow up should also take place in a multidisciplinary setting since additional problems might occur for which further treatments and comprehensive aids might be needed.

Future studies

Prospective cohort studies are needed to establish the long-term effect of ITB in PND. International collaboration is necessary considering the very limited number of patients with PND who receive ITB. With larger number of patients, a distinction can be made between the effect of ITB in different PNDs. Large international registers using harmonized outcome measures are needed to assure sufficient and adequate data collection. Outcome measures should represent all levels of the ICF-CY. Hereby we can get more insight in the course of symptoms over time when disease progresses. Furthermore, we should gain insight in which patient characteristics, such as type of PND, age of onset, age of loss of mobility, and timing of initiating of ITB, influence outcome. This information can be used to improve patient selection and counselling.

REFERENCES

- Himmelmann K, Beckung E, Hagberg B, Uvebrant P. Gross and fine motor function and accompanying impairments in cerebral palsy. Dev Med Child Neurol 2006;48:417-23.
- Himmelmann K, Hagberg G, Wiklund LM, Eek MN, Uvebrant P. Dyskinetic cerebral palsy: a population-based study of children born between 1991 and 1998. Dev Med Child Neurol 2007;49:246-51.
- Himmelmann K, McManus V, Hagberg G, et al. Dyskinetic cerebral palsy in Europe: trends in prevalence and severity. Arch Dis Child 2009;94:921-6.
- AACPDM Dystonia Care Pathway. 2018. (Accessed July 27th 2018, at https://www. aacpdm.org/publications/care-pathways/ dystonia.)
- Fehlings D, Brown L, Harvey A, et al. Pharmacological and neurosurgical interventions for managing dystonia in cerebral palsy: a systematic review. Dev Med Child Neurol 2018;epub before print.
- Monbaliu E, Himmelman K, Lin JP, et al. Clinical presentation and management of dyskinetic cerebral palsy. Lancet Neurol 2017;16:741-9.
- Masson R, Pagliano E, Baranello G. Efficacy of oral pharmacological treatments in dyskinetic cerebral palsy: a systematic review. Dev Med Child Neurol 2017;59:1237–48.
- Buizer AI, Martens BHM, Grandbois van Ravenhorst C, Schoonmade LJ, Becher JG, Vermeulen RJ. Effect of continuous intrathecal baclofen therapy in children: a systematic review. Dev Med Child Neurol 2018;epub ahead of print.
- Dan B, Motta F, Vles JSH, et al. Consensus on the appropriate use of intrathecal baclofen (ITB) therapy in paediatric spasticity. Eur J Paediatr Neurol 2010;14:19-28.

- Eek MN, Olsson K, Lindh K, et al. Intrathecal baclofen in dyskinetic cerebral palsy: effects on function and activity. Dev Med Child Neurol 2018;60:94-9.
- Albright AL, Barry MJ, Shagron DH, Ferson SS. Intrathecal baclofen for generalized dystonia. Dev Med Child Neurol 2001;43:652-7.
- 12. Motta F, Stignani C, Antonello CE. Effects of intrathecal baclofen on dystonia in children with cerebral palsy and the use of functional scales. Journal of Pediatric Orthopaedics 2008;28:213-7.
- Albright AL, Barry MJ, Fasick P, Barron W, Schulz B. Continous intrathecal baclofen infusion for symptomatic generalized dystonia. Neurosurg 1996;38:934-9.
- 14. Monbaliu E, Ortibus E, de Cat J, et al. The dyskinesia Impairment Scale: a new instrument to measure dystonia and choreoathetosis in dyskinetic cerebral palsy. Dev Med Child Neurol 2012;54:278-83.
- 15. Monbaliu E, Ortibus E, Roelens f, et al. Rating scales for dystonia in cerebral palsy: reliability and validity. Dev Med Child Neurol 2010;52:570-5.
- Barry MJ, VanSwearingen JM, Albright AL. Reliability and responsiveness of the Barry-Albright Dystonia Scale. Dev Med Child Neurol 1999;41:404-11.
- Hoving MA, Van Raak EPM, Spincemaille GHJJ, et al. Safety and one-year efficacy of intrathecal baclofen therapy in children with intractable spastic cerebral palsy. Eur J Paediatr Neurol 2009;13:247-56.
- Steenbeek D, Ketelaar M, Galama K, Gorter JW. Goal attainment scaling in paediatric rehabilitation: a critical review of the literature. Dev Med Child Neurol 2007;49:550-6.

- Turner-Stokes L. Goal attainment scaling (GAS) in rehabilitation: a practical guide. Clinical Rehabilitation 2009;23:362-70.
- Urach S, Gaasterland C, Posch M, et al. Statistical analysis of Goal Attainment Scaling endpoints in randomised trials. Stat Methods Med Res 2018:962280218777896.
- Monbaliu E, De Cock P, Mailleux L, Dan B, Feys H. The relationship of dystonia and choreoathetosis with activity, participation and quality of life in children and youth with dyskinetic cerebral palsy. European Journal of Paediatric Neurology 2017;21:327-35.
- 22. Monbaliu E, de Cock P, Ortibus E, Heyrman L, Klingels K, Feys H. Clinical patterns of dystonia and choreoathetosis in participants with dyskinetic cerebral palsy. Dev Med Child Neurol 2016;58:138-44.
- Albright AL. Intrathecal baclofen in cerebral palsy movement disorders. Journal of Child Neurology 1996;11:S29-S35.
- Sandella DE, O'Brien LM, Shank LK, Warschausky SA. Sleep and quality of life in children with cerebral palsy. Sleep Medicine 2011;12:252-6.
- Bensmail D, Marquer A, Roche N, Godard A, Lofaso F, Quera-Salva M. Pilot Study Assessing the Impact of Intrathecal Baclofen Administration Mode on Sleep-Related Respiratory Parameters. Arch Phys Med Rehabil 2012;93:96-9.
- 26. Bensmail D, Quera-Salva M, Roche N, et al. Effect of intrathecal baclofen on sleep and respiratory function in patients with spasticity. Neurology 2006;67:1432-6.
- 27. Narayanan UG, Fehlings D, Weir S, Knights S, Kiran S, Campbell K. Initial development and validation of the Caregiver Priorities and Child Health Index of Life with Disabilities (CPCHILD). Dev Med Child Neurol 2006;48:804-12.

- Zalmstra TAL, Elema A, Boonstra AM, et al. Validation of the Caregiver Priorities and Child Helath Index of Life with Disabilities (CPCHILD) in a sample of Dutch nonambulatory children with cerebral palsy. Disability and Rehabilitation 2015;37:411-6.
- 29. Stewart K, Hutana G, Kentish M. Intrathecal baclofen therapy in paediatrics: a study protocol for an Australian multicentre, 10-year prospective audit. BMJ Open 2017;7:e015863.
- Hjartarson HT, ehrstedt C, Tedroff K. Intrathecal baclofen treatment an option in X-linked adrenoleukodystrophy. Eur J Paediatr Neurol 2018;22:178-81.
- 31. van der Veldt N, van Rappard DF, van de Pol LA, et al. Intrathecal baclofen in metachromatic leukodystrophy. Dev Med Child Neurol 2019;61:232-5.
- 32. Vles GF, Soudant D, Hoving MA, et al. Long-term follow-up of continous intrathecal baclofen therapy in nonambulant children with intractable spastic cerebral palsy. Eur J Paediatr Neurol 2013;17:639-44.


SUMMARY

Movement disorders in childhood are mostly caused by dysfunction of the developing brain due to brain lesions or brain abnormalities. The International Classification of Functioning, disability and health for Children and Youth (ICF-CY) can be used as a framework to categorize personalized treatment goals. The ICF-CY shows that impairments on the level of body functions and structures (e.g. spasticity or dyskinesia) can lead to problems in activities of daily life and participation in society such as mobility, self-care, communication and learning. Activities and participation are furthermore influenced by environmental factors including the availability of assistance for personal care and aids for mobility.

The spectrum of the severity of cerebral movement disorders is broad. This thesis will focus on severely affected children: children with cerebral palsy (CP) classified with the Gross Motor Functioning Classification System (GMFCS) in levels IV and V and children with progressive neurological disorders (PND) who are equally affected. These children are not able to walk unassisted or do not walk at all, and mainly use a (powered) wheelchair for mobility.

The most common cause of cerebral movement disorders and physical disability in childhood is CP. Spastic and dyskinetic CP are the two most common movement disorders (72-91% and approximately 15% of CP respectively). Spastic and dyskinetic movement disorders can also be caused by PND. There are many different diagnoses related to PND in childhood, all of them rare.

When spasticity or dystonia is severe and interferes with comfort, quality of life or activities of daily life, the first step is oral pharmacological treatment. When this is insufficient, the next steps are advanced treatment options. One of these options is intrathecal baclofen (ITB). With ITB, baclofen is administered intrathecally using an implanted micro-infusion pump (Medtronic Synchromed II). The pump is implanted subcutaneously, mostly in the left lower abdomen. A catheter connects the pump with the intrathecal space.

There is some evidence for the short-term effectiveness of ITB for treatment of spasticity in children with spastic CP, provided by single bolus randomized trials. For the effect of ITB in children with dyskinetic CP, the level of evidence is low. For PND, it is unclear what the level of evidence for the effect of ITB is.

The first aim of this thesis was to investigate the effect of ITB in the treatment of dyskinetic cerebral palsy (CP). The focus was primarily on the effect on individual treatment goals, mostly on the levels of activities and participation, and environmental factors. The secondary focus was on the level of body functions and structures (dystonia, choreoathetosis, spasticity, pain, comfort). The second aim was to describe the effect and the current level of evidence of ITB treatment in PND of childhood on all ICF-CY levels.

A narrative review on dyskinetic CP is provided in **chapter 2**. In dyskinetic CP, dystonia and choreoathetosis are often present simultaneously, with dystonia being the dominant feature. Most treatment options focus on the treatment of dystonia, little is reported for the treatment of choreoathetosis. The use of pharmacological treatment in both dystonia and choreoathetosis is not supported by scientific evidence. Neuromodulation interventions such as ITB and deep brain stimulation (DBS) are advanced treatment options. ITB is used to decrease pain, improve comfort, prevent deformities and ease care giving. ITB decreases dystonia but does not lead to changes in functional independence of daily activities. Dystonia shows variable responsiveness with DBS for secondary dystonia (where there is structural damage to the basal ganglia, as in dyskinetic CP), and despite the reported decrease in some patients, the effects on functionality and quality of life are not clear. Multicenter studies are necessary to provide more evidence for the effect for both neuromodulation treatment options.

Chapter 3 describes the results of a pilot study, looking into the effects of ITB in dyskinetic CP, in four patients admitted for ITB test treatment via an external spinal catheter. They received either intrathecal baclofen or intrathecal placebo for four consecutive days after completion of the regular test treatment period. Individual problems of daily life were scored on a visual analogue scale (VAS) from 0 (no problems) to 10 (impossible to do) on three time points: before treatment, during ITB test treatment and during blinded treatment (ITB or placebo). Secondary outcome measures were dystonia (Barry-Albright-Dystonia Scale (BADS)), pain (VAS), and comfort (VAS). The clinically significant difference was determined. Both problems of daily life and dystonia scores improved in all patients during ITB test treatment. Pain and discomfort improved for two patients and worsened in one. This pilot was hampered by several serious complications making reliable assessment during the blinded phase difficult. Despite this limitation and the limited level of evidence of this study, the results of the pilot were promise enough to continue with a clinical trial.

The Intrathecal baclofen in Dyskinetic cerebral palsy (IDYS) trial is a multi-centre, randomised, double-blind, placebo-controlled trial with the aim to provide evidence for the effect of ITB in dyskinetic CP. The study protocol is presented in **chapter 4**. Patients with dyskinetic CP, GMFCS IV and V, who are eligible for ITB treatment were included. They were assigned by blocked randomization (2:2), to receive either placebo or ITB for 3 months via an implanted micro-infusion pump. After three months, outcome measures were assessed and all patients continued on ITB thereafter. The primary outcome measurement was attainment of individual treatment goals using Goal Attainment Scaling (GAS). Secondary outcome measures included dystonia (BADS and Dyskinesia Impairment Scale (DIS)), spasticity (spasticity test (SPAT) and Hoffmann reflex (h-reflex)), range of motion (ROM), pain (VAS), comfort (VAS) and the change in risk of sleep-related breathing disorders (questionnaire). Adverse events were monitored. Power calculations were done and a total of 13 patients per group would provide enough power.

The results of the IDYS trial are described in **chapter 5**. To prevent the study from becoming underpowered due to complications, 18 patients per group were included. Data for final GAS analysis were available for 16 patients in the placebo group and 17 in the ITB group. The mean GAS T-scores at three months were significantly more favourable for ITB, compared to placebo. The DIS dystonia subscore and dystonia rest subscore showed a significant difference in favour of ITB. This difference was caused by an increase of dystonia in the placebo group, compared to no change in the ITB group. On the other secondary outcome measures, no difference between placebo and ITB was found. Number and types of (serious) adverse events were similar between groups. With the IDYS trial, we provide high level of evidence for the effect of ITB on the attainment of individual treatment goals. Current clinical observation scales to rate dystonia are limited in their use due to unknown test-retest reliability. More reliable measures are needed to measure changes in dystonia in patients on ITB.

Chapter 6 presents the results of a systematic literature search on the effect of ITB in patients with PND in childhood. A total of six studies were identified. Five studies were case reports and one study a case series, and therefore of low level of evidence. Outcome measures used were mainly on the level of body functions and structures. On this level, spasticity was objectively measured in three studies and decreased in all during ITB treatment. Furthermore, pain was reported to decrease. On the level of activities and participation no structural outcome measures were used. Subjective improvement of dressing, hygienic care and positioning in a wheelchair were described.

One of the studies included in the review of chapter 2 reported satisfaction on ITB in six patients with PND. This study is described in **chapter 7**. The mean follow up time of these patients was 3.3 years (SD 2.9). Parents were asked if they were satisfied with ITB treatment (yes/no/partially) and asked to score satisfaction on a Visual Analogue Scale (VAS) ranging from 0 (poorest score) to 10 (most optimal score). Four parents reported satisfaction with ITB treatment, one was partially satisfied and one was not satisfied. The mean satisfaction score (VAS) was 7.5 (SD 1.6, range 6–10). Since parents in this study reported moderate to good satisfaction, we conclude that ITB could be a valuable treatment option in patients with PND.

Intrathecal baclofen is used for treatment in spastic CP, dyskinetic CP and PND. CP and PND share symptoms, but the etiology and the course of these disorders is very different. In **chapter 8** the effect of ITB on the domains of mobility, personal care, comfort and communication was assessed. Satisfaction was scored (VAS). Results were compared between groups. Caregivers of 68 patients completed the questionnaire. Thirty-nine patients were diagnosed with spastic CP, 13 with dyskinetic CP and 16 with PND. The PND group had the shortest follow-up time. They scored significantly less favorably for the effect on mobility and comfort. The positive effect on personal care and communication was similar in all groups. Expectations were met in approximately 80% of the patients in all groups. Satisfaction scores were similar between groups as well. Given the progressive nature of PND and the only short follow up period in this study compared to the CP groups, it is interesting to see if the positive effects of ITB found in PND remain present after a longer follow up period.

Chapter 9 presents a critical appraisal of the findings of the studies described in this thesis. ITB seemed promising for treatment of dyskinetic CP but the level of evidence was low. With the results of the IDYS trial we provided high level of evidence for the effectiveness of ITB for attainment of individual treatment goals. Measurement of dyskinesia is troublesome with the current available outcome measures and should not be primarily used for evaluation. Attainment of treatment goals provides more reliable information about the actual treatment effect. Other options for measuring the severity of dystonia should be explored in the future. Furthermore, future research assessing long term effects and the influence of treatment and patient characteristics is needed. An (inter)national register such as the Nederlands CP register (Dutch CP register), will provide a good basis for a prospective longitudinal cohort study, ensuring sufficient patient numbers and harmonisation of outcome measures.

For patients with PND, the level of evidence for the effect on ITB is very low. However, most patients seem to benefit from ITB and ITB should be considered as part of palliative care. International trials with a longer follow up period and larger numbers of patients are needed to provide information about, amongst others, adequate selection of patients and appropriate timing of initiation of ITB.

SAMENVATTING

Cerebrale bewegingsstoornissen omvatten een breed spectrum aan uittingsvormen waarbij ook de ernst erg uiteenlopend is. Dit proefschrift richt zich op ernstig aangedane kinderen. Dit zijn kinderen met cerebrale parese (CP) die geclassificeerd worden op de Gross Motor Functioning Classification System (GMFCS) in niveau IV en V, of kinderen met progressieve neurologische aandoeningen (PND) die net zo ernstig aangedaan zijn in hun motorisch functioneren. Deze kinderen kunen niet zelfstandig lopen, of kunnen helemaal niet lopen, en gebruiken voornamelijk een (elektrische) rolstoel voor mobiliteit.

Bewegingsstoornissen bij kinderen worden voornamelijk veroorzaakt door dysfunctie van het in ontwikkeling zijnde brein door beschadigingen of aanlegstoornissen. De International Classification of Functioning, disability and health for Children and Youth (ICF-CY) kan gebruikt worden als kader om de complexiteit van problemen die optreden bij cerebrale bewegingsstoornissen te omvatten. De ICF-CY maakt verbinding tussen het niveau van lichaamsfuncties en structuren (bijv. spasticiteit of dyskinesie), welke kunnen leiden tot problemen op het gebied van activiteiten in het dagelijks leven of participatie in de maatschappij zoals mobiliteit, zelfverzorging, communicatie en leervaardigheid. Activiteiten en participatie kunen daarnaast nog worden beïnvloed door omgevingsfactoren, waaronder de beschikbaarheid van hulp bij persoonlijke verzorging en hulpmiddelen voor mobiliteit.

De meest voorkomende oorzaak van cerebrale bewegingsstoornissen en lichamelijke beperkingen in de kindertijd is CP. Spastische en dyskinetische CP zijn de twee meest voorkomende bewegingsstoornissen (respectief 72-91% en ongeveer 15% van CP). Spastische en dyskinetische bewegingsstoornissen kunen daarnaast ook veroorzaakt worden door PND. De meeste onderliggende diagnosen van PND zijn zeldzaam.

Wanneer spasticiteit of dystonie ernstig zijn en een negatieve invloed hebben op comfort, kwaliteit van leven of activiteiten in het dagelijks leven, zal als eerste stap orale farmacologische behandeling overwogen worden. Als dit onvoldoende werkt, zal de volgende stap invasieve behandelopties omvatten. Eén van deze opties is intrathecale baclofen (ITB). Met deze behandeling wordt baclofen intrathecaal afgegeven door middel van een geïmplanteerde micro-infusie pomp (Medtronic Synchromed II). De pomp wordt subcutaan geplaatst, meestal in het linker onder kwadrant van het abdomen. Een katheter verbindt de pomp met de intrathecale ruimte waar de baclofen wordt afgegeven. Er is enig bewijs voor het korte termijn effect van ITB in de behandeling van spasticiteit bij kinderen met spastische CP. Dit bewijs wordt geleverd door gerandomiseerde trials waarbij intrathecale baclofen middels eenmalige bolus wordt toegediend. Voor het effect van ITB bij kinderen met dyskinetische CP, is de bewijskracht laag. Voor PND is het onduidelijk wat het effect is, en wat de bewijskracht voor het effect is.

Het eerste doel van deze thesis was om te onderzoeken wat het effect is van ITB bij patiënten met dyskinetische CP. De primaire focus lag hierbij op het effect op individuele behandeldoelen, in het bijzonder op het niveau van activiteiten en participatie, en omgevingsfactoren. Secundair lag de focus op het niveau van lichaamsfuncties en structuren (dystonie, choreoathetose, spasticiteit, pijn, comfort). Het tweede doel van deze thesis was om het effect en de huidige bewijskracht te beschrijven voor behandeling met ITB bij PND op alle niveaus van de ICF-CY.

Hoofdstuk 2 bevat een beschrijvend review over dyskinetische CP. Bij dyskinetische CP zijn dystonie en choreoathetose vaak simultaan aanwezig. Dystonie is hierin meestal dominant. De meeste behandelopties richten zich op de behandeling van dystonie. Er is slechts weinig bekend over de behandeling van choreoathetose. Het wetenschappelijk bewijs voor het gebruik van orale farmacologische middelen voor zowel dystonie als choreoathetose is zwak. Neuromodulatie met bijvoorbeeld ITB of deep brain stimulation (DBS) zijn een volgende stap in de behandeling. ITB wordt gebruikt om pijn te verminderen, comfort te verbeteren, deformiteiten te voorkomen en de verzorging te vergemakkelijken. Dystonie vermindert door ITB, maar dit leidt niet tot veranderingen in de onafhankelijkheid bij het uitvoeren van dagelijkse activiteiten. Het effect van DBS op dystonie is variabel en meestal slechts zeer beperkt bij patiënten met secundaire dystonie (dystonie veroorzaakt door hersenenschade zoals bij dyskinetische CP). Ondanks dat dystonie bij sommige patiënten verbetert, is het niet duidelijk wat het effect op het functioneren en de kwaliteit van leven is. Multicenter studies zijn nodig om meer bewijs te krijgen voor beide neuromodulatie-opties.

Hoofstuk 3 beschrijft de resultaten van een placebo gecontroleerde pilotstudie naar de effecten van ITB bij vier patiënten met dyskinetische CP. Dit werd gedaan tijdens een ITB proefbehandeling via een externe lumbale katheter. Na het afronden van de reguliere proefbehandeling, ontvingen zij gerandomiseerd en geblindeerd intrathecale baclofen of intrathecale placebo gedurende vier opeenvolgende dagen. Individuele problemen van het dagelijks leven werden gescoord op een visual analogue scale (VAS) van 0 (geen problemen) tot 10 (onmogelijk om te doen). Het scoren werd op drie tijdpunten gedaan: voor het starten van de behandeling, tijdens de proefbehandeling en tijdens geblindeerde behandeling met ITB of placebo. Secundaire uitkomstmaten waren dystonie (Barry-Albright-Dystonia Scale (BADS)), pijn (VAS), en comfort (VAS). Het klinisch significante verschil werd beoordeeld. Zowel problemen in het dagelijks leven als dystonie verbeterden bij alle patiënten tijdens de proefbehandeling. Pijn en comfort verbeterden bij twee patiënten en verslechterde bij één. Tijdens de pilot waren er enkele serieuze complicaties, waardoor betrouwbare beoordeling tijdens de geblindeerde fase moeilijk was. Ondanks deze beperking en de beperkte bewijskracht van deze studie, waren de resultaten veelbelovend genoeg om een klinische trial te starten.

De intrathecale baclofen bij dyskinetische cerebrale parese (IDYS) trial, is een multicenter, gerandomiseerde, dubbel geblindeerde en placebo gecontroleerde trial met het doel bewijs te leveren voor het effect van ITB bij dyskinetische CP. Het studieprotocol staat beschreven in **hoofdstuk 4**. Patiënten met dyskinetische CP, GMFCS IV en V, die in aanmerking komen voor ITB werden geïncludeerd. Nadat er baseline metingen gedaan waren, werden patiënten gerandomiseerd middels blok randomisatie (2:2) om gedurende drie maanden placebo of ITB te krijgen via een geïmplanteerde micro-infusie pomp. Na drie maanden, werden wederom metingen gedaan. De primaire uitkomstmaat was het behalen van individuele behandeldoelen waarvoor gebruikt gemaakt is van Goal Attainment Scaling (GAS). Secundaire uitkomstmaten waren: dystonie (BADS en Dyskinesia Impairment Scale (DIS)), spasticiteit (spasticiteit test (SPAT) en Hoffmann reflex (H-reflex)), range of motion (ROM), pijn (VAS), comfort (VAS) en de verandering van het risico op slaap gerelateerde ademhalingsproblemen (middels een vragenlijst). Bijwerkingen en complicaties werden gemonitord. Power berekeningen zijn gedaan en er waren 13 patiënten per groep nodig om genoeg power te verkrijgen.

De resultaten van de IDYS trial worden beschreven in **hoofdstuk 5**. Op basis van de power-analyse en rekening houdend met mogelijke uitval werden er 18 patiënten per groep geïncludeerd. Data voor GAS-analyse was beschikbaar voor 16 patiënten in de placebo groep en 17 in de ITB groep. De gemiddelde GAS-score na drie maanden was significant beter voor ITB in vergelijking met placebo. De DIS dystonie subscore en dystonie rust subscore lieten een significant verschil zijn, ten gunste voor ITB. Het verschil werd veroorzaakt door een toename van dystonie in de placebo groep, in vergelijking met een onveranderde score in de ITB groep. Voor de andere secundaire uitkomstmaten, werd geen verschil tussen de groepen gevonden. Bijwerkingen en complicaties, zowel type als aantal, waren gelijk tussen de groepen. Met de IDYS trial, leveren we een hoog niveau van bewijskracht voor het effect van ITB op het behalen van individuele behandeldoelen bij patiënten met dyskinetische CP. Huidige klinische observatie schalen voor het scoren van dystonie zijn mogelijk beperkt in het gebruik door de nog onbekende test-hertest betrouwbaarheid. Er is behoefte aan betrouwbaardere uitkomstmaten om veranderingen in dystonie bij patiënten met ITB te meten.

Hoofdstuk 6 geeft de resultaten weer van een systematische literatuurstudie naar het effect van ITB bij patiënten met PND in de kindertijd. In totaal zijn zes studies gevonden. Vijf waren case reports en één was een case serie. De bewijskracht is derhalve laag. Uitkomstmaten waren voornamelijk op het niveau van lichaamsfuncties en structuren. Spasticiteit werd in drie studies objectief gemeten en verbeterde in al deze studies tijdens ITB. Daarnaast werd gerapporteerd dat pijn verminderd. Op het niveau van activiteiten en participatie werden geen structurele uitkomstmaten gebruikt. Subjectieve verbetering van kleden, hygiënische verzorging en positioneren in een rolstoel werd beschreven.

Eén van de studies geïncludeerd in het review in hoofdstuk 2 rapporteert de tevredenheid over ITB bij zes patiënten met PND. Dit onderzoek staat beschreven in **hoofdstuk 7**. De gemiddelde follow up tijd van deze patiënten was 3.3 jaar (SD 2.9). Ouders werd gevraagd of zij tevreden waren met ITB (ja/nee/deels) en of ze een rapportcijfer (VAS) konden geven voor hun tevredenheid van 0 (slechtste score) tot 10 (beste score). Vier ouders gaven aan tevreden te zijn met ITB, een was deels tevreden en een was niet tevreden. De gemiddelde tevredenheidsscore (VAS) was 7.5 (SD 1.6, range 6-10). Aangezien ouders matige tot goede tevredenheid rapporteerden, concluderen we dat ITB waardevol kan zijn in de behandeling van patiënten met PND.

Intrathecale baclofen wordt gebruikt voor de behandeling van spastische CP, dyskinetische CP en PND. CP en PND hebben overeenkomstige symptomen, maar de etiologie en het ziektebeloop is erg verschillend. In **hoofdstuk 8** werd het verschil tussen deze groepen voor het effect van ITB op verschillende domeinen (mobiliteit, persoonlijke verzorging, comfort en communicatie) beoordeeld. Tevredenheid werd gescoord (VAS). Verzorgers van 68 patiënten vulden de vragenlijst in. Er waren 39 patiënten met spastische CP, 13 met dyskinetische CP en 16 met PND. De PND groep had de kortste follow-up tijd. De PND groep scoorde significant slechter voor het effect op mobiliteit en comfort. Het effect op persoonlijke verzorging en communicatie was gelijk voor alle groepen. Bij ongeveer 80% voldeed de behandeling aan de verwachtingen. Tevredenheidsscores waren gelijk tussen de groepen. Gezien de progressieve karakter van PND en de korte follow-up tijd voor deze groep in vergelijking met de CP groepen,

is het interessant om te zien op welke manier het effect aanwezig blijft naarmate de follow up langer is.

Hoofdstuk 9 bevat een kritische beschouwing van de bevindingen van de studies in dit proefschrift. ITB leek veelbelovend voor de behandeling van dyskinetische CP maar de bewijskracht was laag. Met de resultaten van de IDYS trial leveren we een hoog niveau van bewijskracht voor de effectiviteit van ITB op het behalen van individuele behandeldoelen. Het meten van dyskinesie met de huidig beschikbare methoden is lastig en zou daarom niet primair gebruikt moeten worden voor evaluatie van behandeling. Het behalen van individuele behandeldoelen geeft meer betrouwbare informatie over het werkelijke behandeleffect. Andere opties voor het meten van de ernst van dystonie moeten in de toekomst onderzocht worden. Daarnaast zijn studies nodig die het lange termijn effect en de invloed van behandel en patiënt karakteristieken beoordelen. Een (inter)nationaal register, zoals het Nederlands CP register, zorgt voor een goede basis voor een prospectief, longitudinaal cohort onderzoek waarin voldoende patiënt aantallen en harmonisatie van uitkomstmaten gewaarborgd zijn.

Voor patiënten met PND is de bewijskracht voor het effect van ITB laag. Echter lijken de meeste patiënten wel profijt te hebben van ITB. Op grond van deze bevindingen zijn wij van mening dat ITB overwogen moet worden als deel van het palliatieve beleid. Internationale trials met een langere follow up periode en grote patiënt aantallen zijn nodig om informatie te verschaffen over onder andere adequate selectie van patiënten en het geschikte moment van starten van ITB.

CHAPTER 11

Appendix I. Dyskinesia Measurement Scales

Appendix II. IDYS Study Group



Appendix I. Dyskinesia Measurement Scales

A. The Barry-Albright_Dystonia Scale (BADS)

Directions Assess the patient for dystonia in each of the following regions: eyes, mou Write the scores on the lines provided. Rate severity based only on dyston functional limitations, do not score as dystonia-induced functional limitati deficits, persistent primitive reflexes, and/or other movement disorders a Definitions of movement disorders: Dystonia – sustained muscle contractions caused by twisting and repetitiv Spasticity – velocity-dependent resistance to passive movement Athetosis – distal writhing or contorting movements Chorea – brief, rapid, unsustained, irregular movements	th, neck, trunk, and each upper and lower extremity (eight body regions). is as evidenced by abnormal movements or postures. When assessing on if other factors such as weakness, lack of motor control, cognitive recontributing to functional limitation. e movements or abnormal postures
Ataxia - incoordination of movement characterized by wide based unstead	dy gait and falling movements
Eyes Signs of dystonia of the eyes include prolonged eyelid spasms and/or forced eye deviations: 0 - Absent 1 - Slight: dystonia less than 10% of the time and does not interfere with tracking 2 - Mild: frequent blinking, without prolonged spasms of eyelid closure and/or eye movements less than 50% of the time 3 - Moderate: prolonged spasms of eyelid closure, but eyes open most of the time and/or eye movements more than 50% of the time that interfere with tracking, but able to resume tracking 4 - Severe: prolonged spasms of eyelid closure, with eyelids closed at least 30% of the time, and/or eye movements more than 50% of the	Trunk: Signs of dystonia of the trunk include pulling of the trunk into any plane of motion – extension, flexion, lateral flexion or rotation: 0 – Absent 1 – Slight: pulling less than 10% of the time and does not interfere with lying, sitting, standing and/or walking 2 – Mild: pulling less than 50% of the time and does not interfere with lying, sitting, standing, and/or walking 3 – Moderate: pulling more than 50% of the time and/or dystonia that interferes with lying, sitting, standing, and/or walking 4 – Severe: pulling more than 50% of the time and dystonia that prevents sitting in a standard wheelchair (e.g. requires adapted seating system), standing, and/or walking
time that prevent tracking	*Unable to assess trunk movements.
*Unable to assess eye movements.	Trunk:
Eyes:	Upper extremities:
Mouth Signs of dystonia of the mouth include grimacing, clenched or deviated jaw, forced open mouth, and/or forceful tongue thrusting: 0 – Absent 1 – Slight: dystonia less than 10% of the time and does not interfere with speech, and/or feeding 2 – Mild: dystonia less than 50% of the time and does not interfere with speechand/or feeding 3 – Moderate: dystonia more than 50% of the time and/or dystonia that interferes with speech and/or feeding 4 – Severe: dystonia more than 50% of the time and/or dystonia that prevents speech and/or feeding *Unable to assess mouth movements. Mouth:	 Upper extremities: Signs of dystonia of the upper extremities include sustained muscle contractions causing abnormal postures: O - Absent 1 - Slight: dystonia less than 10% of the time and does not interfere with normal positioning and/or functional activities 2 - Mild: dystonia less than 50% of the time and does not interfere with normal positioning and/or functional activities 3 - Moderate: dystonia more than 50% of the time and/or dystonia that interferes with normal positioning and/or upper extremity function 4 - Severe: dystonia more than 50% of the time and/or dystonia that prevents normal positioning and/or upper extremity function (e.g. arms restrained to prevent injury) *Unable to assess upper extremity movements. Left upper extremity:
Neck:	Right upper extremity:
Signs of dystonia of the neck include pulling of the neck into any plane of motion – extension, flexion, lateral flexion, or rotation: 0 – Absent 1 – Slight: pulling less than 10% of the time and does not interfere with lying, sitting, standing, and/or walking 2 – Mild: pulling less than 50% of the time and does not interfere with lying, sitting, standing, and/or walking 3 – Moderate: pulling more than 50% of the time and/or dystonia that interferes with lying, sitting, standing, and/or walking 4 – Severe: pulling more than 50% of the time and dystonia that prevents sitting in a standard wheelchair (e.g. requiresspecial head rest), standing, and/or walking *Unable to assess neck movements. Neck:	Lower extremities Signs of dystonia of the upper extremities include sustained muscle contractions causing abnormal postures 0 – Absent 1 – Slight: dystonia less than 10% of the time and does not interfere with normal positioning and/or functional activities 2 – Mild: dystonia less than 50% of the time and does not interfere with normal positioning and/or functional activities 3 – Moderate: dystonia more than 50% of the time and/or dystonia that interferes with normal positioning and/or lower extremity weight bearing and/or function 4 – Severe: dystonia more than 50% of the time and/or dystonia that prevent normal positioning and/or lower extremity weight bearing and/or function *Unable to assess lower extremity movements Left lower extremity: Right lower extremity:
	L
TOTAL SCORE:	

B. The Burke-Fahn_Marsden Dystonia Rating scale (BFMDRS)

Region Eyes	Provoking factor	Severity factor			Weight		Region Score
	0-4	×	0-4	×	0,5	=	0-8
Mouth	0-4	x	0-4	x	0,5	=	0-8
Speech/ swallow	0-4	х	0-4	×	1,0	=	0-16
Neck	0-4	х	0-4	x	0,5	=	0-8
R arm	0-4	×	0-4	×	1,0	=	0-16
Larm	0-4	×	0-4	×	1,0	=	0-16
Trunk	0-4	×	0-4	×	1,0	=	0-16
R leg	0-4	×	0-4	×	1,0	=	0-16
L leg	0-4	×	0-4	×	1,0	=	0-16
					Sum	=	0-120

I. Provoking factor

A. General

- 0 No dystonia at rest or with action
- 1 Dystonia only with particular action
- 2 Dystonia with many actions
- 3 Dystonia on action of distant part of body or intermittently at rest
- 4 Dystonia present at rest

II. Severity factors

Eyes

- 0 No dystonia 1 – Slight: occasional blinking
- 2 Mild: frequent blinking without prolonged spasms of eye closure
- 3 Moderate: prolonged spasms of eyelid closure, but eyes open
- mostofthetime
- 4 Severe: prolonged spasms of eyelid closure, with eyes closed at least 30% of the time

Mouth

- 0 No dystonia present
- 1 Slight: occasional grimacing or other mouth movements (e.g. jaw opened or clenched; tongue movement)
- 2 Mild: movement present less than 50% of the time
- 3 Moderate dystonic movements or contractions present most of
- the time 4 – Severe dystonic movements or contractions present most of the time

Speech and swallowing

0-Normal

- 1 Slightly involved: speech easily understood or occasional choking
- 2 Some difficulty in understanding speech or frequent choking
- 3 Marked difficulty in understanding speech or inability to swallow firm foods
- 4 Complete or almost complete anarthria, or marked difficulty swallowing soft foods and liquids

B. Speech and swallowing

- 0 No dystonia
- 1-Occasional, either, or both
- 2- Frequent either 3- Frequent one and occasional other
- 4- Frequent both
- , inside and only

Neck

- 0 No dystonia present
- 1 Slight: occasional pulling
- 2 Obvious torticollis, but mild
- 3 Moderate pulling
- 4 Extreme pulling

Arm

- 0 No dystonia present
- 1-Slight dystonia: clinically insignificant
- 2 Mild: obvious dystonia, but not disabling
- 3 Moderate: able to grasp, with some manual function
- 4 Severe: no useful grasp

Trunk

- 0 No dystonia present
- 1-Slight bending: clinically insignificant
- 2 Definite bending: but not interfering with standing or walking
- 3 Moderate bending: interfering with standing or walking
- 4 Extreme bending of trunk preventing standing or walking

Leg

- 0 No dystonia present
- 1 Slight dystonia, but not causing impairment: clinically insignificant
- 2 Mild dystonia: walks briskly and unaided
- 3 Moderate dystonia: severely impairs walking or requires
- assistance
- 4 Severe: unable to stand or walk on involved leg

C. The Dyskinesia Impairment Scale (DIS)



A standardized video protocol is used in which activities and postures are described.

Appendix II. IDYS study group

Department of Rehabilitation Medicine

L.A. (Laura) Bonouvrié, J.G. (Jules) Becher, K. (Karin) Boeschoten, A.I. (Annemieke) Buizer, V. (Vincent) de Groot, and J.J.M. (Joke) Geytenbeek, Amsterdam UMC, VU University, Amsterdam

Department of Child Neurology

R.J. (Jeroen) Vermeulen, Amsterdam UMC, VU University , Amsterdam (until October 2014), Maastricht University Medical Center, Maastricht (from October 2014)

J.S.H (Hans) Vles, D. (Dan) Soudant, and S. (Sabine) Fleuren, Maastricht University Medical Center, Maastricht

L.A. (Laura) van de Pol, Amsterdam UMC, VU University, Amsterdam

Department of Neurosurgery

W.J.R. (Pim) van Ouwerkerk, K.M. (Mariam) Slot and S.M. (Saskia) Peerdeman. Amsterdam UMC, VU University , Amsterdam

O.P. (Onno) Teernstra, Maastricht University Medical Center, Maastricht

Department of Clinical Neurophysiology

R.L.M. (Rob) Strijers, Amsterdam UMC, VU University , Amsterdam

Department of Neurology

E.M.J. (Elisabeth) Foncke, Amsterdam UMC, VU University, Amsterdam

Department of Epidemiology and Biostatistics

J.W.R. (Jos) Twisk and P. (Peter) van de Ven, Amsterdam UMC, VU University, Amsterdam



ASH,(modified) Ashworth scale	LP, lumbar punction
BADS, Barry-Albright dystonia scale	M, male
BFMDRS,Burke-Fahn-Marsden dystonia rating	m(o), month(s)
scale	MACS, manual ability classification system
C, cervical	METC, medical ethical committee
cm, centimeters	MRI, magnetic resonance imaging
CSF, cerebrospinal fluid	MUmc, Maastricht University medical center
DCP, dystonic cerebral palsy	N, number
DBS, deep brain stimulation	OL, open label
DIS, dyskinesia impairment scale	PEDI, Pediatric Evaluation of Disability
C-BiLLT, computer based instrument for low	Inventory
motorlanguage testing	PEG, percutaneous endoscopic gastrostomy
CFCS, communication functioning classification	PND, progressive neurological disease
system	RCT, Randomized clinical trial
CP, cerebral palsy	Ref, reference number
CR, case report	ROM, range of motion
CS, case series	SCP, spastic cerebral palsy
F, female	SD, standard deviation
FSP, familial spastic paraplegia	SDR, selective dorsal rhizotomy
FU, follow up	SPAT, spasticity test
GAS, goal attainment scaling	SRBD, sleep related breathing disorders
GMFCS, gross motor functioning classification	Th, thoracic
system	UE, upper extremities
h, hour	VAS, visual analogue scale
HSP, hereditary spastic paraparesis	VUmc, VU University medical center
ITB, intrathecal baclofen	y, year(s)
kg, kilograms	yr(s), year(s)
LE, lower extremities	μg, microgram
LOE, level of evidence	



DANKWOORD / WORD OF THANKS

Dit kan nog wel eens het langste hoofdstuk van dit proefschrift worden, want zoals het gezegde luidt: "Alleen ga je sneller, maar samen kom je verder". En een proefschrift schrijven doe je zeker, en gelukkig, niet alleen. Niet alleen tijdens de ruim negen jaar durende weg van het schrijven van mijn proefschrift, maar ook voor de start van mijn promotietraject, ben ik mensen tegengekomen die mijn pad, voor grotere of kleinere delen, met mij mee hebben bewandeld. Zonder deze mensen zou ik niet staan waar ik nu sta, en ik ben velen daarvoor dankbaar.

Als eerst dank aan de **kinderen, jongeren en hun ouders** die meegedaan hebben aan de studies in dit proefschrift, zonder jullie deelname was dit proefschrift er nooit gekomen.

Prof. dr. J.G. Becher, prof. dr. J.G. Vermeulen en dr. A.I. Buizer waren als begeleiders nauw betrokken bij mijn promotie. Beste Jeroen, we go back long time! Vanaf 2004 als ik mij niet vergis. Jij was mijn kritische begeleider op afstand terwijl ik als medisch student een wetenschappelijke stage in Canada deed. Het begin van een mooie samenwerking als je het mij vraagt. Je wijze adviezen over zowel mijn loopbaan als andere, meer persoonlijke, toekomstplannen zijn belangrijk geweest in keuzes die ik heb gemaakt. Beste Jules, het eerste college wat ik van jou had tijdens mijn geneeskunde opleiding kan ik mij nog levendig herinneren. Ik heb werkelijk geen idee waar het over ging (waarschijnlijk over CP) maar je enthousiasme en levendige manier van presenteren zijn me altijd bij gebleven. Jaren later begon ik onder jouw hoede mijn opleiding tot revalidatiearts. Ik ben blij dat je ondanks je pensioen toch mijn promotor bent gebleven. Dank voor al jouw enthousiasme, steun, kennis en kunde waarop ik altijd heb kunnen vertrouwen. En ook niet onbelangrijk: dat je me hebt geleerd dat ik beter niet kan gaan pokeren. Lieve **Annemieke**, in vele rollen kom ik je tegen en werk(t)en we samen: eerst als een van mijn supervisoren, nu als hoofd van de kinderrevalidatiegeneeskunde, directe collega en copromotor. Ondanks je overvolle agenda mag ik altijd bij je binnen komen voor advies of gewoon om mijn hart te luchten. Dank voor al je begrip, adviezen, steun, begeleiding en geduld de afgelopen jaren. Ik ben trots dat ik met jou verder mag werken in het CP expertise centrum.

Prof. dr. H. van Goudoever, Prof. Dr. V. de Groot, , Prof. dr. M. de Koning-Tijssen, Prof. dr. E. Monbaliu en dr. O. Teernstra, de **leescommissie**, dank voor jullie tijd en kritische blik bij het lezen en beoordelen van mijn proefschrift. Prof. van Goudoever, **Hans**, het is fijn te merken dat jij de kinderrevalidatiegeneeskunde een warm hart toe draagt. Dank dat je voorzitter van de leescommissie wilde zijn. **Vincent**, zonder jou had ik hier niet gestaan want met jou had ik mijn sollicitatiegesprek voor de opleidingsplek tot

revalidatiearts. Ik ben blij dat je me destijds aangenomen hebt en dat ik een aantal jaar later als staflid mocht blijven met jou inmiddels als afdelingshoofd. **Elegast**, bedankt voor het beantwoorden van al mijn vragen over de DIS. We hebben de afgelopen jaren frequent heen en weer gebeld, gemaild en besprekingen gehad. Ik verheug me op de toekomstige Belgisch-Nederlandse en natuurlijk ook de Europese samenwerking die we vorm aan het geven zijn. **Marina**, dank voor de fijne samenwerking binnen het landelijk overleg bewegingsstoornissen bij kinderen en het kijkje dat ik heb mogen nemen in de Groningse DBS keuken.

Speciale dank voor de kerngroep van het IDYS onderzoeksteam in zowel het VUmc als MUmc: Prof. dr. J.G. Becher, prof. dr. J.G. Vermeulen, dr. A.I. Buizer, prof. dr. J.S.H. Vles, mw. K. Boeschoten, dhr. D. Soudant, mw. S. Fleuren, dr. H. Haberfehlner. Hans, na jouw pensioen heb je het stokje aan Jeroen overgedragen. Je kritische blik heb ik altijd zeer gewaardeerd. Zo ook de hartelijkheid waarmee je me altijd ontvangen hebt tijdens mijn bezoekjes aan Maastricht, en onze mailwisselingen over onderzoek en de goede dingen in het leven. Karin, als meet-assistent in het VUmc was jij voor mij van onschatbare waarde. Bedankt voor al je tijd en inzet. Het was veel werk. Dan en Sabine, voor jullie geldt eigenlijk hetzelfde. Dank voor het plannen en uitvoeren van alle metingen in Maastricht. Helga, fijn dat jij als nieuwste lid ons team bent komen versterken. Ik kijk uit naar onze toekomstige projecten. De IDYS study group wil ik bedanken voor alle input tijdens de verschillende fasen van het onderzoek. In het VUmc waren dit prof. dr. J.G. Becher, dr. A.I. Buizer, mw. K. Boeschoten, dr. J.J.M. Geytenbeek prof. dr. V. de Groot (afdeling revalidatiegeneeskunde), dr. L.A. van de Pol (afdeling kinderneurologie), dr. W.J.R. van Ouwerkerk, drs. K.M. Slot, prof. dr. S.M. Peerdeman (afdeling neurochirurgie), dr. R.L.M. Strijers (afdeling klinische neurofysiologie), dr. E.M.J. Foncke (afdeling neurologie), prof. dr. J. Twisk, en dr. P. van de Ven (afdeling epidemiologie en biostatistiek). In het **MUmc** waren prof. dr. R.J. Vermeulen, prof. dr. J.S.H. Vles, dhr. D. Soudant, mw. S. Fleuren (afdeling neurologie) en dr. O. Teernstra (afdeling neurochirurgie) betrokken. Een deel van de metingen is uitgevoerd op de afdeling klinische neurofysiologie voornamelijk door dr. R.L.M. Strijers en mevr. A. Jongkind en later ook door dr. A. Gouw. Rob, Anita en Alida, hartelijk dank voor de fijne samenwerking. Daarnaast is de C-BiLLT afgenomen door dr. J.J. Geytenbeek, mw. J. Bootsma en mw. E. Vaillant. Joke, dank voor onze gesprekken onder andere over het wel en wee van het promoveren. Jaël, gelukkig blijf je aan de andere kant van het grote water betrokken bij de afdeling. Ik denk met veel plezier terug aan onze gesprekken over vega(n)-recepten en grappen over baby-spinazie. Emma, super fijn dat je ons team bent komen versterken. Over een tijdje mag jij jouw proefschrift verdedigen! De MRI's zijn gescoord door de kinderneurologen prof. dr. R.J. Vermeulen en dr. L.A. van de Pol.

Beste Laura, ik ben blij met onze fijne samenwerking en word erg enthousiast van onze brainstormsessies over wat we in de toekomst allemaal nog willen bereiken met onder andere verder onderzoek. Ik kijk ernaar uit hier verder vorm aan te geven. Grote dank aan de **neurochirurgen**, dr. W.J.R. van Ouwerkerk, dr. M. Slot, en dr. O. Teernstra voor het uitvoeren van de pompimplantaties. **Pim**, zo simpel is dat handwerken van jou niet. Veel dank aan de fantastische ITB verpleegkundigen/P.A. die betrokken zijn geweest: **Tamara, Sharony** en **Renske.** Zij zorgden voor het vullen en bijstellen van alle pompen en dat ik op tijd nieuwe onderzoeksmedicatie bestelde. We hebben kritische vragen en goede adviezen gekregen van de **Data Safety Monitoring Board** en de **monitor** van het Clinical Research Bureau, in het bijzonder wil ik prof. dr. P. Huijgens, prof. dr. M. Boers en mw. M. van Leeuwen bedanken.

Naast onderzoeker ben ik natuurlijk ook nog kinderrevalidatiearts waarbij ik in een fantastisch gedreven en bedreven **kinderrevalidatieteam** mag werken. Hier ben ik ontzettend trots op! Lieve Els, Karin, Karen, Mirjam, Eline, Petra, Veronique, Danny, Joke, Emma, Liza, Eefje, Jacqueline, Hestien, dank daarvoor. Ook de lieve dames van de **backoffice, planning en secretariaat**, in het bijzonder Monique, Marit en Heleen, wil ik bedanken voor hun geduld, gezelligheid en creatieve oplossingen.

Een aantal van mijn **mede-onderzoekers** op de afdeling ben ik extra dank verschuldigd. José Maas, Lizeth Sloot en Helga Haberfehlner, dank voor jullie hulp bij mijn EMG vragen en andere onderzoeksperikelen die mijn doktersbrein te boven gingen. Mijn collega **revalidatieartsen in het VUmc**: **Vincent, Annemieke, Louise, Eveline, Carel** en oud-collega's **Mirjam en Renske**, dank voor jullie interesse in de voortgang van het onderzoek en begrip voor mijn afwezigheid als de onderzoekstaken daarom vroegen. **Sanne**, bedankt dat je tijdens jouw fellowship en mijn zwangerschapsverlof de honneurs waar wilde nemen tijdens de metingen. **Eveline,** top dat je mijn kamergenoot bent en ik je ten alle tijden kan storen met al mijn promotie vragen! Dank voor alle goede adviezen. Ik vind het super fijn dat je me bij wil staan als paranimf! We hebben wel een Breezertje verdiend. Het team van het **CP-expertise centrum: Annemieke, Laura, Jaap, Marjolein, Petra, Els, Annet, Pim, Mariam, Melinda, Hans, Agnita en Barend,** wil ik bedanken voor de inspirerende samenwerking binnen onderzoek en onze gezamenlijke patiëntzorg.

Manin Konijnenbelt, Kirsten Nienhuis, Judith Vloothuis en Hanneke Aussems wil ik bedanken voor het enthousiasme waarmee jullie me begeleid hebben bij mijn eerste baan als anios in de revalidatiegeneeskunde. Jullie hebben me enthousiast gemaakt voor ons fantastische vak. De opleiding en het fellowship zouden niet zo leuk geweest zijn zonder jullie: Marloes, Francisca, Conny, Olga, Wietske, Irene, Margriet, Maaike, Rimke en Eveline.

Ook wil ik nog even terug naar het begin van mijn onderzoekscarrière, een wetenschappelijke stage in Canada bij **prof. dr. M. Tarnopolsky**. Mark, I finally am a double doctor, and I owe you many thanks. Without knowing me, you agreed for me to come to Hamilton to conduct a study in boys with Duchenne Muscular Dystrophy. You made me see the beauty of scientific research and how wonderful it goes together with work as a clinician. This has been an inspiration for me troughout my career. Furthermore, I would never have met these lovely ladies if it weren't for you: **Alissa, Erin** and **Michaela**, my dear friends, I so wished you lived closer so we could see more of each other. I really hope we will one day plan a grand reunion together with our families, preferably somewhere on a tropical island!

Mijn lieve vriendinnetjes, ook al zijn we allemaal druk en wonen we niet meer allemaal bij elkaar om de hoek, het is zeker niet uit het oog uit het hart. Het is altijd fijn om jullie te spreken over werk maar vooral ook over de andere dingen in het leven. Elsbeth, ik ben zo blij dat we nadat we tijdens onze basisschooltijd en middelbare schooltijd waarin we voor tweeling, stelletje en wie weet wat nog meer zijn aangezien, nog steeds zulke goede vriendinnen zijn. Ik verheug me op het volgende etentje, met of zonder onze mannen. Carlijn, ook al waren we jong en hebben we maar een paar jaar bij elkaar in de klas gezeten (groep 2 t/m 4) voordat ik naar het zuiden van het land, we zijn elkaar altijd blijven zien. Ik vind het bijzonder dat we na zoveel jaren, nog steeds zo'n goed contact hebben. Lieve geneeskunde (en bewegingswetenschappen) vriendinnetjes: Jantina, Nina, Annelies, Tineke en Laurien, jarenlang hebben we lief en leed gedeeld, dit blijft waardevol. Jantina, dank voor alle fijne en gezellige gesprekken tijdens onze eet dates, je steun en enthousiasme. Voor mij ben je goud waard! Marloes, jouw blik op zaken levert altijd weer nieuwe verfrissende inzichten op en ik ben blij dat je ook van gedoseerd klagen, wijn, eten en gezelligheid houdt. De andere kant van het land is gelukkig minder ver gebleken dan het lijkt! Andere zeer gewaardeerde vrienden en vriendinnen bedankt voor alle leuke middagen, avonden, (ski)tripjes, de vrolijke en minder vrolijke momenten in het leven die we samen mochten delen en nog mogen gaan delen in de toekomst, in het bijzonder: Menno en Maartje, Lukas en Michi, Willemijn en Arno, Frits en Floor, Kees en Lizet, Erik en Saskia, Steef en Gijs, Tom en Eveline, Mark en Maureen, Liselot en Vincent.

Lieve **familie en schoon familie**, sommigen ver weg en anderen dichtbij, bedankt voor jullie belangstelling in het onderzoek en het meeleven tijdens dit traject. Het is zo

fijn dat ieders deuren altijd open staan voor geplande en spontane familiebezoekjes, etentjes, borrels en noem maar op.

Marleen en **Huub**, wat ben ik blij met jullie als zus en zwager, het is altijd fijn om tijd met elkaar door te brengen. Liefste sis, wie had er in onze tienerjaren ooit gedacht dat wij zij aan zij zouden staan op een dag als deze. Dank dat je mijn paranimf wil zijn. Ik kan altijd op jou, en Huub, rekenen en ben ontzettend trots op wie je bent, en dol op jullie prachtige kinderen, Jack en Kate.

Lia en Ton, papa en mama, ik heb het getroffen met ouders zoals jullie. Door jullie levenslessen, 2 in het bijzonder, heb ik mij altijd gesteund gevoeld maar ook gemotiveerd en vrij om mijn eigen pad uit te stippelen. Dank voor jullie vertrouwen. Speciale dank voor het oppassen op de jongens, helemaal in de laatste fase van mijn proefschrift, hierdoor had ik tijd en rust om mijn boekje af te ronden.

Loek en Quint, mijn jongens, mijn alles. Ik ben zo ontzettend trots op wie jullie zijn en dankbaar dat jullie in ons leven zijn. Jullie maken de wereld mooi.

Lodewijk, mijn liefde, mijn leven. Wat hebben we het goed samen! En wat boffen we met onze jongens! Met jou wil ik later oud worden maar eerst samen nog heel veel mooie momenten beleven.

Dankwoord / Word of thanks





fotograaf: Anita Edridge

Laura Alexandra Bonouvrié was born on Sunday March 15th, 1981 in the Military Hospital in Utrecht. Most of her youth she lived in Waalre with her parents and sister Marleen. She completed high school education (VWO) at the Augustinianum in Eindhoven in 1999. After graduation she moved to Amsterdam to study Human Movement Sciences. The year after she was accepted for medical school at the VU University. For her final bachelor thesis, she spend seven months in Hamilton, Ontario, Canada for a research project on the effect of antioxidants in boys with Duchenne Muscular with prof. dr. Mark Tarnopolsky. She obtained her medical degree in 2007 and continued with research, working with dr. Jeroen Vermeulen at the Department of Child Neurology at the VU University medical center (VUmc). Her interest in intrathecal baclofen and dyskinetic cerebral palsy (CP) was aroused during this project.

After working as a resident (not in training) at two neurology departments in Amsterdam, she worked at the rehabilitation center Reade in Amsterdam (formerly Rehabilitation Center Amsterdam) where her interest in rehabilitation medicine was aroused. Laura started her physiatry residency shortly after, in November 2009, at the Department of Rehabilitation Medicine of the VUmc (now Amsterdam UMC). At that time her PhD project also started with writing of the protocol. After completion of her residency in December 2014, she continued with a 2 year fellowship in pediatric physiatry under supervision of prof. dr. Jules Becher and dr. Annemieke Buizer. During this fellowship

she specialized in complex care for patients with spasticity and dystonia (mainly due to CP). Since January 2017, she works as a pediatric physiatrist at the Amsterdam UMC, location VUmc, combining clinical work, research and education.

During her PhD training, as a member of the JOK (young researchers in pediatrics Amsterdam UMC), she organized several trainings for PhD students. She is currently the coordinator of the minor Body in Motion and a member of the national training committee for pediatric physiatry. Furthermore, she is part of the CP center of expertise in the Amsterdam UMC which is led by Annemieke Buizer (www.cpexpertisecentrum. nl), and a promotor of the Dutch National Consultation Group Movement Disorders in Children (LOBBK; Landelijk Overleg Bewegingsstoornissen Bij Kinderen) and the European Dyskinetic CP Network.

Laura lives in Amstelveen with Lodewijk van der Meulen and their two sons, Loek (born Oct 2013) and Quint (born Nov 2015).

LIST OF PUBLICATIONS AND PRESENTATIONS

This thesis is based on the following international peer-reviewed publications:

1. <u>Bonouvrié LA</u>, van Schie PEM, Becher JG, Ouwerkerk WJR, Vermeulen RJ.Satisfaction with Intrathecal baclofen treatment in pediatric patients with progressive neurological disease. *Developmental Medicine and Child Neurology 2008;50(8):636-638*

2. <u>Bonouvrié LA</u>, van Schie PEM, Becher JG, van Ouwerkerk WJR, Reeuwijk A, Vermeulen RJ. Effects of intrathecal baclofen on daily care in children with secondary generalised dystonia:a pilot study. *European Journal of Paediatric Neurology 2011;15:539-543*

3. <u>Bonouvrié LA</u>, van Schie PEM, Becher JG, Ouwerkerk WJR, Vermeulen RJ.Intrathecal baclofen for progressive neurological disease in childhood; a systematic review of literature. *European Journal of Paediatric Neurology 2012;16:279-284*

4. <u>Bonouvrié LA</u>, Becher JG, Vles JS, Boeschoten K, Soudant D, de Groot V, van Ouwerkerk WJ, Strijers RL, Foncke E, Geytenbeek J, van de Ven PM, Teernstra O, Vermeulen RJ. Intrathecal baclofen treatment in dystonic cerebral palsy: a randomized clinical trial: the IDYS trial. *BMC Pediatrics 2013; 13: 175*

5. <u>Bonouvrié L</u>, Becher J, Soudant D, Buizer A, van Ouwerkerk W, Vles G, Vermeulen J. The effect of intrathecal baclofen treatment on activities of daily life in children

and young adults with cerebral palsy and progressive neurological disorder. *European Journal of Paediatric Neurology, 2016 ;20 :538-544*

6. Monbaliu E, Himmelmann K, Lin J-P, Ortibus E, <u>Bonouvrié L</u>, Feys H, Vermeulen RJ, Dan
B. Clinical presentation and management of dyskinetic cerebral palsy. *Lancet Neurology* 2017;16 :743-749

7. <u>Bonouvrié LA</u>, Becher JG, Vles JSH, Vermeulen RJ, Buizer AI, on behalf of the IDYS study group. The effect of intrathecal baclofen in dyskinetic cerebral palsy : the IDYS trial. *Annals of Neurology 2019; epub ahead of print*

Other international peer-reviewed publications:

1. Vles GF, Soudant DL, Hoving MA, Vermeulen RJ, <u>Bonouvrié LA</u>, Vles JSH. Long-term follow-up on Intrathecal Baclofen therapy in non-ambulant children with intractable spastic Cerebral Palsy. *European Journal of Paediatric Neurology 2013 Nov;17(6):639-44*

Book chapters:

1. Vermeulen RJ, <u>Bonouvrié LA</u>, Buizer AI, van Ouwerkerk WJR, van Aalst J, Teernstra O, Soudant D, Becher JG (2018). Intrathecal baclofen for dyskinetic cerebral palsy. In: Dressler D, Altenmüller E and Krauss JK (Eds). Treatment of Dystonia (p 229-232). Cambridge: Cambridge University Press.

Invited editorial commentaries:

1. <u>Bonouvrie LA</u>. Intrathecal baclofen in progressive neurological disease: to be considered before all other options fail. Editorial commentary on 'Intrathecal Baclofen Treatment: an option in X-linked Adrenoleukodystrophy' by Helgi Thor Hjartarson, Christoffer Ehrstedt and Kristina Tedroff. *Europ J Paediatr Neurol 2018; 22(1):1*

2. <u>Bonouvrié LA</u>. How to assess goals in intrathecal baclofen therapy. Commentary on: 'Intrathecal baclofen therapy in children: identifying and improving individualised goals' by Peck Yee Liew et al. *Dev Med Child Neurol 2018; 60(4):332*

Contributions to scientific conferences:

Poster presentations:

1. <u>Bonouvrié LA</u>, Becher JG, van Schie PEM, van Ouwerkerk WJR, Reeuwijk A, Vermeulen RJ. Effects of intrathecal baclofen on daily care in children with secondary generalized dystonia: results of a pilot study and a glimpse into the future. Poster. Kindersymposium Amsterdam, 16 februari 2012.

2. <u>Bonouvrié LA</u>, Becher JG, Vles JSH, Soudant D, van Ouwerkerk WJR, Teernstra O, Vles GF, Vermeulen RJ. Comparing the effects of intrathecal baclofen treatment: diagnosis does not matter (much). Poster. VUmc Science Exchange Day, 12 februari 2013.

3. <u>Bonouvrié LA</u>, Becher JG, Vles JSH, Soudant D, van Ouwerkerk WJR, Teernstra O, Vles GF, Vermeulen RJ. The effect of Intrathecal baclofen in spastic cerebral palsy, dystonic cerebral palsy and progressive disease: does diagnosis matter? Poster. Kindersymposium Amsterdam, 6 februari 2014.

4. <u>Bonouvrié LA</u>, Becher JG, Soudant D, Buizer AI, van Ouwerkerk WJR, Vles GF, Vermeulen RJ. The effect of intrathecal baclofen treatment on activities of daily life: does diagnosis matter? Poster. Paediatric Movement Disorders, Barcelona, Spanje, 20 februari 2015.

Oral presentations:

5. <u>Bonouvrié LA</u>, Becher JG, van Schie PEM, van Ouwerkerk WJR, Reeuwijk A, Vermeulen RJ. Intrathecale baclofen behandeling bij dystonie: resultaten van een pilot studie en een blik op de toekomst. Oral presentation. Nederlandse Vereniging voor Kinderneurologie, Middelburg, the Netherlands, April 27th 2012.

6. <u>Bonouvrié LA</u>. Intrathecal baclofen treatment. Invited speaker. European Society for Paediatric Neurosurgeons, Amsterdam, the Netherlands, April 5th 2012.

7. <u>Bonouvrié LA</u>, Becher JG, van Schie PEM, van Ouwerkerk WJR, Reeuwijk A, Vermeulen RJ. Intrathecal baclofen treatment in dystonic cerebral palsy. Oral presentation. 4th International Cerebral Palsy Conference, Pisa, Italy, October 12th 2012.

8. <u>Bonouvrié LA</u>. Intrathecale baclofen bij dystone cerebrale parese. Invited speaker. Medtronic User Day, St. Michielsgestel, the Netherlands, June 13th 2014.

9. <u>Bonouvrié LA</u>, Becher JG, Vermeulen RJ. IDYS trial: Intrathecal baclofen treatment for dystonic cerebral palsy. Oral presentation. European Society Paediatric Neurology, Boekarest, Roemenia, September 12th and 13th 2014.

10. <u>Bonouvrié L</u>. Oral presentation. Dyskinesieën en tonusregulatie. Invited speaker. Spasticiteits Symposium, Lunteren, the Netherlands, March 27th 2015.

11. <u>Bonouvrié L</u>. Intrathecal baclofen treatment in dyskinetic cerebral palsy. Part of the mini-symposium 'Dyskinetic cerebral palsy: from a better understanding towards a targeted management'. Speakers: Monbaliu E, Himmelmann K, Ortibus E, Bonouvrié L, Lin J-P. European Academy of Childhood Disabilities, Stockholm, Sweden, June 1st-4th 2016.

12. <u>Bonouvrié LA</u>. 1. Clinical patterns and the influence on activities and participation, and 2. Intrathecal baclofen treatment. Parts of the mini-symposium 'Dyskinetic Cerebral Palsy – from movement disorder to targeted treatment'. Speakers: L.A. Bonouvrié, A.I. Buizer, R.J. Vermeulen, Y. Temel. Dutch Congress of Rehabilitation Medicine, Maastricht, the Netherlands, November 9th and 10th 2017

13. <u>L.A. Bonouvrié</u>. Intrathecal baclofen in dyskinetic cerebral palsy. Invited speaker. Symposium "Dystonia and choreo-athetosis in dyskinetic cerebral palsy: clinical presentation and management". Brugge, Belgium, March 17th 2018.

14. <u>L.A. Bonouvrié</u>. Workshop "ITB in dyskinetic CP: clinical cases". Symposium "Dystonia and choreo-athetosis in dyskinetic cerebral palsy: clinical presentation and management". Invited speaker. Brugge, Belgium, March 17th 2018.
About the author

